# DIVERSIFICATION OF POTATO FARMING SYSTEMS THROUGH LEGUME INTERCROPPING FOR IMPROVED RESOURCE USE EFFICIENCY

SHADRACK OTIENO NYAWADE

# A THESIS SUBMITTED IN FULFILMENT OF THE REQUIREMENTS FOR THE DEGREE OF DOCTOR OF PHILOSOPHY IN SOIL SCIENCE

# DEPARTMENT OF LAND RESOURCE MANAGEMENT AND AGRICULTURAL TECHNOLOGY,

# COLLEGE OF AGRICULTURE AND VETERINARY SCIENCES

UNIVERSITY OF NAIROBI

©2020

#### DECLARATION

This thesis is my original work and has not been presented for award of a degree in any other university.

Signature .....

Date .....

Otieno Shadrack Nyawade

Department of Land Resource Managemnet and Agricultural Technology, University of Nairobi

This thesis has been submitted to the Graduate School for examination with our approval as the student supervisors.

Navan ja

Signature .....

Date: 2<sup>nd</sup> November 2020

Prof. Nancy Karanja

Department of Land Resource Managemnet and Agricultural Technology, University of Nairobi

harmin

Signature .....

Date: November 2<sup>nd</sup> 2020

Prof. Charles K.K Gachene

Department of Land Resource Managemnet and Agricultural Technology, University of Nairobi

mal

Signature .....

Date: 2<sup>nd</sup> November 2020

Dr. Monica Lyn Parker

International Potato Centre (CIP)

## DEDICATION

To my parents Charles Ochuodho, Teresa Akong'o and Consolata Auma for their love, care and passion for education

and

To my brother Dr. Benson Okinyi for his inspirational phylosophy '*Chak achaka*' (Whatever *looks like a mountain soon becomes an anthill only if you begin*)

#### ACKNOWLEDGEMENT

That this document is presented in this form is an absolute manifestation of the good health and protection from His Almighty God. My supervisor, Prof. Nancy Karanja, best described as a mother and a mentor remains the centerpiece to this work. Your motivation, comments, guidance, expertise and rigor are highly regarded. Prof. Charles Gachene, whom I best describe as 'Mwalimu Mkuu', remains the cornerstone to my academic excelence. Your comments, suggestions and mentorship remain groundbreaking. This work would stand no initiation and progress without the unconditional support and love offered by Dr Monica Lyn Parker. Thank you for granting me a full scholarship and the necessary equipment and personnel to collect data. Your comments, guidance and motivation are highly valued. Abundant regards go to Prof. Elmar-Schulte Geldermann for identifying and igniting the potential in me. Your all-round guidance and well-articulated advice are immeasurable. Dr Kalpana Sharma was so instrumental in providing technical guidance, advice and mentorship.

The host farmers are highly acknowledged for unreservedly providing land and routine crop husbandry during the study period. I equally hold with high regard the immense support, love and motivation accorded by my parents and siblings. I am indebted to Boniface Muliro, John Kimotho, Ferdinand Anyika, Stanley Kisamuli, Dr Anne Karuma, Dr Karuku George and Annerose Mwangi for their technical support. The following represent an endless list of great friends whose love, motivation and advice inspired this work: Dr Harun Gitari, Dr Dinah Borus, Dr Oliver Otieno, James Njeru, Bruce Ochieng, Elly Atieno, Naman Arodi, Gidraf Okeyo, Hillary Otieno, Herman Machavure, Collins Ouma, Rodgers Rogito, Alphayo Luta, Sussy Munialo, Janneth Chepkemoi and Getrude Kavochi.

The Department of LARMAT and the University of Nairobi at large offered a great tutelage. It is also worthy of note that this work was implemented and funded as part of the CGIAR Research Program on Climate Change, Agriculture and Food Security (CCAFS) and with further funding from the CGIAR Research Program on Roots, Tubers and Bananas (RTB).

TABLE OF	CONTENTS
----------	----------

DECLARATIONii
DEDICATIONiii
ACKNOWLEDGEMENTiv
TABLE OF CONTENTS v
LIST OF FIGURES xii
LIST OF TABLES xiii
LIST OF APPENDICES xiv
LIST OF ACRONYMS AND ABBREVIATIONS xv
GENERAL ABSTRACT xvii
CHAPTER ONE 1
GENERAL INTRODUCTION1
1.1 Background information1
1.2 Statement of the problem
1.3 Justification of the study
1.4 Research objectives
1.4.1 Overall objective
1.4.2 Specific objectives
1.5 Hypotheses
1.6 Thesis outline7
CHAPTER TWO
LITERATURE REVIEW
2.1 Potato production systems in Kenya
2.2 Relationships among soil temperature, soil water content and nutrient uptake
2.3 Effect of cropping systems on soil organic matter fractions and microbial activity

2.4 Effect of legume intercropping and crop residue on soil nitrogen content	
2.5 Nutrient uptake by potato under legume intercropping	
2.6 Crop water productivity and light use efficiency	17
2.7 Considerations for a sustainable potato-legume intercropping	
CHAPTER THREE	
GENERAL MATERIALS AND METHODS	
3.1 Description of the study sites	
3.2 Treatments	
3.3 Conceptual framework	
3.4 Experimental design and layout	
3.5 Agronomic practices	
3.6 Soil sampling and soil physico-chemical analyses	
3.7 Data management and analyses	
CHAPTER FOUR	
RELETION OF SOIL TEMPERATURE, SOIL WATER BALANCE A	AND NUTRIENT
UPTAKE UNDER POTATO-LEGUME INTERCROPPING	
Abstract	
4.1 Introduction	
4.2 Materials and methods	
4.2.1 Study site	
<ul><li>4.2.1 Study site</li><li>4.2.2 Experimental layout and crop husbandry</li></ul>	
<ul> <li>4.2.1 Study site</li> <li>4.2.2 Experimental layout and crop husbandry</li> <li>4.2.3 Climatic data</li> </ul>	
<ul> <li>4.2.1 Study site</li> <li>4.2.2 Experimental layout and crop husbandry</li> <li>4.2.3 Climatic data</li> <li>4.2.4 Soil sampling and analyses</li> </ul>	
<ul> <li>4.2.1 Study site</li> <li>4.2.2 Experimental layout and crop husbandry</li> <li>4.2.3 Climatic data</li> <li>4.2.4 Soil sampling and analyses</li> <li>4.2.5 Determination of soil water content</li> </ul>	

4.2.7 Estimation of root water uptake	32
4.2.8 Estimation of soil water balance	34
4.2.9 Estimation of root length density	34
4.2.10 Nutrient uptake	35
4.2.11 Statistical analyses	36
4.3 Results	36
4.3.1 Vegetal cover establishment by different cropping systems	36
4.3.2 Relationship between soil moisture and soil temperature	37
4.3.3 Root length density	38
4.3.4 Root water uptake	39
4.3.5 Soil water storage as a function of evapotranspiration and soil evaporation	41
4.3.6 Principal component analyses of factors influencing soil water storage	42
4.3.7 Soil water balance	43
4.3.8 Intercropping effect on tuber yield	44
4.3.9 Nutrient (NPK) uptake and use efficiency	46
4.3.10 Relationship among soil water content, nutrient uptake and soil temperature	47
4.4 Discussions	48
4.4.1 Root water uptake	48
4.4.2 Evapotranspiration, soil evaporation and soil water storage	49
4.4.3 Principal factors explaining changes in soil water storage	52
4.4.4 Intercropping effect on crop yield	54
4.4.5 Nutrient uptake and use efficiency	54
4.5 Conclusions	56
CHAPTER FIVE	57

SHORT-TERM DYNAMICS OF SOIL ORGANIC MATTER FRACTIONS	S AND
MICROBIAL ACTIVITY IN SMALLHOLDER POTATO-LEGUME INTERCRO	PPING
SYSTEMS	57
Abstract	57
5.1 Introduction	58
5.2 Materials and methods	61
5.2.1 Study sites	61
5.2.2 Trial design and crop husbandry	61
5.2.3 Soil sampling and analyses	62
5.2.4 Soil moisture and soil temperature	62
5.2.5 Estimation of shoot and root biomass	62
5.2.6 Estimations of crop biomass lignin, polyphenols, C and N contents	63
5.2.7 Particle size separation and sequential density partitioning of soil organic matter	r 64
5.2.8 Soil microbial biomass	66
5.2.9 Microbial respiration	66
5.2.10 Estimation of enzyme activities	67
5.2.10.1 Fluorescein diacetate hydrolysis	67
5.2.10.2 Dehydrogenase activity	67
5.2.11 Data analysis	67
5.3 Results	68
5.3.1 Distributions of rainfall and air temperature	68
5.3.2. Biomass accumulation and residue composition	70
5.3.3 Distributions of aggregate sizes and soil organic matter fractions	71
5.3.4 Effect of cropping systems and sampling depths on distributions of SOM fraction	ons 74
5.3.5 Microbial biomass carbon to total SOC ratio	75
5.3.6 Microbial activities	

5.3.7 Principal component analyses of factors influencing soil organic matter dynam	ics 77
5.4. Discussions	79
5.4.1. Distributions of aggregate sizes and soil organic matter fractions	79
5.4.2. Effect of legume intercropping on soil microbial activity	83
5.4.3 Principal factors influencing dynamics of SOM fractions	85
5.5 Conclusion	87
CHAPTER SIX	88
OPTIMIZING SOIL NITROGEN BALANCE IN A POTATO CROPPING S	YSTEM
THROUGH LEGUME INTERCROPPING	88
Abstract	88
6.1 Introduction	89
6.2 Materials and methods	91
6.2.1 Experimental site and design	91
6.2.2 Experimental design and crop husbandry	
6.2.3 Soil sampling and analyses	
6.2.4 Soil moisture, rainfall and ambient temperature measurements	
6.2.5 Root sampling and estimation of root length density	
6.2.6 Computation of nitrogen balance	
6.2.7 Data analysis	
6.3 Results	99
6.3.1 Runoff and sediment yield	99
6.3.2 Residue yield and chemical composition	100
6.3.3 Nitrogen mineralization and uptake	101
6.3.4 Soil nitrogen balance	103
6.4 Discussion	105

6.4.1 Influence of potato-legume intercropping on soil erosion	105
6.4.2 Residue N mineralization and N uptake	
6.4.3 Soil N balance and its components	
6.5 Conclusion	
CHAPTER SEVEN	
INTERCROPPING OPTIMIZES SOIL TEMPERATURE AND INCRE	ASES CROP
WATER PRODUCTIVITY AND RADIATION USE EFFICIENCY OF POT	АТО 111
Abstract	
7.1. Introduction	
7.2 Materials and methods	
7.2.1 Experimental site and trial design	
7.2.2 Determination of soil water content and soil temperature	
7.2.3 Determination of leaf area index, light interception, crop yield, light exti	nction and
radiation use efficiency	
7.2.4 Estimation of crop water productivity	116
7.2.5 Assessment of intercrop productivity	117
7.2.6 Statistical analyses	
7.3 Results	
7.3.1 Climatic variables measured during the study period	
7.3.2 Leaf area index development	
7.3.3 Soil water content and soil temperature	119
7.3.4 Light interception and leaf extinction coefficient	
7.3.5 Intercropping effect on potato yield equivalent	121
7.3.6 Intercropping effect on crop water productivity and radiation use efficient	ncy of potato
7.3.7 Intercropping effect on land productivity	123

7.3.8 Relationships among potato yield, soil water content, soil temperature, crop water	
productivity and radiation use efficiency 1	24
7.4 Discussions 1	25
7.4.1 Variability of crop leaf area index 1	25
7.4.2 Soil temperature and soil water content 1	28
7.4.3 Effect of legume intercropping on radiation interception 1	28
7.4.4 Intercropping effect on biomass accumulation and land productivity 1	29
7.4.5 Relation of potato yield, crop water productivity and radiation use efficiency to	
temperature and soil water content 1	130
7.5 Conclusion 1	133
	121
CHAPTER EIGHT	134
GENERAL DISCUSSIONS, CONCLUSIONS AND RECOMMENDATIONS 1	134 134
GENERAL DISCUSSIONS, CONCLUSIONS AND RECOMMENDATIONS 1 8.1 General discussions	1 <b>34</b> 1 <b>34</b>
GENERAL DISCUSSIONS, CONCLUSIONS AND RECOMMENDATIONS 1 8.1 General discussions	1 <b>34</b> 1 <b>34</b> 134
GENERAL DISCUSSIONS, CONCLUSIONS AND RECOMMENDATIONS       1         8.1 General discussions       1         8.2 Conclusions       1         8.3 Implications of the study       1	134 134 137 137
GENERAL DISCUSSIONS, CONCLUSIONS AND RECOMMENDATIONS       1         8.1 General discussions       1         8.2 Conclusions       1         8.3 Implications of the study       1         8.3.1 Implications to potato and legume breeders       1	134 134 137 137
GENERAL DISCUSSIONS, CONCLUSIONS AND RECOMMENDATIONS       1         8.1 General discussions       1         8.2 Conclusions       1         8.3 Implications of the study       1         8.3.1 Implications to potato and legume breeders       1         8.3.2 Agronomic implications       1	<b>134</b> 134 137 137 137
GENERAL DISCUSSIONS, CONCLUSIONS AND RECOMMENDATIONS       1         8.1 General discussions       1         8.2 Conclusions       1         8.3 Implications of the study       1         8.3.1 Implications to potato and legume breeders       1         8.3.2 Agronomic implications       1         8.3.3 Policy implications and recommendations       1	<ul> <li>134</li> <li>134</li> <li>137</li> <li>137</li> <li>138</li> <li>138</li> </ul>
GENERAL DISCUSSIONS, CONCLUSIONS AND RECOMMENDATIONS       1         8.1 General discussions       1         8.2 Conclusions       1         8.3 Implications of the study       1         8.3.1 Implications to potato and legume breeders       1         8.3.2 Agronomic implications       1         8.3.3 Policy implications and recommendations       1         8.4 Research gaps       1	<ul> <li>134</li> <li>134</li> <li>137</li> <li>137</li> <li>137</li> <li>138</li> <li>138</li> <li>141</li> </ul>
GENERAL DISCUSSIONS, CONCLUSIONS AND RECOMMENDATIONS       1         8.1 General discussions       1         8.2 Conclusions       1         8.3 Implications of the study       1         8.3.1 Implications to potato and legume breeders       1         8.3.2 Agronomic implications       1         8.3.3 Policy implications and recommendations       1         8.4 Research gaps       1         REFERENCES       1	<ul> <li>134</li> <li>134</li> <li>137</li> <li>137</li> <li>137</li> <li>138</li> <li>138</li> <li>141</li> <li>142</li> </ul>

# LIST OF FIGURES

Figure 3.1: Map of study sites selected for establishment of the trials
Figure 3.2: Conceptual framework developed for the study
Figure 3.3: Schematic illustration of inter row spacing deployed for the cropping systems 24
Figure 4.1: Scanning (a) and processing (b) of roots using Epson Expression 1680 Scanner 35
Figure 4.2: Development of vegetal cover by different treatments
Figure 4.3: Relation of soil temperature and water content under different cropping systems 38
Figure 4.4: Cumulative evapotranspiration, evaporation and soil water storage under different
treatments
Figure 4.5: Principal component analyses of factors influencing soil water storage in a potato-
legume intercropping
Figure 4.6: Total water balance measured under different cropping systems
Figure 4.7: Potato and legume yields expressed in tonnes per ha and kg per plant in sole cropping
and intercropping systems
Figure 5.1: Schematic diagram of the combined aggregate size and density fractionation
Figure 5.2: Rainfall and air temperature recorded in the study sites
Figure 5.3: Soil aggregate distribution at the end of the study
Figure 5.4: Distributions of SOM fractions within aggregate classes under different cropping
systems73
Figure 5.5: Principal component analyses of factors influencing SOM dynamics
Figure 6.1: Mean seasonal soil loss and runoff measured under different cropping systems 99
Figure 6.2: Accumulations of residue N mineralized and N taken up by the crops 102
Figure 6.3: Amount of nitrogen leached below the active root zone
Figure 7.1: Development of leaf area index by different treatments
Figure 7.2: Land equivalent ratios measured under different cropping systems
Figure 7.3: Canopy overlap by potato grown alone (a) and potato intercropped with lima bean (b)
and dolichos

# LIST OF TABLES

Table 3.1: Climatic and soil descriptions of the experimental sites    22
Table 4.1: Initial soil physical and chemical properties       30
Table 4.2: Changes in crop root length density (m m <sup>-3</sup> ) at tuber bulking stage
Table 4.3: Root water uptake and the corresponding soil matric potential measured in different
cropping systems
Table 4.4: Nutrient uptake (NPK) and use efficiency by different cropping systems in the three
agro-ecological zones4'
Table 4.5: Pearson's correlation of soil temperature, nutrients (NPK) uptake and crop wate
balance47
Table 5.1: Initial soil properties.    62
Table 5.2: Accumulative residue biomass, carbon and nitrogen supply, residue chemical
composition and carbon to nitrogen ratio by different cropping systems
Table 5.3: Soil organic matter fractions and microbial biomass carbon measured under different
cropping systems74
Table 5.4: Microbial biomass carbon (MBC) to total soil organic matter carbon (SOMC) ratio as
influenced by cropping systems
Table 5.5: Soil microbial respiration and enzyme activities in response to cropping systems and
soil depth
Table 6.1: Initial soil chemical properties measured at different soil depth
Table 6.2: Dry matter yield, nitrogen concentrations and chemical composition of crop residues
at time of field incorporation
Table 6.3: Nitrogen balance and its components under different cropping systems
Table 7.1: Mean rainfall amount, diurnal and night temperatures for the period between potato
planting and harvesting
Table 7.2: Intercropping effect on soil moisture content and soil temperatures
Table 7.3: Intercropping effect on light interception and leaf orientation.       12
Table 7.4:Potato yield as affected by intercropping across the four study seasons
Table 7.5: Crop water productivity and radiation use efficiency in response to intercropping. 123
Table 7.6: Response of potato yield, crop water productivity and radiation use efficiency to soil
moisture and temperature

# LIST OF APPENDICES

Appendix 1: Soil water balance components measured under different cropping systems at the
depth of 0–30 cm
Appendix 2: Ratio of actual evapotranspiration (ETa) to potential evapotranspiration (ETp)
under different treatments in the three agro-ecological zones
Appendix 3: Distributions of SOM fractions within aggregate classes ( $\mu m$ ) under different
cropping systems
Appendix 4: Carbon to nitrogen ratio of microbial biomass and total soil organic matter as
influenced by soil depth (cm) and cropping systems
Appendix 5: Microbial carbon biomass (MBC) in relation to lignin content under different
cropping systems
Appendix 6: Seasonal variations in nitrogen leached under different cropping systems
Appendix 7: Seasonal nitrogen fixation (BNF) by legume intercropping under different cropping
systems
Appendix 8: Seasonal nitrogen uptake under different cropping systems
Appendix 9: Seasonal soil nitrogen balance under different cropping systems
Appendix 10: Schematic presentation of soil organic matter fractionation
Appendix 11: Schematic presentation of determination of soil microbial respiration. Procedures
are outlined in section 5.2.9
Appendix 12: Schematic presentation of fluorescein diacetate hydrolysis
Appendix 13: Runoff plots at 2 weeks after planting in the lower highland (a) and upper highland
(b) agro-ecological zones

## LIST OF ACRONYMS AND ABBREVIATIONS

AEZ	Agro-Ecological Zone
ANOVA	Analysis of Variance
CAN	Calcium Ammonium Nitrate
CGIAR	Consultative Group for International Agricultural Research
CIP	International Potato Center
CWP	Crop Water Productivity
DAP	Days After Planting
DHA	Dehydrogenase Activity
DM	Dry Matter
DOM	Dissolved Organic Matter
Dp	Deep percolation
Ep	Evaporation
Es	Saturation vapor pressure
ЕТо	Reference Evapotranspiration
FAO	Food and Agricultural Organization
FDA	Fluorescein Diacetate
fPOM	Free Particulate Organic Matter
GPS	Global Position System
HFOM	Heavy Fraction Organic Matter
HSD	Honest Significant Difference
INTF	Iodo Nitro Tetrazolium Formazan
Kc	Crop coefficient factor
Ks	Saturated hydraulic conductivity
LAI	Leaf Area Index
LFOM	Light Fraction Organic Matter
LH	Lower Highland
Masl	Meters above sea level
MBC	Microbial Carbon

MBN	Microbial Nitrogen
MoALF	The Kenya Ministry of Agriculture, Livestock and Fisheries
NH4 <sup>+</sup> -N	Ammonium Nitrogen
NO <sub>3</sub> <sup>-</sup> -N	Nitrate Nitrogen
OECD	The Organization for Economic Co-operation and Development
oPOM	Occluded Particulate Organic Matter
PAR	Photosynthetically Active Radiation
PEY	Potato Equivalent Yield
qCO <sub>2</sub>	Microbial metabolic quotient for Carbon (ii) Oxide
RD	Rooting Depth
RLD	Root Length Density
RO	Runoff
RUE	Radiation Use Efficiency
RWU	Root Water Uptake
SMR	Soil Microbial Respiration (SMR)
SOC	Soil Organic Carbon
SOM	Soil Organic Matter
SWC	Soil Water Content
Тр	Potential transpiration rate
TSP	Triple Super Phosphate
UH	Upper Highland
UM	Upper Midland
θfc	Soil water content at field capacity
θs	Soil water content at saturation
θν	Volumetric soil water content
θwp	Soil water content at permanent wilting point

#### **GENERAL ABSTRACT**

Expanding production of potato to the midlands and lowlands of Kenya will only be possible if the effect of water deficit, high temperatures and nutrient limitation on potato growth is well understood. Potato (Solanum tuberosum L.) was grown singly and intercropped with lima bean (Phaseolus lunatus L.) or dolichos (Lablab purpureus L.), and the respective single crop of the legumes in three agro-ecological zones (AEZs) of Kenya; upper midland (1552 meters above sea level (masl)), lower highland (1894 masl) and upper highland (2552 masl). The objectives were as follows: i) determine the interelationships between soil water balance, soil temperatures and crop nutrient (NPK) uptake under potato-legume intercropping ii) assess the short-term effect of potatolegume intercropping on dynamics of microbial activity and SOM fractions, iii) evaluate the effect of potato-legume intercropping on soil nitrogen balance and iv) quantify the radiation and crop water productivity of potato-legume intercropping system. The study was laid out in a randomized complete block design with four replications. All the potato treatments received basal fertilization at a rate of 50 kg N ha<sup>-1</sup>, 90 kg P ha<sup>-1</sup> and 100 kg K ha<sup>-1</sup> and topdress of 40 kg N ha<sup>-1</sup>. Soil temperatures and soil water contents (SWC) were quantified at different stages of potato growth. Residues from each cropping system were quantified at the end of each season and incorporated back into the soil at start of the subsequent season. Soil samples (0-120 cm depths) were taken at the interrows of each plot prior to planting, at vegetative growth, and at the end of each season. Physical and density fractionation procedure was used to separate the soil in macro-aggregates (>250 µm), micro-aggregates (250–50 µm) and silt plus clay fractions (<50 µm), while SOM was partitioned into labile (density of 1.65 to 1.85 g cm<sup>-3</sup>) and stable (2.60 g cm<sup>-3</sup>) fractions. Microbial biomass carbon was determined by chloroform fumigation while enzymatic activities were assessed by hydrolyses of fluorescein diacetate and dehydrogenase. Nitrogen gains from mineralization, fertilization, and outputs from biomass accumulation, leaching, and soil erosion

were quantified throughout the potato growth cycle. N losses through volatilization were quantified using ventilation chambers while the amount of N leached was extracted using solusamplers located at vertical depths of 0-1.5 m and the leachetes analyzed for nitrate contents. Leaf area index (LAI) and light interception were quantified at different stages of potato growth and related with the radiation use efficiency (RUE) and crop water productivity (CWP). The data was subjected to mixed model analyses of variance using R software with Tukeys mean separation test set at 95% propability level. Intercropping increased LAI by 26-57% relative to sole potato and significantly (p < 0.05) lowered soil temperatures in the 0–30 cm depth by up to 7°C. This increased SWC by up to 38%, thus increasing RUE by 56–78%, CWP by 45–67% and nutrient use efficiency by 40-67%. Compared with the sole potato, intercropping increased the contents of labile fraction organic matter by 12–28%, dissolved organic matter by 7–21% and microbial biomass by 15-38%, thus stimulating enzyme activities. Trends in soil microbial respiration followed those of enzyme activity and were 20–34% higher in intercropping than in sole potato. Soil N balance was significantly influenced by legume intercropping (p<0.05) and ranged between -10.7 to -18.1 kg N ha<sup>-1</sup> for sole potato, 4.1 to 6.6 kg N ha<sup>-1</sup> for intercropping and 2.9 to 22.3 kg N ha<sup>-1</sup> for sole legumes. The residue mixture of potato and legume intercrops enhanced N mineralization with peak N release of 5 to 9 kg N ha<sup>-1</sup> occurring within 8 to 10 weeks of residue decomposition. This period coincided with the peak N uptake by potato (19.9 to 31.2 kg N ha<sup>-1</sup>) thus showing a close synchrony of N supply and potato N demand. These results provide a possible entry point to restoring the impoverished soil productivity in the smallholder potato farming systems and offer the possibility of potato expansion to midland agro-food systems.

**Keywords:** Agro-ecological zone; crop residue; intercropping; legumes; microbial activity; soil nitrogen balance; nutrient uptake; resource use efficiency; soil organic matter; soil temperature.

#### **CHAPTER ONE**

#### **GENERAL INTRODUCTION**

#### **1.1 Background information**

Potato remains one of the most important staple crops with a global production area of 19,302,642 ha, of which 1,439,329 ha is in sub Saharan Africa and 754,116 ha in East Africa (FAOSTAT, 2017). Potato is ready for harvest around 90–120 days after planting, which is 30–60 days before other staple crops such as maize, wheat and barley (CIP, 2008). This makes potato a hunger breaking and food security crop with a high market demand. In Kenya, potato is ranked the second most important food crop supporting approximately 800,000 farmers and further 2 million people along the value chain (CIP, 2008). The crop is mainly grown by smallholder farmers in the highland zones where temperature range of 15–20°C is optimal for tuberization (Muthoni, 2016).

Despite being a food security crop, potato is underperforming in Kenya and is not meeting its full potential for subsistence and commercial production. The average yield per hectare has stagnated at 7–15 t ha<sup>-1</sup> against attainable yield of 20–40 t ha<sup>-1</sup> (Parker et al. 2019). Deteriorating soil fertility, use of unimproved potato varieties and weak seed system partly account for the low potato yields. This situation is aggravated by upsurge in harsh climatic conditions which mainly results from persistent droughts, elevated heat stress, recurrent floods, increased soil erosion rates and disruption of traditional rainfall patterns (Karanja, 2013; Murakami et al. 2016). Globally, drought is projected to decrease yield potential of potato by 18–32% in the period of 2040–2069 (Hijmans, 2003). In the regions where a 1°C increase in mean temperatures has occurred, extreme heat waves, warmer night temperatures and hotter days have been experienced resulting in more than 50% potato yield reduction (Holden et al. 2003; Obidiegwu, 2015; MoALF, 2016). Such yield decline has been accompanied by devastated processes of canopy formation, tuberization and tuber

bulking (Dahal et al. 2019; Li and Zhang 2020). In the lowlands, only heat and water stress tolerant cultivars capable of setting tubers at elevated soil temperatures may successfully complete the tuberization process (Rykaczewska, 2015). Future impacts of climate change will exacerbate these effects if no adaptation and mitigation measures are ensured (Zemba et al. 2013; Parker et al. 2019).

The deteroriating soil fertility in smallholder potato farms is largely a consequence of potato cropping systems which is done mainly in pure stands (Muthoni et al. 2013; Nyawade et al. 2018a; Gitari et al. 2018b). This practice involves maximum disturbance of soil during the growth cycle which makes the soil loose and more prone to erosion (Nyawde, 2015). Potato also delays to establish protective cover after planting and does not yield sufficient surface mulch upon harvest which leaves the soil bare at critical times when rainfall intensities are high, thus exposing soil to erosion (Chow and Rees, 1994). Soil loss under potato monocropping systems in Kenya is therefore high, with up to 169 t ha<sup>-1</sup>yr<sup>-1</sup> having been recorded (Nyawade, 2015).

Further, the major soils in potato growing areas of Kenya are acidic with low soil organic matter (SOM) and high phosphorus sorption (Kibunja et al. 2018). These soils are mainly Nitisols, Planosols and Ferralsols and require high residue returns to replenish the low SOM contents, a quantity that cannot be attained by potato monocropping (Muchena and Gachene, 1988). Additionally, attaining reasonable yields of 20 t ha<sup>-1</sup> of fresh tuber removes 235 kg N, 80 kg P, 320 kg K and 15 kg S from the soil (Burton, 2018). These nutrients must be returned through fertilizer application, a strategy which resource constrained smallholder farmers may not sustain. Where farmers apply fertilizers, the quantity is often below the recommended rates and is mainly of the di-ammonium phosphate which most studies have shown to contribute to soil acidification

(Muthoni, 2016). The use of organic manure as alternative source of nutrients faces challenges due to the bulkiness of this input coupled with its limited availability and varied quality.

Innovative and sustainable strategies capable of enhancing resilience of potato production is therefore required. One such strategy is shifting from monocropping to intercropping (Nyawade, 2015; Gitari et al. 2018a, b). Intercropping systems may add 40–250 kg SOC ha<sup>-1</sup> from crop residue incorporation and biologically fix 20–140 N kg ha<sup>-1</sup> which benefits the companion non-legumes and crops grown in succession (Jarecki and Lal, 2003; Giller, 2015; Naresh et al. 2017). Similarly, potato-legume intercropping enhances light interception and nutrient use efficiency and stabilizes crop yield (Egbe, 2010; Giller, 2015).

#### **1.2 Statement of the problem**

The major potato growing areas of Kenya are particularly vulnerable to climate change, having already experienced increases in extreme weather conditions including persistent droughts and floods which jeopardize food security and weaken efforts to eradicate poverty (MoALF, 2016). Potato yields average 7–15 tons per ha, far below attainable yields of up to 40 tons per ha (FAOSTAT, 2017). These problems have been accelerated by lack of crop diversification and unavailability of heat and water stress tolerant potato varieties. This has resulted in intense risk of crop failure and loss of livelihood (Demo et al. 2015; Muthoni, 2016; Gitari et al. 2019; Parker et al. 2019). Where low rainfall amounts is coupled with high heat stress and yield devastation, farmers have altogether abandoned potato production (Nyawira, 2018).

In cases where the high demand for potato has continued to compel production amidst the extreme weather conditions, farmers have resorted to expand the area under potato cultivation as opposed to improving productivity per ha (Burke, 2017). Furthermore, where potato is grown, cultivation is mainly done in the highlands which are characterized by sloping terrains and up-

down slope operations (Gitari et al. 2019b). This makes soil erosion by water a major challenge (Fabeiro et al. 2001; Jemison, 2019). In these areas, sediment yield and nutrient losses due to erosion occur mainly at the onset of rains and later after potato harvest, a period which is characterized by low ground cover and high-intensity rainfall events (Gachene et al. 2019). The maximum soil disturbance during cycles of potato cultivation and harvest makes the soil loose and more prone to water erosion with values as high as 169 t ha<sup>-1</sup>yr<sup>-1</sup> having been recorded under potato monocropping (Nyawade, 2015; King, 2017; Gachene et al. 2019; Nyawade et al. 2019). The steep side slopes of the potato mounds concentrate water in the furrows, which detaches soil particles and transports them out of the field (Xing et al. 2011; Eshel et al. 2015) while harvest translocates the soil, making it more prone to both water and wind erosion (Dong et al. 2000; Auerswald et al. 2006; Tiessen et al. 2007). The nitrate remaining in the soil after the potato-growing season are lost to leaching or denitrification between crop harvest and the next planting season (Auerswald et al. 2006).

Thus, sustainable intensification capable of increasing productivity of water and land under the current climatic scenario must be ensured to support food, income and nutrition security without causing further damage to the environment. Legume intercropping is one of such intensification measures. However, intercropping integrating potato as the base crop is not commonly practiced in the subtropics and is currently not packaged in the good agricultural practice recommendations (Nyawade et al. 2020b). Mostly, the legumes are intercropped with cereals such as maize (Maitra et al. 2020; Kermaha et al. 2017), sorghum (Dereje et al. 2016; Gebremichael et al. 2020), millet (Mohammed et al. 2008), wheat (Monti et al. 2019) or cassava (Loi, 2000). Unlike legumes, cereals and cassava are competitor for water and nutrients especially nitrogen and phosphorus hence not suitable intercrop for potato. In addition, the compatibility within potato-legume intercrop systems and resource use is poorly understood by both the researchers and intended beneficiaries such that information dissemination and farmer adoption remain low (Nyawade et al. 2020b).

#### **1.3 Justification of the study**

Potato-legume intercropping exhibits enhanced capacity to control soil erosion thus making potato production more sustainable in the highlands. By optimizing soil temperatures and soil water contents through enhanced canopy overlap, legume intercropping can enable production of potato in the drier midland agro-food systems. Legumes can fix up to 400 kg N/ha/year into a form of nitrogen available for plant uptake under optimal field conditions (Giller, 2015; Sanginga et al. 2009). This nitrogen is either used by the legume itself or become available to other species in the system after decomposition of root nodules or plant material. When deep-rooted legumes are used, intercropping has capacity to diverge the nitrate leached beyond the active root zone of potato. This is because these legumes have increased capacity to explore N supplements in the subsoil thus allowing potato to take up N in the surface soil. Nitrogen released from legume biomass residues through mineralization and nitrification, can supply enough N needs of the following crop.

Due to increased diversity of crops in intercropping, this system provides insurance against total crop failure and cushions farmers against unstable market prices, and therefore averts risks to the farmer. Along with the promotion and extension of legume intercropping, potato yields per unit area is expected to increase tremendously thus increasing income and livelihood of farmers. The legumes are besides rich in proteins and provide nutritious diet to man and forage to livestock thus contributing to increased food security. Increase in water, nutrient, light and land productivity is expected under potato-legume intercropping. Where precipitation is the main source of soil moisture and is poorly distributed in time, intercropping potato into annual crops will reduce the competitive effect for soil moisture.

#### **1.4 Research objectives**

#### **1.4.1 Overall objective**

To increase productivity of soil, nutrients, light and water through systematic introduction of legumes in smallholder potato farming systems of Kenya.

#### 1.4.2 Specific objectives

- i. To determine the relationships between soil temperatures, soil water content and crop nutrient uptake (NPK) under potato-legume intercropping.
- ii. To quantify the short-term effect of potato-legume intercropping on soil organic matter fractions and microbial activity.
- iii. To assess the effect of potato-legume intercropping on soil nitrogen balance.
- iv. To quantify the radiation use efficiency and crop water productivity of potato-legume intercropping system.

#### **1.5 Hypotheses**

- i. Soil temperature, soil water content and crop nutrient uptake exhibit positive relationships under potato-legume intercropping.
- ii. Potato-legume intercropping increases the contents of labile SOM fractions and microbial activity.
- iii. Potato-legume intercropping has significant positive effect on soil nitrogen balance.
- iv. Potato-legume intercropping increases the crop water productivity and crop radiation use efficiency.

#### **1.6 Thesis outline**

This thesis is structured in chapter format as follows: chapter one presents the general introduction, problem statement, justification, objectives and hypotheses. Chapter two gives a detailed review of literature while chapter three describes the general methods which captures detailed site description and crop husbandry, and provides the conceptual framework developed for the study. Chapter four highlights the linkages between soil temperature, soil water balance and crop nutrient uptake. Chapter five presents the changes that occur in SOM fractions and microbial activity due to incorporation of crop residues and is cognizance of agro-ecological influence on SOM build-up and transformations. Chapter six quantifies the effect of legume intercropping and residue retention on soil nitrogen dynamics while Chapter seven assesses the productivity of potato-legume intercropping as relates to water use, land use and light use. Chapter eight synthesizes the results described in different chapters and provides the general conclusions, recommendations and policy implications.

#### **CHAPTER TWO**

#### LITERATURE REVIEW

#### 2.1 Potato production systems in Kenya

Potato is grown by smallholder farmers in the lower and upper highland zones of Kenya (CIP, 2008; Muthoni, 2016). Production is for both cash and subsistence and is primarily rainfed using family labor. The soil is tilled at onset of rains to loosen it prior to planting (CIP, 2008). Potato seeds are then planted and hilled at 14–21 days after emergence. Production system is mainly monocropping with only 5–12% of the producers intercropping potato with maize (*Zea mays* L.). This is despite the fact that potato has high uptake for NPK and returns little residue thus depleting the soil organic matter (Nyawade, 2015). For this reason, potato cultivation has been shown to contribute up to 60% of nutrient losses in Kenya highlands (Nyawade et al. 2019). Yields have remained low ranging from 7 to 15 t ha<sup>-1</sup> despite potential of 40 t h<sup>-1</sup> (Parker et al. 2019).

The major potato cultivar produced in Kenya is Shangi, with a few number of farmers slowly adopting the new varieties such as Unica, Wanjiku, Nyota, Chulu, Lenana, Konjo (NPCK, 2017; Schulte-Geldermann et al. 2012; Kaguora et al. 2015; Harahagazwe et al. 2018). Other varieties available but in limited supply include Kenya Mpya, Kenya Karibu, Kenya Sifa, Asante, Sherekea, Tigoni, Panamera, Rudolph, Markies, Mayan Gold, Taurus, Desiree, Destiny, Dutch Robijn among others (NPCK, 2017). Farmers prefer Shangi variety due to its short tuber dormancy coupled with limited knowledge about other alternative varieties.

Heat stress and low soil moisture content often occurs during early stages of growing seasons, and has resulted in huge yield losses especially with the Shangi cultivar (Karanja, 2013; Gitari et al. 2018b). As highlighted in CIP (2017) report, releases of heat and water stress tolerant potato cultivars such as Unica has shown great adaptability to erratic weather conditions with

greater potential to perform well under midland agro-ecological conditions. These cultivars are however not readily available in the local seed markets. Thus, diversification of the cropping systems is necessary to cushion the farmers from the risk of total crop failure.

#### 2.2 Relationships among soil temperature, soil water content and nutrient uptake

High ambient temperatures and increased soil moisture stress limit potato production in the lowlands and midlands despite the greater consumer demand for the crop (Burke, 2017). Appreciable decrease in tuber yield occurs if balance between soil temperature, soil moisture and crop nutrient uptake is not attained (Ferreira and Gonc alves, 2007). High soil temperatures at tuber initiation cause fluctuations in soil moisture status, resulting in tuber malformation and tuber growth cracks (Polgar et al. 2017). When the soil water content drops below the field capacity, uptake of water and nutrients by potato crop is suppressedy leading to irregular plant growth (Levy et al. 2013; Nyawade et al. 2018a).

As water moves within the soil profile, it directly affects soil temperature and movement of crop nutrients, thus influencing nutrient availability to crops (Li et al. 2017). Losses of nitrogen in deep percolating water have been reported under potato growing systems when soil moisture contents are excessive (Claudia et al. 2013) while nitrogen volatilization occurs with increasing soil temperatures (Courtney et al. 2005). High soil temperatures reduce export of nitrate to the shoots and limit allocation of assimilates to the roots (Burke, 2017). Similarly, elevated soil temperatures within the potato rhizosphere may induce moisture stress which in turn increases the total dry matter allocation to roots and stems at the expense of tubers (Thornton et al. 1996). Claudia et al. (2013) examined the nitrogen balance under potato cropping system and found that nitrogen recovery in tubers decreased as gravitational water increased which implies that nitrogen losses in potato growing fields may become the largest nitrogen sink when there is excess water. The major effects of high soil temperatures on potato growth are through its negative impacts on photosynthesis, respiration, reproduction, water relations and hormone production (Burke, 2017). The decrease in photosynthetic rate under high soil temperature conditions has been attributed to mesophyll conductance suppression and to stomatal closure under severe heat stress (Chaves et al. 2002). Under high-temperature conditions, tuberization is significantly inhibited and photoassimilate partitioning to tubers is greatly reduced (Lafta and Lorenzen 1995; Rykaczewska, 2017). Haulm growth is fastest in temperature range of 20–25°C whereas optimal range for tuberization and tuber growth is 15–20°C (FAO, 1996; Xiaolin et al. 2016). Though potato root growth can occur at a temperature range of 10–35°C, active root development occurs at soil temperature range of 15–20°C (Vandam et al. 1996; Rykaczewska, 2017).

High ambient temperatures occurring late in potato crop development stage is detrimental to tuber yield, particularly when haulms are lodged and the soil is exposed to incoming radiation (Alva et al. 2012). Raising night temperatures over the range of 0–20°C increases potato root length while night temperature increase to 25–27°C induces sharp reduction in the number and weight of tubers (Wilkinson and Davies, 2002; Kelly et al. 2017). High night temperatures above 20°C may induce production of gibberellic acid in the stolon tip, disrupting tuber growth (Wilkinson and Davies, 2002). Onset of tuberization may delay with night temperatures above 25°C, an observation associated with accelerated metabolism due to induction of specific inhibitory effects (Burke, 2017).

Potato requires 500–700 mm of water in a growing season (Sood and Singh, 2003) which is constrained by the shallow fibrous root system of the crop. This makes potato highly vulnerable to water deficit (Nyawade et al. 2018a; Aliche et al. 2018). Generally, water deficit modifies the activity of some enzymes and affects accumulation of sugars and proteins in potato plant. At stolon initiation, water deficit restricts stolon development and the number of stolons formed, and hinders root growth, nutrient uptake and plant response to nutrients (Rykaczewska, 2015, 2017). Water limitation at tuber initiation suppresses foliage and plant development, restricts tuber initiation and limits the number of tubers. At tuber filling stage, water shortage limits plant and tuber development, restricts tuber size, promotes tuber distortion and fastens leaf senescence. At maturity, water deficit limits relative tuber density and tuber size (Aliche et al. 2018).

Therefore, there is need to design cropping systems that can optimize the soil temperatures and increase soil moisture systems in the major potato producing areas. Under sole cropping, as much as 40% of rainfall can be lost through unproductive pathways such as evaporation and runoff (Kinama et al. 2011). In the drier lowland and midland agroecological conditions, combined use of legume intercropping and improved potato cultivars might exert potential synergy on tuber yield. Root system of potato mainly distributes within 0–40 cm soil layer (El-Abedin et al. 2017), but axial root of most legumes can reach up to 1 m soil depth (Ren et al. 2018) resulting in difference in spatial water requirement if grown in intercropping. Therefore, intercropping creates a spatio-temporal niche which allows the deep root system of the legume to colonize larger volume of the soil profile and exploit soil moisture from a larger surface area.

#### 2.3 Effect of cropping systems on soil organic matter fractions and microbial activity

Intercropping and rotation systems have great potential to increase carbon sequestration in the soil due to their ability to return larger quantity of crop residue into soil (Jarecki and Lal, 2003). Al-Kaisi et al. (2005) demonstrated that corn-soybean–alfalfa rotation could replenish soil nutrients in degraded soils due to their potential to enrich soil organic matter. Jagadamma et al. (2007) also showed that legume integration into cropping systems could sequester SOC and enhance soil organic matter build-up. Due to greater biomass production in corn than soybean, SOC concentration under continuous corn is normally higher than under corn-soybean rotation (Lal et al. 1997). West and Post (2002) showed that increasing rotation complexity increased nitrogen storage and SOC sequestration rate by 200 kg C ha<sup>-1</sup> yr<sup>-1</sup>. A meta-analysis result by McDaniel et al. (2014) found that cropping systems that incorporate high quality legumes have stimulatory effect on microbial community as measured by both microbial biomass carbon (MBC) and nitrogen (MBN).

The rate and extent of these benefits depend on which size fraction is affected (Kapkiyai et al. 1999; Pikul et al. 2007; Nyawade et al. 2018b). These fractions differ in their quality, quantity and turnover rates, and respond differently to soil management systems. The mineral associated organic matter is a more processed SOM characterized by higher density and includes well-decomposed microbial products forming organo-mineral complexes with the finer mineral particles (Six et al. 2002). This organic matter fraction is relatively recalcitrant (Camberdella and Elliot, 1992). Soil labile organic matter fractions such as light fraction, microbial biomass and dissolved organic matter consist mainly of recently incorporated organic materials such as microbial and plant-derived polysaccharides, fungal hyphae and roots (Six et al. 2002; Haynes, 2005; Naresh et al. 2017). These SOM fractions have rapid turnover rates and respond to soil management intervention more rapidly than the total SOM (Louise et al. 2016). The rate at which the labile SOM fractions decompose determines the release of nutrients to plants (Balesdent et al. 2000).

The microbial biomass is a living fraction of SOM consisting mainly of bacteria and fungi that decompose litter and crop residues (Naresh et al. 2017). This process releases nutrients that are made available for plant uptake (Smith and Paul, 1990). The microbial biomass is released to the soil in forms that can be taken up by plants when the soil microbes die (Louise et al. 2016). About 50% of microbial biomass is located in the surface 10 cm of soil profile (Naresh et al. 2017), an indication that most of the nutrient release occurs at this depth. Dissolved organic matter (DOM) is mostly a product of microbial metabolism and is highly used as substrate by the soil microbes (Jones et al. 2004; Roper et al. 2010). Any activity that rapidly alters the contents of DOM, MBC and microbial activity are regarded to be sensitive early indicators of long-term changes in SOM (Kapkiyai et al. 1999). The stable SOM fractions are product of microbial decomposition (Chabbi et al. 2009) thus reflecting long-term effect of microbial activity.

Microbial biomass just like any other SOM fractions is influenced by soil particle size distributions (Gupta and Germida, 1988; Camberdella and Elliot, 1992). The aggregate size affects the storage of SOM by occluding organic residues making them less accessible to degrading organisms (Six et al. 2000). The fresh residues provide nucleation centers for aggregation when returned to cropland and therefore promote formation of macro-aggregates which are the favorable habitat for soil microbes (Ghani et al. 2003). The high microbial activity due to enhanced substrate supply induces binding of soil particles into macro-aggregates, thus increasing aggregate stability. This results into fixation of unstable organic carbon, increasing the SOM concentration and carbon sequestration (Chen et al. 2009).

Quantification of soil microbial biomass activity is based on the rate at which respiration and enzyme hydrolyses by soil microbes occurs (Naresh et al. 2017). The overall activity or energy spent by indigenous soil microbial pool is measured by the basal respiration (CO<sub>2</sub> evolution) of a soil (Anderson and Domsch, 1990). If the basal respiration is related to corresponding microbial biomass size, a microbial metabolic quotient for CO<sub>2</sub> (qCO<sub>2</sub>) is obtained (Lupwayi et al. 1998). Just like the microbial biomass, qCO<sub>2</sub> also responds readily to changes in soil management and can therefore provide an effective early warning of soil quality deterioration. Perturbations such as high soil temperatures, soil water deficits and substrates depletion elevate  $qCO_2$  as the microorganisms become less efficient at conserving carbon (Anderson and Domsch, 1993). These processes may greatly vary with cropping systems, soil management practices and agro-ecological zones (Dinesh et al. 2009; Bichel et al. 2016).

#### 2.4 Effect of legume intercropping and crop residue on soil nitrogen content

Intercropping that includes legume cover crops can reduce soil N leaching due to their ability to take up residual nitrogen, converting it into crop biomass which is later mineralized (Constantin et al. 2012; Plaza-Bonilla et al. 2015). Velvet bean (*Mucuna pruriens*) was able to accumulate 160 kg N ha<sup>-1</sup> in 12 weeks when it was intercropped with maize (Sanginga and Woomer, 2009). Similarly, Eaglesham et al. (1981) found that N<sub>2</sub> fixed by cowpea was about 41 kg N ha<sup>-1</sup> in maize-cowpea intercropping system. Roots of the legumes can decompose and release nitrogen into the soil where it is made available to the subsequent crops (Giller, 2015). Legume cover crop integration into potato cropping systems can appreciably reduce nitrogen losses due to soil erosion thereby increasing efficiency of nitrogen use (Nyawade, 2015; Nyawade et al. 2019).

Nitrogen mineralization from crop residues is an important source of plant available N, often accounting for N taken up by a crop (Stevens et al. 2005; Osterholz, 2016). Positive net N mineralization occurs when inorganic N is produced in excess of microbial demands while negative net N mineralization occurs when microbial consumption processes exceed N mineralization in the absence of plant uptake (Schimel and Bennett, 2004). If rates of N mineralization are high and soil inorganic N pools are small, N mineralization could readily replenish the inorganic N pool. The N mineralization process varies with residue quality, particularly with respect to residue N concentrations, C, lignin and polyphenol contents (Palm and Sanchez, 1991; Tian et al. 1992; Handayamoto et al. 1994).

Soil moisture and soil temperature also determine N mineralization rates by affecting the soil microbial activity (Cassman and Munns, 1980; Vigil and Kissel 1995). The rate of N mineralization is greater in warmer and wetter ecosystems than in colder and drier ecosystems (Henriksen and Breland, 1999; Chantigny et al. 2002). Release and retention of available N in soil increases with rise in soil moisture and soil temperature up to a threshold and then declines (Baldock et al. 2018). Low temperature conditions lower microbial respiration rate thus delaying residue decomposition and subsequent release of N (Tian et al. 1997). The N mineralized can be lost through soil erosion, removal in harvested products, leaching, denitrification, and volatilization (Brye et al. 2003; Ma et al. 2015; Inselsbacher, 2010; Loecke et al. 2012). Denitrification occurs when N is lost through the conversion of nitrate to gaseous forms of N. These forms include nitric oxide, nitrous oxide and dinitrogen gas. This occurs when the soil is saturated and the bacteria use nitrate as an oxygen source (Courtney et al. 2005). Volatilization occurs when N is lost through the conversion of ammonium to ammonia gas, which is released to the atmosphere. The volatilization losses increase at higher soil pH and conditions that favor evaporation (e.g. hot and windy) (Courtney et al. 2005). Leaching occurs mainly where increased accumulation of nitrates is followed by a period of high drainage (Arregui and Quemada, 2006).

#### 2.5 Nutrient uptake by potato under legume intercropping

Nutrient uptake in plants is a function of capacity of the soil to supply adequate nutrients and the ability of plants to translocate the nutrients from roots to shoots (Baligar et al. 2001). This process is influenced by plant-soil interactions (Fageria and Baligar, 2005). In a study conducted to assess the effect of potato-legume intercropping systems on nutrient uptake and use efficiency in central Kenya, Gitari et al. (2018a) observed a significantly lower nutrient uptake in sole potato crop compared to potato-dolichos intercropping. Legume cover crops when integrated into potato cropping systems can control soil erosion, suppress weeds, conserve soil moisture, control diseases thereby increasing nutrient uptake, and use efficiency of potato (Xing et al. 2011; Nyawade, 2015; Giller, 2015).

Legumes such as lupin (*Lupinus albus* L.), chickpea (*Cicer arietinum* L.) and faba bean (*Vicia faba* L.) induce root acidification and exudate carboxylates such as citrate, malate, malonate, citrate and oxalate which dissolve and enhance P acquisition (Hinsinger, 2001; Vance et al. 2003; Wang et al. 2010). The exuded organic acids precipitate Ca, Fe and Al ions responsible for P fixation depositing them on the root surface. The P mobilized in excess of plant requirement is utilized by the less P efficient crops grown in rotation and/or intercropping. However, the processes underlying these observations remain poorly investigated, but speculated to be due to carboxylate exudation, formation of cluster roots, secretion of organic P-hydrolyzing phosphatase and increased biomass allocation to roots (Hinsinger, 2001; Vance et al. 2003; Hinsinger et al. 2011; Schneider et al. 2019). In a greenhouse experiment, intercropping potato with either chickpea or white lupin enhanced P acquisition by intercropped potato (Nyawade et al. 2020 c, d). The legumes secreted enzyme phosphatase and exuded citrate, malonate and malate, the concentrations of which were greater in intercropping and resulted in enhanced P solubilization.

Nutrient uptake by potato crop may differ with the nutrient under consideration. Higher uptake of Cu and Fe by potato under legume intercropping systems has been attributed to the low uptake of these elements by legumes (Naresh et al. 2017). Unlike calcium, potassium is highly mobile in the phloem making its utilization more efficient as it is readily redistributed from older leaves to younger shoot. As a result, symptoms of K deficiency in potato first appear in the older leaves. Potato crop has poor uptake and use efficiency for phosphorus as the crop has poorly developed P diffusion paths (Thorup-kristensen et al. 2001, 2003; Constantin et al. 2012). Legumes

with higher nitrogen fixation potential such as as soybean may increase nutrient uptake by potato compared to poor fixers such as common bean (*Phaseolus vulgaris* L.), when grown in intercropping with potato (Giller, 2015).

#### 2.6 Crop water productivity and light use efficiency

Under optimum growing conditions, biomass productivity of different crops can be described by the amount of solar radiation intercepted by the foliage and the efficiency with which the amount of water is converted to plant dry matter (Monteith, 1977). The productivity of water by crops may vary with cropping systems. Singh et al. (2013) observed that the total water used in intercropping system was almost same as for sole crops, but yields were increased with intercropping, thus crop water productivity was higher than in the sole crops. Increased soil moisture content under legume intercropping increases the ability of soils to supply nutrients which in turn increases the plant nutrient uptake (Alva et al. 2012).

Light interception and use efficiency processes are governed by the amount of solar radiation, leaf area index (LAI), soil temperature, soil moisture contents, soil nutrient levels and crop development stage. As LAI increases, more radiation is intercepted per unit ground area resulting in increased nutrient assimilation (Ewert, 2004). Increase in LAI to optimal levels is critical for maintaining high photosynthetic rates and crop yield (Xiaolei and Zhifeng, 2002). If the index is too low, not enough light will be absorbed and if too high, lower leaves will not receive enough light and will thus be a liability (Brintha and Seran, 2009). Soil moisture and nutrient levels affect foliage expansion and radiation capture (Stone et al. 2001; Inman-Bamber, 2004; Massignamet al. 2011).

Radiation use efficiency (RUE) has been reported as a stable parameter for many crops (Hughes et al. 1987), but variability in this parameter has too been shown (Sinclair and Muchow,

17

1999). This is because sunlight intercepted is not utilized with similar efficiency by all crops (Haxeltine and Prentice, 1996). There are therefore, clear differences in radiation use efficiency between crop species, particularly between those with C3 and C4 photosynthetic pathways (Kinama et al. 2011). The C4 crops have been shown to have higher RUE compared to C3 plants while combining C4 and C3 in intercropping systems increases RUE due to their additive effect on soil moisture extraction (Kinama et al. 2011). Kemanian et al. (2004) demonstrated that intercropping cereals with legumes results into higher photosynthetically active radiation (PAR) interception than when produced in their pure stands. This is because of differences that exist in vertical foliage arrangement and canopy architecture of crop varieties.

The net benefit derived from enhanced radiation interception with intercropping has largely been inferred to alteration of crop architecture and geometry (Acreche et al. 2009; Baldé et al. 2011), both of which have direct relationship with the interception efficiency of direct and diffuse irradiance (Valladares and Pearcy, 2000). Reddy et al. (2003) found that intercropping millet with groundnut was 28% more efficient in conversion of light intercepted to biomass than their monocrops, an observation attributed to 30% greater leaf area index development by the intercrop. Similarly, Ong and Black (1992) observed that intercropping pearl millet with groundnuts and pigeon pea produced 15% higher radiation interception compared to sole pigeon pea and twice that intercrops were to the sole crops may be due to spread of light over greater leaf area and its more efficient distribution during early stages of growth (Addo-Quaye et al. 2011). Therefore, factors related to genotype selection and populations, which remain grey to resource-poor farmers, are important in the distribution of photosynthetically active radiation within the intercrop canopy.
#### 2.7 Considerations for a sustainable potato-legume intercropping

A sustainable intercropping system must consider the base crop and its husbandry needs, in this case potato. The legumes should exhibit minimum competition with potato crop for nutrients, solar radiation, water and space and the peak nutrient demand should not coincide with that of potato (Nyawade et al. 2020). The intercrops should be integrated as per the local need and farmer preferences. The system must improve potato tuber yield if farmers are to take it up (Nyawade et al. 2019). Preferably, the legume should be used for human food or livestock feed and should have capacity to enhance soil productivity (Nyawade et al. 2020). If potato is used as food crop, the legume can be used as a cash crop or fodder to ensure sustenance and cash income.

Complementarities and facilitation should also exist between the crop components to minimize competition, but to increase mutual benefits (Gitari et al. 2020). Generally, potato takes up nutrients from the upper soil layers while most legume intercrops develop robust tap root systems that explore the soil matrix more extensively and absorb water and nutrients from deeper soil layers (Ren et al. 2018). In the highlands, lucerne and desmodium intercrops were able to increase their rooting zone beyond 1 m thus providing mechanism for divergence of the nitrate leached beyond potato root zone (Nyawade et al. 2020).

The intercrops must be chosen based on the prevailing agro-ecological zone (AEZ). In moisture-stressed zones of semi-humid areas, the influential factors that determine the benefits of intercrops are water and to a lesser extent, nutrients (Ren et al. 2018). In contrast, in humid zones nutrient deficiency and light because of the shading effect are more profound (Burke, 2017). Soil erosion is prevalent in these zones and thus legume cover crops might be prioritized (Schulte-Geldermann, 2013). The choice of potato cultivar is very important and is related to the agro-ecological conditions (Hellmuth, 2019).

Seedbed preparation should be done as per the need of potato, taking into considerations husbandry needs of the legume intercrops (Rana and Rana, 2011). Modifying plant spacing to increase the distance between adjacent potato and the intercrops has been considered to reduce the shading of potato by bush bean varieties and to facilitate potato harvest and hilling without damage to the legumes (Nyawade et al. 2018). However, the yield consequences of these modifications have not been conclusively tested. Nyawade et al. (2018) devised a pre-hilling practice that entailed planting potato plant in pre-mounded earths with just a slight piling up of the soil at 2 weeks after potato germination and 2 weeks later after first hilling. This practice minimized soil disturbance caused by repeated hilling, optimized soil temperature, and enhanced moisture distributions within the soil.

Temporal and spatial arrangement of the intercrop components become important as the net benefit derived from enhanced light interception in intercropping is inferred to alteration of crop architecture and geometry (Acreche et al. 2009), both of which have direct relationship with the interception efficiency of direct and diffuse irradiance (Valladares and Pearcy, 2000). In most reports, planting potato and legumes in 2 rows of potato alternating with 2 rows of legume (Fig 3) resulted in a higher potato tuber yield (Ren et al. 2018; Nyawade et al. 2020). This system improves radiation interception through development of more complete ground cover. Woomer et al. (2004) related the superior crop yields with 2:2 row-intercropping arrangement to advantages in root distribution and reduced belowground competition. When potato and legumes exhibit overlap in resource use, staggering planting time may separate rooting systems in time and result in yield advantage.

# **CHAPTER THREE**

# **GENERAL MATERIALS AND METHODS**

# 3.1 Description of the study sites

Field trials were conducted during the dry season of 2016 and wet and dry seasons of 2017 and 2018 in Kirinyaga (Kianyaga), Nairobi (Kabete) and in Nyandarua (Ol Kalou) (Fig. 3.1).



Figure 3.1: Map of study sites selected for establishment of the trials.

The region forms part of Mount Kenya's potato growing belt and borders the Great Rift Valley to the East (Muchena and Gachene, 1988; Kibunja et al. 2018). Potato production is mainly rainfed and depends primarily on family labor. The crop is grown both in the dry season that extends from October to December and in the wet season extending from March to June (Jaetzold et al. 2012; MoALF, 2016). Soil and climatic characteristics of these sites are described in Table 3.1.

Paramatar	Site							
Farameter	Kirinyaga	Kabete	Nyandarua					
GPS coordinates	0°29'35.71"S 37°20'55.29"E	1°14'45.00"S 36°44'19.51"E	0°14'39.08"S 36°17'18.99"E					
Agro-ecological zone	Upper midland	Lower highland	Upper highland					
Altitude (masl)	1552	1894	2552					
Mean annual rainfall (mm)	900	1000	1500					
Mean annual max temp (°C)	27.3	24.4	20.5					
Evapotranspiration (mm)	1442	1152	1021					
Soil type	Rhodic Ferralsol	Humic Nitisol	Ferric Luvisol					
Soil description	Well drained, shallow to very deep, dark reddish brown silty loam with moderate P sorption.	Dark reddish brown clay-loam soils with low to high SOM and an acid top soil. Have a high P sorption.	Moderately to well drained, deep to very deep, dark brown to very dark red brown firm clay to silt loam. Have moderate P sorption.					

Table 3.1: Climatic and soil descriptions of the experimental sites.

Source: Jaetzold et al. (2006, 2012); Sombroek et al. (1982)

# **3.2 Treatments**

The treatments constituted potato grown in monocropping and in intercropping with two legumes

as follows:

- 1. Potato (Solanum tuberosum L.) monocrop
- 2. Dolichos lablab (Lablab purpureus L.) monocrop
- 3. Lima bean (Phaseolus lunatus L.) monocrop
- 4. Potato-dolichos lablab intercrop
- 5. Potato-lima bean intercrop

"Unica" (CIP 392797.22) potato cultivar was used in this study. This cultivar is heat and water stress tolerant making it well adapted to wide range of agro-ecological conditions (CIP, 2008; NPCK, 2017). The Rongai cultivar of *Lablab purpureus* L. (KT 003) is a species in the family *Fabacea* that produces white flowers and light brown seeds and has a high biomass production.

Lima bean cultivar was of improved bush Burpee with ability to produce high biomass under wide range of climatic conditions.

# **3.3 Conceptual framework**

Figure 3.2 shows the conceptual framework developed for this study.



Figure 3.2: Conceptual framework developed for the study. SOM, Soil organic mater, DOM, Dissolved organic matter, LFOM, light fraction organic matter, SWC, soil water content.

Soil temperature, soil water content and cropping systems were the major independent variables explaining the changes in nutrient uptake, microbial activity, nutrient balance, soil organic matter contents, crop water productivity, radiation use efficiency and potato equivalent yields. The variables exhibited interrelationships with soil biochemical processes such as leaching, volatilization, decomposition, nutrient cycling and nutrient transformations.

# **3.4 Experimental design and layout**

The treatments were laid out in a randomized complete block design with four replications. The plots measured 4.75 m long by 3 m wide, and were separated by 1 m path and 1.5 m distance from one block to the other.



Figure 3.3: Schematic illustration of inter row spacing deployed for sole cropping and intercropping systems.

An intercrop plot accommodated 8 rows of crops (4 potato plants and 4 legume plants) planted in 2 rows of potato alternating with 2 rows of legumes (Figue 3.3). Rows were 0.75 m from potato to potato, 0.75 m from potato to legume, and 0.5 m from legume to legume. Potatoes were planted at a depth of 0.1 m on pre-hilled ridges as this practice could optimize soil temperatures and enhance soil moisture distribution (Nyawade et al. 2018a).

# **3.5 Agronomic practices**

Fertilizer application was based on soil analysis and crop nutrient requirements (CIP, 2008). In Nyandarua, basal application of 50 kg N ha<sup>-1</sup>, 90 kg P ha<sup>-1</sup>, 100 kg K ha<sup>-1</sup> and topdressing with 50 kg ha<sup>-1</sup> calcium ammonium nitrate (CAN). In Kirinyaga, 60 kg N ha<sup>-1</sup>, 80 kg P ha<sup>-1</sup>, 90 kg K ha<sup>-1</sup>, and 70 kg ha<sup>-1</sup> CAN topdressing was applied, while in Kabete, 45 kg N ha<sup>-1</sup>, 100 kg P ha<sup>-1</sup>, 100 kg K ha<sup>-1</sup> and 55 kg ha<sup>-1</sup> CAN topdressing was applied. Topdressing was done 15–25 days after potato emergence depending on the general soil moisture conditions. The fertilizer applied at planting was NPK 15-9-20 while topdressing was done with CAN-26%. Legumes received triple superphosphate (TSP– 46%) at rate of 50 kg P ha<sup>-1</sup>.

Weeding was performed by hand hoeing at 15–25 days after potato emergence when hilling the soil around the potato vines' to about 20 cm height. The legumes were sprayed with Duduthrin 1.7 EC (Lambda-cyhalothrin 17.5 g L<sup>-1</sup>) alternating with Bestox 100 EC (Alpha-cypermethrin 50 g L<sup>-1</sup>) to control aphids while potato crops were sprayed with Ridomil Gold MZ 68WG (Mefenoxam 40 g kg<sup>-1</sup> + Mancozeb 640 g kg<sup>-1</sup>) to control late blight disease. Potatoes were harvested at maturity by uprooting the tubers using hand hoes while legumes were harvested by plucking out the pods. Crop residues in each plot were incorporated into the soil at start of the subsequent seasons.

#### **3.6 Soil sampling and soil physico-chemical analyses**

Soil samples were taken from each experimental plot at 0–120 cm depths just before planting and at the end of the season using soil auger. For each plot, soil samples were drawn from the inner rows and bulked to give a composite sample. Once in the laboratory, the soils were passed through 2 mm sieve, analyzed for gravimetric moisture content, and stored at 4°C. Soil pH was measured in 1:2.5 soil to water suspensions, soil texture by hydrometer method (Gee and Bauder, 1986), total N by modified Kjeldahl method (Bremner and Mulvaney, 1982) and organic carbon by modified Walkley and Black method (Nelson and Sommers, 1996). Extraction of soil samples for analysis of available P and extractable K was done using Mehlich 1 procedures (Mylavarapu et al. 2002) and determined using UV–vis spectrophotometer (Murphy and Riley, 1962) and flame photometry (Jackson, 1967), respectively.

#### 3.7 Data management and analyses

The statistical analyses were performed in R software, version 3.4.2 (R Core Team, 2017) using linear mixed model analysis of variance (ANOVA). The mixed-models were defined with the R-package lme4 (Bates et al. 2015) using likelihood based inference (Demidenko, 2013). Treatments and their interactions were considered as fixed factors while season and blocks (replications) were regarded as random factors. Significant interactions between fixed factors were detected using likelihood ratio tests for generalized linear mixed models. Whenever treatment effects were significant, the response variable was subjected to Tukey's Honest Significant Difference (HSD) test ( $p \le 0.05$ ) for multiple means comparisons with the agricolae package. Differences between treatments are indicated by different letters. Whenever necessary, data were either log transformed or an unequal variance model used to meet assumptions of normality and equal variance.

#### **CHAPTER FOUR**

# RELETION OF SOIL TEMPERATURE, SOIL WATER BALANCE AND NUTRIENT UPTAKE UNDER POTATO-LEGUME INTERCROPPING

# Abstract

Expansion of potato production to the subtropical midlands will only be possible if the joint effect of water deficit and high soil temperatures is uncoupled. Potato (Solanum tuberosum L.) was intercropped with lima bean (Phaseolus lunatus L.) or dolichos (Lablab purpureus L.) in three agro-ecological zones of Kenya; upper midland-UM (1552 meters above sea level (masl)), lower highland-LH (1894 masl) and upper highland-UH (2552 masl). The objective was to evaluate intercropping effect on evapotranspiration, soil water balance, soil temperature and their interrelationships with crop nutrient (NPK) uptake. Crop cover, evapotranspiration, soil evaporation, soil temperatures, volumetric soil water contents and crop water balance were quantified from crop emergence to harvest. Peak crop cover was greater in intercropping (62–95%) than in sole potato (48–83%) except in the UH under potato-dolichos (74%). The amount of soil water retained under different cropping systems decreased exponentially with increasing soil temperature; the relationship was stronger with sole potato ( $R^2 = 0.81$ ) than with the intercrops  $(0.54 \le R^2 \le 0.68)$  and weakest with sole legumes  $(0.45 \le R^2 \le 0.58)$ . Soil temperatures were 3– 7°C lower under intercropping compared to sole potato. Intercropped potato attained tuber yield of 0.45 to 0.86 kg tuber weight per plant compared to 0.26–0.68 for sole potato. The tap root systems by legume intercrops recaptured the water percolated below potato root-zone, converting it into crop biomass. Legume intercropping optimizes the soil temperatures and soil water balance thus enabling potato production in the drier midland agro-food systems.

Keywords: Soil moisture; intercropping; nutrient uptake; nutrient use efficiency; crop cover.

# 4.1 Introduction

The current climate change scenarios of decreasing precipitation and increasing temperatures point to a devastation of potato yield across the subtropical highlands. In Kenya for instance, a moderate (1°C) increase in mean temperatures over the last 37 years has caused potato yield to drop by up to 50% (MoALF, 2016). The adverse effect of increasing temperatures on potato growth and tuber yield becomes particularly important when it is coupled with water limitations (Ferreira and Gonc alves, 2007). Increase in soil temperatures to values above 15–20°C hinders tuberization process (Rykaczewska, 2015; Xiaolin et al. 2016) and causes damage to potato roots thus hindering their capacity to take up soil water and nutrients (Li et al. 2017; Onwuka and Mang, 2018). In cases where seasonal rainfall amounts fall below the range of 500–700 mm, tuber formation and filling is devastated (Sood and Singh, 2003; Pavlista, 2015).

Introduction of simple technologies capable of optimizing soil temperatures and soil water balance in Kenya and within the subtropics therefore becomes necessary. Intercropping due to its ability to achieve greater complementarity of resource utilization through use of crop mixtures having different rooting ability, canopy structure, height, and nutrient requirements has been recommended (Burke, 2017; Gitari et al. 2018a, b; Gitari et al. 2019). Intercropping that integrates legumes improves soil fertility through biological nitrogen fixation, increases soil conservation through greater groundcover, provides better lodging resistance for crops, reduces pest and disease incidence and provides insurance against crop failure or unstable market prices (Jensen, 1996, Giller, 2015; Nyawade et al. 2020a). Legume residues provide nucleation useful in formation of macro-aggregates which are the favorable habitat for soil microbes (Ghani et al. 2003; Liang, 2013). However, when the intercrops are not matched with the appropriate AEZs, and do not adjust

to specific spatial arrangements, yield penalties on the main crops ranging from 2–18% has been observed (Nyawade et al. 2020).

Despite the good benefits demonstrated by legume intercropping, this practice is highly neglected among smallholder potato farmers in the tropics. In Kenya, only about 5–12% of the farmers practice potato-legume intercropping (Muthoni et al. 2013; Gitari et al. 2018a). The intercrop commonly used in potato cropping systems is maize (*Zea mays* L.), the system of which cannot fix the atmospheric nitrogen and competes with potato for water, nitrogen, phosphorus and potassium. Further, lack of scientific evidence on the role of potato-legume intercropping on yield, crop water productivity and nutrient dynamics reinforce bottlenecks that prevent effective uptake of this technology in smallholder potato farming systems.

In this regard, a number of field studies have been initiated across different agro-ecological zones of Kenya to test the effect of potato-legume intercropping on crop water productivity, light use efficiency, nutrient use efficiencies and net returns. When potato was planted at the same time with *Lablab purpureous and Phaseolus lunatus*, the system reduced soil erosion by 30–60%, increased crop water productivity by 20–56% and diverged the nitrate leached below potato root-zone (Nyawade, 2015; Gitari et al. 2018a,b; 2019; 2020; Nyawade et al. 2018a, b; 2019a, b, c; 2020a). However, proper understanding of legume intercropping on yield is only plausible if relationships among evapotranspiration, soil temperature and soil water storage is understood. This study presents the changes in soil water storage, evapotranspiration and crop water balance, and reports crop yield and nutrient uptake increase due to legume intercropping.

#### 4.2 Materials and methods

## 4.2.1 Study site

Field experiments were conducted during the short and long rains of 2017 and 2018 in Kirinyaga (0°29'35.71"S and 37°20'55.29"E), Kabete (1°14'45.00"S and 36°44'19.51"E) and in Nyandarua (0°14'39.08"S and 36°17'18.99"E). Detailed description of the study sites are described in section 3.1. Measured soil properties (0–120 cm depth) before the experiment are provided in Table 4.1. According to the ratings by Gough and Wolf (1996), SOC, total N, and available P were below the optimal levels for potato growth irrespective of the study site.

Table 4.1: Initial soil	physical an	nd chemical	properties.
-------------------------	-------------	-------------	-------------

Agro- ecological zone	Soil depth	Clay	Silt	Sand	Textural class	Pb	Θwp	θfc	Θs	Ks	pН	SOC	Ν	Р	Κ
	(cm)	%					gcm <sup>-3</sup>	c	cm cm <sup>-3</sup>	m	m h <sup>-1</sup>	9	%	% p	pm cmol /kg
Upper midland	0-30	24.5	33.3	42.2	CL	1.19	0.07	0.21	0.51	55.13	4.99	1.82	0.13	33.30	1.23
	30-60	24.2	36.9	38.9	CL	1.24	0.04	0.19	0.52	49.21	4.99	1.04	0.23	23.40	1.33
	60-90	28.9	29.8	41.3	CL	1.34	0.03	0.25	0.58	40.33	4.93	0.88	0.11	24.40	1.19
	90-120	23.8	32.4	43.8	CL	1.35	0.03	0.28	0.55	32.22	4.92	0.33	0.09	20.20	1.09
Lower highland	0-30	49.7	22.5	27.8	С	0.99	0.09	0.38	0.41	33.33	5.11	2.06	0.19	24.40	1.13
	30-60	49.2	24.2	28.9	С	1.04	0.06	0.37	0.41	27.56	5.14	1.56	0.11	18.20	1.16
	60-90	50.1	24.2	25.7	С	1.14	0.04	0.36	0.49	23.28	5.16	0.98	0.06	17.70	1.11
	90-120	51.3	24.8	23.9	С	1.19	0.05	0.38	0.48	14.98	5.20	0.42	0.02	16.60	1.00
Upper highland	0-30	38.3	56.1	5.6	SC	0.97	0.09	0.37	0.43	29.89	5.21	3.09	0.22	16.60	1.16
	30-60	36.9	58.4	4.7	SC	1.00	0.08	0.39	0.42	26.87	5.22	2.34	0.24	17.90	1.15
	60-90	34.6	59.5	5.9	SCL	1.08	0.03	0.35	0.45	18.32	5.26	1.92	0.11	15.50	1.03
	90-120	33.9	57.9	8.2	SCL	1.11	0.04	0.36	0.41	9.04	5.28	0.98	0.09	14.90	1.02

C, CL, SC, SCL denote clay, clay loam, silt clay and silt clay loam respectively. Critical levels for potato growth: SOC, 2.5%, N, 0.2%, K, 1.0 cmol/kg, and P, 30 ppm (Gough and Wolf, 1996). Soil bulk density (pb); soil water content at permanent wilting point ( $\theta$ wp), at field capacity ( $\theta$ fc) and at saturation ( $\theta$ s); saturated hydraulic conductivity (Ks); soil organic carbon (SOC); total nitrogen (N); available phosphorus (P) and exchangeable potassium (K).

#### 4.2.2 Experimental layout and crop husbandry

The trials were laid out in a randomized complete block design with four replications. The treatments comprised of potato (*Solanum tuberosum* L.) grown alone or intercropped with either lima bean (*Phaseolus lunatus*, improved bush Burpee cv.) or dolichos (*Lablab purpureus* L.,

brown Rongai cv.) and pure stands of legumes. Detailed description of experimental design, crop husbandry and agronomic activities are described in section 3.4 and 3.5.

#### 4.2.3 Climatic data

The daily climatic data used to compute crop evapotranspiration (air temperature, wind speed, actual vapor pressure, solar radiation and daylight hours) were obtained from agrometeorological stations located at about 200–500 m from the study sites.

#### 4.2.4 Soil sampling and analyses

Soil sampling was conducted at vegetative (about 45 days after planting (DAP), and tuber bulking stages (about 75 DAP) of potato development, and at the end of potato growing period. Ten (10) soil replicates were taken within and between the crop rows of each plot with a 14 mm inner-diameter soil piston auger for every 30 cm interval to a depth of 120 cm. This depth interval represented the maximum possible root penetration zone for the crops used in this study. The samples were mixed into a composite for each depth and plot and stored at 4°C until analysis. Determination of total N content was done following the method of Keeney and Nelson (1982). Extraction of soil samples for analysis of available P and extractable K was done using Mehlich 1 method (Mylavarapu et al. 2002).

#### 4.2.5 Determination of soil water content

Soil water content of each plot was measured for every 30 cm interval in soil profile of 0-120 cm using tensiometers (0-100 kPa). The tensiometers were located between the plant rows. Prior to their installation, the tensiometers were calibrated using gravimetric soil moisture measurements taken at soil depth 0-30 cm. The gravimetric data were plotted against tensiometer readings in a scatter plot and a regression equation between the two methods was established. The values of matric potential measured with the tensiometers were used to plot soil moisture characteristic curve from which a relationship used to compute volumetric soil water content was derived. The average soil moisture content of a given treatment and layer in each period was calculated using Eq. (4.1).

$$SM_{ij} = \frac{1}{4*N_y} \sum_{m=1}^{4} \sum_{k=1}^{N_y} SM_{ijmk}$$
(4.1)

where,  $SM_{ij}$  is the soil moisture of ith treatment at jth soil layer;  $SM_{ijmk}$  is the actual value of soil moisture (m) measurement at each time;  $N_y$  is the total number of measurements of soil moisture over four seasons of the study and 4 is the number of replications.

# 4.2.6 Estimation of crop cover

A point frame was used to quantify the crop cover. The frame consisted of a single row of 10 pins spaced 10 mm with tripods measuring 2 m in height (Coxson and Looney, 1986). Sampling was started from 7 days after planting and continued at 2 weeks interval until harvest. Crop cover was estimated using Eq. 4.2.

$$Crop \ cover \ (\%) = \frac{Number \ of \ pins \ that \ hit \ plant \ leaves}{Total \ number \ of \ pins \ (10)}$$
(4.2)

# 4.2.7 Estimation of root water uptake

Root water uptake at a given soil depth was described by the extended Prasad model (Eq. (4.3). The model assumes that root water uptake is limited by a function of potential transpiration which decreases linearly with soil depth (Feddes et al. 1978; Hayhoe and De Jong, 1988). Estimations were made for every 30 cm interval in soil profile of 0–120 cm.

$$S_{max}(z) = \frac{RLD(z)}{\int_{-RD}^{0} RLD(z)dz} T_p$$
(4.3)

Where  $S_{max}(z)$  is the maximum root water uptake at a given soil depth (z), RLD (z) is the root length density at depth z, RD is the rooting depth (cm), and Tp is the potential transpiration rate. Rooting depth was taken as the depth at which more than 70% of the root volume was found (Mahdian and Gallichand, 1996) i.e. 30 cm for potato, 60 cm for lima bean and 90 cm for dolichos. For intercropping, the average rooting depth for the legume and potato was taken. Soil depth here refers to soil layers in which root water uptake was measured i.e. 30 cm, 60 cm, 90 cm and 120 cm. Potential transpiration (Tp) was calculated using Eq. (4.4).

$$T_p = ET_p - E_p \tag{4.4}$$

where evaporation (Ep) was estimated using Eq. (4.5) and reference evapotranspiration (ETp) using Eq. (4.6) (Allen et al. 1998).

$$E_p = ET_p exp^{-0.6LAI} \tag{4.5}$$

$$ET_p = \left[\frac{0.408\Delta(R_n - G) + \gamma \frac{900}{T + 273}u_2(e_s - e_a)}{\Delta + \gamma(1 + 0.34u_2)}\right] * KC$$
(4.6)

LAI refers to the measured leaf area index, Rn; net radiation at the crop surface [MJ m<sup>-2</sup> day<sup>-1</sup>] estimated using method outlined by Allen et al. (1998), G; soil heat flux density [MJ m<sup>-2</sup> day<sup>-1</sup>] which due to its small magnitude beneath the reference surface was considered negligible, T; mean daily air temperature at 2 m height [°C], u<sub>2</sub>; wind speed at 2 m height [m s<sup>-1</sup>], es; saturation vapor pressure [kPa], ea; actual vapor pressure [kPa], es–ea; saturation vapor pressure deficit [kPa], $\Delta$ ; slope vapor pressure curve [kPa °C<sup>-1</sup>],  $\gamma$ ; psychrometric constant [kPa °C<sup>-1</sup>]. Crop coefficient (Kc) was calculated using Eq. (4.7) for sole cropping and Eq. (4.8) for intercropping.

$$\frac{ET_a}{ET_o}$$
 (4.7)

$$K_c = \frac{f_1 h_1 k c_1 + f_2 h_2 k c_2}{f_1 h_1 + f_2 h_2} \tag{4.8}$$

where ETa is the actual evapotranspiration estimated using Eq. (4.11) and ETo is the reference evapotranspiration (Eq. (4.6) less Kc factor), f1 and f2 are fractions of the soil surface planted by potatoes and legumes in an intercropping i.e. 0.5, h1 and h2 are the height of potatoes and legumes, respectively; and *K*c1 and *K*c2 are crop coefficients for potatoes and legumes in monoculture.

## 4.2.8 Estimation of soil water balance

Soil water balance was estimated using Eq. (4.9) (Rosa et al. 2012).

$$D_{r,i} = D_{r,i-1} - (R - RO)_i - I + ET_i + \Delta SWC_i$$
(4.9)

where Dr,i and Dr,i-1 are the root-zone moisture depletion at time i, Dr,i-1 is the root zone depletion at previous measurement, R is rainfall at time i, RO is runoff from the soil surface at time i, I is net irrigation depth that infiltrates into the soil at time i, ET is the actual crop evapotranspiration at time i, Dp is deep percolation beyond active root-zone at time i,  $\Delta$ SWC is change in soil water content at time i (difference between final and initial SWC). All terms are expressed in mm. Surface runoff was quantified by flow meters installed at outlets of the plots. Micro-lysimeters installed to a depth of 30 cm were used to estimate soil surface evaporation. Deep percolation (Dp) was estimated as residual term of potato root-zone water balance (Eq. (4.10)) (Bethune et al. 2008).

$$DP = I + R - RO - ET_a - \Delta SWC \tag{4.10}$$

While actual evapotranspiration (Et<sub>a</sub>) was estimated using Eq. (4.11) (Rosa et al. 2012):

$$ET_a = I + R - RO - DP - \Delta SWC \tag{4.11}$$

where all terms of soil water balance are as earlier defined. Capillary rise was not considered in this equation since the water tables across the study sites were deeper than 25 m (Gachene et al. 1989; Karuku et al. 2014).

## **4.2.9** Estimation of root length density

Root length density (RLD) of each cropping system was measured at vegetative and tuber bulking stages. Measurements were taken for every 30 cm interval in soil profile of 0-120 cm using 15 cm x 10 cm x 10 cm metal cores (Bohm, 1979). Roots were excavated at the inter-rows with additional sampling at stem base (about 5 cm from the stem) to take into account roots not extending to the inter-rows. Soil cores containing the roots were placed in a bucket of water and gently agitated to break down larger soil particles, and to remove debris and dead roots. This mixture was sieved through 2 mm mesh placed in a tub of water to wash away fine soil particles attached to the roots. Roots were floated on water in 40 cm  $\times$  40 cm glass tray and scanned using Epson Expression 1680 Scanner (Seiko Epson Corp., Tokyo, Japan) (Fig. 4.1). Root length density was then analyzed using WinRHIZO Root Analyzer System (Regent Instruments Inc., Quebec, Canada) (Eq. (4.12).

```
Root length density (cm \ cm^{-3}) = \frac{Root \ length \ (cm)}{Soil \ volume \ of \ corresponding \ depth \ (cm^3)} (4.12)
```



Figure 4.1: Scanning (a) and processing (b) of roots using Epson Expression 1680 Scanner.

# 4.2.10 Nutrient uptake

At potato flowering stage, 6 plants were randomly sampled from 2 central rows of each plot for determination of total nitrogen, phosphorus and potassium (NPK) uptake. For potatoes, samples were separated into tubers, roots and shoot (Prasad and Hochmuth, 2016) while for legumes, only the shoot was analyzed. The samples were transferred to the laboratory, dried at 60°C to a constant mass, separately ground in a Wiley mill (0.84 mm mesh) and used for NPK determination. The total nutrient (NPK) uptake was calculated as the product between plant nutrient concentration and biomass weight.

#### 4.2.11 Statistical analyses

The data was analyzed using R software, version 3.5.2 (R Core Team, 2017). The effects of intercropping on tuber and legume grain yield, root length density, soil water balance and NPK uptake were tested using generalized linear models (Bates et al. 2015). Tukey's honest significant difference (HSD) test was used for treatment mean separations with threshold probability level set at  $p \le 0.05$ . An exponential function was fitted to give the relative change of soil moisture content as a function of root length density with an equation of the form y = abx, where a = 0. Principal component analysis (PCA) was performed using factoshiny function to examine relationships among evapotranspiration, root water uptake, soil moisture content, soil temperature and vegetal cover.

#### 4.3 Results

## 4.3.1 Vegetal cover establishment by different cropping systems

Crop cover development differed among the cropping systems and between the agroecological zones (Fig. 4.2). In the UM and LH, peak crop cover was significantly greater in intercropping (62–95%) compared to sole potato stands (48–83%). In the UH, peak crop cover was greater in sole potato (89%) than in potato-dolichos intercropping (74%) and sole dolichos (58%). On average, plots with dolichos recorded higher crop cover in the UM (16–94%) and LH (11–95%), and those with lima bean recorded the highest crop cover (12–96%) in the UH. Crop cover percentage taken at 105 and 119 days after planting was only significant for sole dolichos and potato-dolichos intercropping.



Days after planting

Figure 4.2: Development of vegetal cover by different treatments in upper midland, lower highland and upper highland. Values are 4 replicates expressed as averages over the four seasons.

# 4.3.2 Relationship between soil moisture and soil temperature

The amount of soil water retained decreased exponentially with increasing soil temperature (Fig. 4.3). However, the strength of the relationship varied among the different treatments. In the UM and LH, the relationship was stronger with sole potato ( $R^2 = 0.81$ ) than with the intercrops ( $0.54 \le R^2 \le 0.68$ ) and weakest with sole legumes ( $0.45 \le R^2 \le 0.58$ ). In the UH, the relationship was stronger with sole dolichos ( $R^2 = 0.68$ ) and potato-dolichos intercropping ( $R^2 = 0.61$ ) than with potato + lima bean ( $R^2 = 0.56$ ) and sole lima bean ( $R^2 = 0.51$ ).



Figure 4.3: Relation of soil temperature and water content under different cropping systems. Points are values of four replicates expressed as averages over four seasons and in each agro-ecological zone; upper midland (UM), lower highland (LH) and upper highland (UH).

# 4.3.3 Root length density

Root length density (RLD) at 0–0.3 m depth was significantly higher in sole potato stand than in intercropping (Table 4.2). At 0.3-0.6 m depth, highest RLD (1854 to 4146 m m<sup>-3</sup>) was recorded under potato-lima bean intercropping. This is compared to pure potato that showed RLD ranging between 246 and 930 m m<sup>-3</sup>. At 0.6–0.9 m depth in the UM, potato-dolichos intercropping showed highest

RLD of 10326 m m<sup>-3</sup>. At 90–120 cm depth, RLD ranged between 906–972 m m<sup>-3</sup> for potato-lima bean intercropping compared to 2286–2460 m m<sup>-3</sup> in potato-dolichos intercropping. Generally, dolichos established higher RLD in the LH and UM AEZs while lima bean showed higher RLD in the UH AEZ.

Table 4.2: Changes in crop root length density (m m<sup>-3</sup>) at tuber bulking stage in depth interval 0–0.1 m under different cropping systems and agro-ecological zones.

Agro- ecological zone	Cropping system/soil depth	0–0.3 m	0.3–0.6 m	0.6–0.9 m	0.9–1.2 m
Upper	Pure potato	5490 <sub>B</sub> <sup>c</sup>	930 <sub>A</sub> <sup>a</sup>	ŧ	ŧ
Midland	Pure lima bean	$1950_B{}^a$	$9564c^d$	846A <sup>a</sup>	1032 <sub>A</sub> <sup>a</sup>
	Pure dolichos	1470 <sub>AB</sub> <sup>a</sup>	1890 <sub>B</sub> <sup>b</sup>	$10326c^{d}$	1332 <sub>A</sub> <sup>b</sup>
	Potato + lima bean	$2526c^{b}$	$4146 D^{c}$	1566 <sub>B</sub> <sup>b</sup>	906 <sub>A</sub> <sup>a</sup>
	Potato + dolichos	2652 <sub>A</sub> <sup>b</sup>	3870 <sub>B</sub> <sup>c</sup>	5346c <sup>c</sup>	2460 <sub>A</sub> <sup>c</sup>
Lower	Pure potato	5886 <sub>B</sub> <sup>c</sup>	606 <sub>A</sub> <sup>a</sup>	ŧ	ŧ
highland	Pure lima bean	$1824_B^a$	8706c <sup>e</sup>	660 <sub>A</sub> a	906 <sub>A</sub> <sup>a</sup>
	Pure dolichos	1290 <sub>A</sub> <sup>a</sup>	1206 <sub>A</sub> <sup>b</sup>	$9804_B{}^d$	1830 <sub>A</sub> <sup>b</sup>
	Potato + lima bean	2886cb	3786 <sub>D</sub> <sup>d</sup>	1626 <sub>B</sub> b	$972A^{a}$
	Potato + dolichos	2670 <sub>A</sub> b	2892 <sub>A</sub> °	4206 <sub>B</sub> c	2286 <sub>A</sub> <sup>b</sup>
Upper	Pure potato	9270 <sub>B</sub> <sup>d</sup>	$246_{\text{A}}^{a}$	ŧ	ŧ
highland	Pure lima bean	5646 <sub>C</sub> <sup>a</sup>	$1854_{B}^{c}$	216 <sub>A</sub> <sup>a</sup>	ŧ
	Pure dolichos	3570c <sup>b</sup>	1230 <sub>B</sub> <sup>b</sup>	192 <sub>A</sub> <sup>a</sup>	ŧ
	Potato + lima bean	6486 <sub>C</sub> <sup>c</sup>	$2412_B{}^d$	192 <sub>A</sub> <sup>a</sup>	ŧ
	Potato + dolichos	1950 <sub>C</sub> <sup>a</sup>	$1194_{B}^{b}$	$234_{\text{A}}^{\text{a}}$	ŧ

Lower and uppercase letters indicate comparisons for means between the cropping systems and soil depths, respectively at  $p \le 0.05$  by Tukey's HSD test.  $\ddagger$  denotes data not observed. Values are 4 replicates expressed as averages over the four seasons.

# 4.3.4 Root water uptake

Root water uptake (RWU) in the 0–0.3 m depth was significantly higher in sole potato (52.7-99.8 mm) than in intercropping (20.2-69.6 mm) at tuber bulking stage of potato growth

(Table 4.3).

Agro-			Root w uptake	ater (mm)		Soil matric potential (-kpa)				
ecologic al zone	Cropping system				Soil de	pth (m)				
		0–0.3	0.3–0.6	0.6–0.9	0.9–1.2	0–0.3	0.3–0.6	0.6–0.9	0.9–1.2	
	Pure potato	52.7 <sub>B</sub> <sup>c</sup>	$9.7_{A}^{a}$	ŧ	ŧ	15.7 <sub>A</sub> <sup>a</sup>	28.9 <sub>B</sub> <sup>c</sup>	35.7c <sup>d</sup>	37.9c <sup>b</sup>	
Upper	Pure lima bean	$11.6_B{}^a$	$67.4c^{d}$	$47.7 D^{b}$	$5.5_{\text{A}}^{\text{a}}$	$25.5_B{}^b$	$19.5_{\text{A}}^{\text{a}}$	$23.4_{B}^{c}$	$30.2c^{a}$	
midland	Pure dolichos	10.5 <sup>a</sup>	29.3c <sup>b</sup>	75.9 <sub>D</sub> <sup>c</sup>	$21.9_{B}^{c}$	$32.8_B{}^d$	$18.8_{\text{A}}^{\text{a}}$	$12.8_{\text{A}}^{\text{a}}$	$26.8_B{}^a$	
	Potato + lima bean	$21.8_{B}^{b}$	56.7 <sub>C</sub> <sup>c</sup>	$20.6_{B}{}^{a}$	3.2 <sub>A</sub> <sup>a</sup>	$24.7_{\text{A}}^{\text{b}}$	$20.7_{\text{A}}^{ab}$	$23.7_{\text{A}}^{\text{c}}$	$31.7_{B}^{a}$	
	Potato + dolichos	$20.2^{b}_{A}$	$28.9_{B}^{b}$	45.9 <sub>C</sub> <sup>b</sup>	15.9 <sub>A</sub> <sup>b</sup>	$29.3c^{cd}$	$23.3B^{b}$	$16.3_{\text{A}}^{\text{b}}$	27.3 <sub>C</sub> <sup>a</sup>	
	Pure potato	$67.7_{B}^{d}$	$8.9_{A}^{a}$	ŧ	ŧ	$18.7_{\text{A}}^{\text{a}}$	34.6 <sub>B</sub> d	$38.4_{B}^{c}$	39.9 <sub>B</sub> <sup>b</sup>	
	Pure lima bean	$25.8_{B}^{c}$	$55.9c^{d}$	$25.1_{B}^{b}$	$10.5_{\text{A}}^{\text{a}}$	$28.6_B{}^b$	$13.2_{\text{A}}^{\text{a}}$	$26.4_B{}^b$	$30.5_{B}^{a}$	
Lower	Pure dolichos	21.8 <sub>A</sub> <sup>b</sup>	35.9 <sub>B</sub> °	$84.6c^d$	$27.8_{A}^{b}$	$33.2_{B}^{b}$	$19.2_{\text{A}}^{\text{bc}}$	$14.5_{\text{A}}^{\text{a}}$	$26.8_B{}^a$	
highland	Potato + lima bean	$38.6_B{}^a$	$31.8_{B}^{c}$	$14.9_{\text{A}}^{\text{a}}$	$8.3_{A}^{a}$	28.9 <sub>C</sub> b	$15.3_{\text{A}}^{\text{ab}}$	$23.9_B{}^b$	$25.1_{BC}^{a}$	
	Potato + dolichos	35.9 <sub>C</sub> <sup>a</sup>	$23.5_{\text{A}}{}^{\text{b}}$	$47.8_D^c$	$26.1_B{}^b$	30.4 <sub>C</sub> b	$24.3_{B}^{c}$	12.3 <sub>A</sub> <sup>a</sup>	28.3 <sub>C</sub> <sup>a</sup>	
	Pure potato	99.8 <sub>B</sub> e	14.8 <sub>A</sub> a	ŧ	ŧ	30.6 <sub>A</sub> a	$48.4_B{}^b$	$50.2_{B^{a}}$	$52.3{\scriptstyle\rm B}{}^{a}$	
	Pure lima bean	49.8 <sub>C</sub> <sup>b</sup>	$24.9_B{}^b$	$9.9_{\text{A}}^{\text{a}}$	ŧ	38.4 <sub>A</sub> b	$40.8_B{}^a$	$46.7_{BC}^{a}$	51.2 <sub>C</sub> <sup>a</sup>	
Upper highland	Pure dolichos	22.3 <sub>C</sub> <sup>a</sup>	$12.8_B{}^a$	$7.6_{\text{A}}^{\text{a}}$	ŧ	33.2 <sub>A</sub> a	$35.1_{\text{A}}^{\text{a}}$	$49.2_{B}^{a}$	$50.7_{B}^{a}$	
0	Potato + lima bean	69.6 <sub>C</sub> <sup>d</sup>	$26.5_B{}^b$	$9.4_{\text{A}}^{\text{a}}$	ŧ	34.8 <sub>A</sub> a	$38.6_{\text{A}}^{\text{a}}$	$46.6_B{}^a$	$49.9_{B}^{a}$	
	Potato + dolichos	59.3 <sub>C</sub> °	$21.9_{B}^{b}$	$6.5_A{}^a$	ŧ	35.4 <sub>A</sub> ab	41.7 <sup>a</sup>	46.1 <sub>C</sub> <sup>a</sup>	48.3 <sub>C</sub> <sup>a</sup>	

Table 4.3: Root water uptake and the corresponding soil matric potential measured in different cropping systems at tuber bulking stage of potato development.

Lower and uppercase letters indicate mean comparisons between cropping systems and soil depths, respectively by Tukey's HSD test. ‡ denotes data not observed. Values are 4 replicates expressed as averages over the four seasons.

The corresponding soil matric potential at this stage was significantly lower in sole potato (15.7–30.6 kpa) compared to intercropping (20.7–41.8 kpa). In the 0.3–0.6 m depth, potato-lima bean intercropping recorded highest RWU (24.9–56.7 mm). The RWU of sole potato at this depth ranged between 9.7 and 14.8 mm. The corresponding soil matric potentials were 15.3–38.6 kpa in potato-lima bean intercropping and 28.9–48.4 kpa in sole potato. In the 0.6–0.9 m depth, potato-dolichos intercropping showed the highest RWU of 75.9–84.6 mm.

# 4.3.5 Soil water storage as a function of evapotranspiration and soil evaporation

The cumulative soil water storage ( $\Delta$ S), soil evaporation and evapotranspiration differed among the different cropping systems and sites (Fig. 4.4).



Figure 4.4: Cumulative evapotranspiration, evaporation and soil water storage ( $\Delta$ S) under different treatments in the three agro-ecological zones. Values are 4 replicates expressed as averages.  $\Delta$ S is shown for 0–0.3 m of soil layer. Values above and below the zero line in  $\Delta$ S graph indicate soil water recharge and depletion, respectively.

The variance was greater in the UM than in the LH and lowest in the UH. In the UM and LH, the  $\Delta$ S was greater under sole potato (-8.8 to 10.3 mm) compared to intercropping (4.3 to 14 mm) and lowest under sole legumes (3.1 to 9.8 mm). In these sites, soil water storage under dolichos and lima bean responded similarly to intercropping with no significant difference. In the UH,  $\Delta$ S was markedly lower and ranged from 1.4 to 4.8 mm in sole potato and 3.4 to 5.8 mm in intercropping. Peak soil evaporation and ETa coincided with the peak soil-water depletion irrespective of treatments. In the UM, cumulative soil water storage for the sampling conducted between 77 and 119 days after planting were negative in sole potato (-1.6 to -8.8 mm). The corresponding cumulative ETa was at its peak ranging between 146 and 219 mm. Similarly, cumulative soil evaporation in this period was high ranging between 45 and 117 mm. The cumulative soil water storage was negative in sole potato plots in the LH (-0.2 to -4.7) for the period between 63 and 119 days after planting.

#### 4.3.6 Principal component analyses of factors influencing soil water storage

Soil water content was explained by two principal components that accounted for 70% of the variance (Fig. 4.5). The first component explained 53% of the variance and the rest (17%) was accounted for by the second component. Crop cover had the highest factor loading in the PC1 (r = 0.91). This was followed by ETa/ETp ratio (r = 0.84), evapotranspiration (r = -0.79), root water uptake (r = -0.69), soil temperature (r = -0.51) and soil evaporation (r = 0.50). Soil evaporation was the dominance factor in PC2 (r = -0.74). All the correlations were highly significant (p = 0.00).



ETa/ETp is the ratio of actual evapotranspiration to potential evapotranspiration. Vectors indicate the degree of correlation between each factor and the axes.

Figure 4.5: Principal component analyses of factors influencing soil water storage in a potato-

legume intercropping.

# 4.3.7 Soil water balance

Across the AEZs, lowest soil water balance (0–0.3 m depth) occurred in sole potato (6.5–14.7 mm), were intermediate in intercropping (16.4–40.3 mm) and highest in sole legumes (21.7–36.3 mm) (Fig 4.6). Potato-dolichos intercropping exhibited significantly higher soil water balance compared to potato-lima bean intercropping in the UM and LH AEZs. Evapotranspiration was the main component of soil water balance across the cropping systems and AEZs and varied between 218 mm and 282 mm in pure potato stands and 301–379 mm in intercropping (Appendix 1). Runoff contributed significantly to soil water balance in the LH (9.8–55.5 mm) and in the UH AEZs (38.6–

60.2 mm). Similarly deep percolation contributed significantly to soil water balance in the UH and was significantly greater in sole potato than in potato-lima bean intercropping.



Letters indicate comparisons for means between sole cropping and intercropping systems at  $p \le 0.05$  by Tukey's HSD test. Values are 4 replicates expressed as averages over the four seasons. Error bars are standard error of means.

Figure 4.6: Total water balance measured under different cropping systems and sites at the topsoil depth (0–0.3 m).

# 4.3.8 Intercropping effect on tuber yield

In the UM, sole potato (100% potato plants per ha) attained potato tuber yields of 11.4 t ha<sup>-1</sup>, an equivalent of 0.26 kg tuber weight per plant (Fig. 4.7). Potato intercropped with dolichos and lima bean (50% potato plants per ha) attained potato tuber yield of 10.8 and 10.1 t ha<sup>-1</sup> (equivalent of 0.49 and 0.45 kg tuber weight per plant). This was an increase of 88% in tuber weight per plant.



Figure 4.7: Potato and legume yields expressed in tonnes per ha and kg per plant in sole cropping and intercropping systems in the upper midland (UM), lower highland (LH) and upper highland (UH). Letters indicate comparisons for means between sole cropping and intercropping systems at  $p \le 0.05$  by Tukey's HSD test. Values are 4 replicates expressed as averages over the four seasons.

Similarly, in the LH, sole potato (100% potato plants per ha) attained potato tuber yields of 23.2 t ha<sup>-1</sup>, an equivalent of 0.52 kg tuber weight per plant. This is compared to potato intercropped with dolichos and lima bean (50% potato plants per ha) that attained potato tuber yield of 16.4 and 14.3 t ha<sup>-1</sup> (0.74 and 0.64 kg tuber weight per plant) respectively. In the UH, tuber yields were significantly greater for potato-lima bean intercropping (19.2 t ha<sup>-1</sup>; 0.86 kg tuber weight per plant) than for potato-dolichos intercropping (13.5 t ha<sup>-1</sup>, an equivalent of 0.61 kg tuber yield per plant). Potato recorded significantly higher tuber weight per plant in sole potato (0.68 kg/plant) than in potato-dolichos intercropping (0.61 kg/plant).

Legume grain yields per area basis were lower in intercropping  $(0.3-4.8 \text{ t ha}^{-1})$  compared to sole legumes  $(1.3-9.3 \text{ t ha}^{-1})$  (as sole legumes occupied 100% land *vs* 50% in intercropping). However, when yields were computed per plant basis, legume intercrops had markedly higher grain yields (0.06-0.11 kg grain weight per plant) than the sole legumes (0.01-0.07 kg grain weight) per plant). In the upper highland, lima bean grown in sole recorded 9.3 t ha<sup>-1</sup> grain yield, an equivalent of 0.1 kg grain weight per plant compared to 4.8 t ha<sup>-1</sup> in legume intercrops (an equivalent of 0.11 kg grain weight per plant). Dolichos yield was markedly lower in the UH, both in sole and intercrop.

# 4.3.9 Nutrient (NPK) uptake and use efficiency

Nitrogen, phosphorus and potassium uptake differed significantly among the treatments (Table 4.4). Generally, highest NPK uptake occurred in intercropping compared to potato monocropping. The uptake of NPK were highest in the upper highland AEZ and lowest in the upper midland AEZ. Intercropping recorded significantly higher NPK use efficiency ranging from 62–413 kg PEY kg<sup>-1</sup> vs 45–143 kg PEY kg<sup>-1</sup> in sole potato in the UM, 80–387 kg PEY kg<sup>-1</sup> vs 46–

133 kg PEY kg<sup>-1</sup>in the LH and 62–367 kg PEY kg<sup>-1</sup> vs 80–198 kg PEY kg<sup>-1</sup>in the UH (Table 4.4). P use eeficiency was about 2 fold greater in intercropping compared to sole potato across the sites. Table 4.4: Nutrient uptake (NPK) and use efficiency by different cropping systems in the three agro-ecological zones.

Parameter Cropping system		Upper midland			Lov	ver high	land	Uppe	Upper highland		
		Ν	Р	K	N	Р	K	N	Р	K	
Nutrient	Sole potato	90a	45c	143c	133b	46c	167b	197ab	80b	198b	
uptake $(k \alpha h \alpha^{-1})$	Sole lima bean	75b	40c	58d	59d	47c	62c	75c	60c	68c	
(kg lla )	Sole dolichos	79b	51b	68d	83c	85b	77c	67c	83ab	40d	
Potato + lim	Potato + lima bean	98a	61a	176b	157a	98a	196ab	223a	96a	238a	
	Potato + dolichos	96a	53b	203a	148a	70b	216a	188b	34d	201b	
Nutrient	Sole potato	79c	195c	62e	104d	297b	80d	82d	202b	82c	
efficiency (kg PEY Sole dolichos	Sole lima bean	147b	305b	210b	316a	395a	302a	294a	367a	323a	
	Sole dolichos	248a	382a	286a	301a	294b	325a	196b	153c	324a	
kg <sup>-1</sup>	Potato + lima bean	254a	413a	142c	242b	387a	194b	151c	349a	142b	
nutrient)	Potato + dolichos	246a	322b	85d	167c	352a	114c	68e	375a	63c	

Letters indicate mean comparisons between cropping systems by Tukey's HSD test. <sup>‡</sup> denotes data not observed. Values are 4 replicates expressed as averages over the four seasons.

#### 4.3.10 Relationship among soil water content, nutrient uptake and soil temperature

Soil water balance decreased with increasing soil temperature (r= -0.89), root water uptake (r= -0.75) and increasing uptake of N (r = -0.88), P (r = -0.67) and K (r = -0.85) (Table 4.5), but increased with increase in soil water content (r = 0.98).

Table 4.5: Pearson's correlation of soil temperature, nutrients (NPK) uptake and crop water balance.

Parameter	Soil temperature	Soil water balance	RWU	SWC	N uptake	P uptake
Soil water balance	-0.89***					
RWU	-0.75**	-0.59*				
SWC	-0.83**	0.98***	0.79**			
N uptake	-0.68*	-0.88**	0.88***	0.83***		
P uptake	-0.56*	-0.67*	0.61*	0.51*	0.53*	
K uptake	0.52*	-0.85**	0.71**	0.56*	0.59*	0.41

Corelation significant at p < 0.001 (\*\*\*), p < 0.01 (\*\*) and p < 0.05 (\*). RWU; root water uptake, SWC; soil water content.

Soil temperatures related negatively with uptake of N (r = -0.68), P (r = -0.56) and K (r = -0.52). Root water uptake related significantly with N uptake (r = 0.88), P uptake (r = 0.61) and K uptake (r = 0.71). Nutrient uptake increased with increase in SWC (R =0.83, 0.51and 0.56 for NPK, respectively).

#### **4.4 Discussions**

#### 4.4.1 Root water uptake

The variation of root water uptake (RWU) among the cropping systems was attributed to the heterogeneous root distribution in the soil profile and to the temporal and spatial variations in soil water content. While potato crop was characterized by roots that rarely exceeded a vertical depth of 0.3 m across the sites, lima bean and dolichos roots were traced up to 0.9 and 1.1 m respectively. These legumes could therefore access soil water to deeper layers than the potato crop. In this way, intercropping enhanced deep root water extraction and eliminated resource overlap. This is in agreement with the studies which have established that potato has shallow fibrous root systems which are concentrated in the upper soil layer (0–0.3 m) making the crop highly sensitive to fluctuating soil moisture (Aliche et al. 2018). The lack of active roots beyond 0.3 m depth in sole potato is therefore the reason why soil matric potential remained high in deeper soil layers.

At vegetative growth of potato, the soil water recharge by rainfall maintained the soil matric potential high (above the field capacity). Root water uptake at this stage was thus not limited by the soil water availability, but rather by the undeveloped root systems across the treatments. At tuber bulking stage, cessation of rains in the UM and LH caused the soil matric potential to significantly drop in sole potato plots. At this stage in the UM, the potato in sole stands generally showed signs of wilting as soil water was strongly held restricting its movement and uptake by the

roots. Intercropping however ensured high canopy integrity and deeper rooting systems which kept both evapotranspiration and soil water extraction high.

#### 4.4.2 Evapotranspiration, soil evaporation and soil water storage

Intercropping potato with the legumes promoted leaf area expansion leading to rapid ground cover development. This cover created shade that buffered the soil against direct solar and considerably reduced evaporation losses. Sole potato however left bare spaces between crop rows through which solar penetration occurred. Crop cover mediates soil water fluctuations by lowering evaporative losses through shading and holding horizontal and vertical water flow in the soil (Daly and Porporato, 2005; Asbjornsen et al. 2011). Across the sites, evaporation of surface water occurred mainly at the onset and end of the season, a period when the soil was bare and the solar intensity was high.

The cumulative ETa during the growing season was lower under sole potato due to two main reasons. First, the shallow root system by potato limited its capacity to extract the readily available water in the deeper soil layer. Second, potato attained maturity relatively fast compared to legumes. While the crop growth duration was about 90 days in potato, lima bean was characterized by slightly longer duration averaging 110 days while dolichos exhibited indeterminate growth. The extensive canopy with longer duration increased the plant demand for water and therefore evapotranspiration. Cumulatively, the amount of water consumed by legume intercrops after potato harvest averaged 30–60 mm across the seasons. This variability could be explained by changes in the timing of leaf senescence in response to ambient temperatures and soil water availability.

The topsoil changes in soil water storage ( $\Delta S$ ) occurred mainly after rainfall events and only after topsoil saturation occurred did the deeper soil layers respond to additional soil water.

This was most clearly shown in the UM under sole potato crop at a time the rainfall events continued to a period of about 30 days. Thereafter, the topsoil dried up rapidly following higher evaporation that exceeded recharge from rainfall. The high crop cover under intercropping at this time insulated the soil from direct solar and minimized the evaporation rates. The rainfall occurring at full vegetative growth (35–77 DAP) contributed mainly to ETa irrespective of treatments. This resulted in significant changes in soil water storage. As the soil dried up later in the season, deeper soil layers contributed an increasing fraction of water to ETa under intercropping. This was shown by the increased root water uptake in the deeper soil layers. Additionally, the high canopy cover extended by legume intercrops after potato senesces insulated the soil against evaporation and led to more water storage (total rainfall > ETa).

The differences in soil water storage ( $\Delta$ S) between the sites and seasons were in part a function of rainfall. The more the rainfall, the greater the soil water content. Consequently, the larger rainfall amounts that recharged the soil after an extended period of dryness caused significant changes in soil water storage. The soil water storage in the topsoil under sole legumes was almost balanced in the UM and LH indicating that the legumes extracted water at depths beyond 0.3 m. Water fluxes below root-zone has been shown to increase the amassing of soil moisture in higher depths thus making it available for plant uptake by deep roots (Sakai et al. 2009). The coincidence of peak ETa, soil evaporation and soil water depletion in sole potato was an indication that ETa and evaporation were major contributor to changes in soil water storage.

In the UH, the contribution of runoff and deep percolation largely accounted for the changes in soil water storage. Generally, when the rainfall exceeded infiltration rates, a higher proportion of rainfall was routed to surface runoff or recharged to the ground water reservoir. This process was greater in sole potato. In the UH, lima bean achieved a dense canopy that intercepted

and reduced raindrop-hitting force and slowed down the velocity of runoff. Dolichos growth was however, hindered by cold in the highlands, and thus did not establish sufficient cover to intercept the raindrop.

The more stable changes in soil water storage at start of the seasons was because of the large rainfall at a period when demand for transpiration remained low, and so the soil water recharge was rapid. At this time, the lower ETa was due to the undeveloped plants with limited capacity to transpire while the lower ETa at the end of the seasons was a consequence of plant maturity resulting in senesces of shoot. In the UH, the ETa was not primarily limited by soil water availability as there were an average of 3 days when ETa to ETp ratio was above unity; see appendix 2). These days followed relatively large rainfall events that occurred when the plants had developed full foliage.

The lower crop cover and soil moisture under sole potato were a major contributor to the lower ETa/ETp relative to potato-legume intercropping. Potatoes are much less tolerant to water stress than the legumes and prospered only when the soil water content was sufficient to meet the crop water requirement. In intercropping, the better rooting system increased the capacity of crops to extract water thus keeping evapotranspiration high with little impact on topsoil water reservoir. Increased canopy under intercropping also increased the transpiration component of evapotranspiration and enhanced the crop water use. Bachand et al. (2013) reported that transpiration accounts for 50–90% of total ETa during a growing season in intercropping systems.

The high soil evaporation, runoff and evapotranspiration were the main reasons for the low soil water balance under sole potato. Runoff occurred mainly at the onset of rains when crop cover was low and soon after harvest when the soil was bare. The high soil water balance in the UH compared to LH and UM was due to the high rainfall amounts that raised the soil water tables. Besides, the UH exhibited lower ambient temperatures which lowered soil evaporation, thus increasing soil water storage.

#### 4.4.3 Principal factors explaining changes in soil water storage

Soil water storage was influenced by crop cover, evapotranspiration, soil evaporation, root water uptake, and the ratio of ETa/ETp. Crop cover was the main determinant of soil water storage as shown by the highest correlation. Crop cover mediated the soil water fluctuations by lowering the evaporative losses through shading. Vegetation cover lowers the amount of direct solar radiation reaching the ground surface and subsequently buffers the soil temperatures (Salisbury and Spomer, 1964; Pielke, 2001; Blok et al. 2010; Graham et al. 2012). The stronger relationships between  $\Delta$ S and ETa compared to  $\Delta$ S and ETa/ETp showed that ETa was mainly constrained by soil water content (Brutsaert and Chen 1995; Yamanaka et al. 2007).

The decrease in soil water storage with both soil temperature and soil evaporation was because of the warming up of soil, the energy of which converted water into vapor thus minimizing the liquid phase of soil water. The exponential decrease in soil moisture with increase in soil temperature was an indication that the migration of soil moisture was accompanied by heat conduction thus affecting the variations in soil temperature. This is consistent with soil water-heat system relation studies that have shown that changes in soil temperature affect the phase changes of soil water content (Lu et al. 2011; Li et al. 2018). When only the wet season data was considered in this relation, the effect was weaker and not significant. This was because of the rainwater recharge that moistened the soil, lowered the soil temperature, and increased the soil water storage. Further, the crop cover weakened this association by restraining the influence of direct solar on soil-water-heat relation. Nyawade et al. (2019) reported that crop cover maintains the stability of

soil moisture and temperature and therefore maintains a strong interaction between the two variables.

Root water uptake by legume intercrops occurred mainly in the deep soil layers while that by potato occurred in the topsoil layers thus guaranteeing resource complimentarity. When the soil was moist, water moved freely towards the roots making root water uptake high. Scanlon et al. (2005) showed that the high evapotranspiration rates under elevated soil temperatures reduced soil water storage and lowered RWU. This behavior was also reflected in numerical modeling studies by Walvoord et al. (2004) and Scanlon et al. (2003) who observed that high soil temperatures lowered the soil water potential and thus restricted the RWU.

The contribution of intensive root system by intercropping on moderation of soil temperature could not be overruled in this study as soil water storage related significantly with the root length density (RLD). Vegetation that supports a dense rooting system absorbs moisture stored in the soil and reduces soil evaporation and water loss (Lu et al. 2011). As soil temperatures increased, the soil dried out and water mobility declined, making increased RLD more important for lateral water extraction. This is consistent with findings reported by related studies which unraveled that high soil temperatures and water deficit within the root-zone create a mechanism that forces plants to develop deeper root systems so as to access the subsoil water reserves (Kirkegaard and Ryan, 2014; Thorup-Kristensen et al. 2016).

No significant variation occurred in soil temperature at soil layer greater than 0.30 m indicating that heat could be barely transported beyond this depth. We partly attributed this to the reduced heat flux contributed by high soil water content (Nyawade et al. 2019). The enhanced soil water infiltration under intercropping enhanced the soil water recharge but weakened the stability of soil temperature. As the thermal capacity of water is larger than that of soil (Garcia et al. 2011),

the greater the soil water content, the less varied the range in soil temperature. Therefore, the soil under intercropping that was characterized by higher soil moisture content released more heat than that under sole potato, especially during the dry conditions when heat flux was greater.

# 4.4.4 Intercropping effect on crop yield

Intercropping had 50% fewer potato plants per ha compared to sole potato (100%), thus potato tuber yields per area unit were generally lower. However, when the yields were expressed in tubers per plant, intercropping showed a marked increase in yield over sole potato. Due to their deep taproot systems, legume intercrops were capable of recapturing the percolated water, absorbing it more efficiently and converting it into crop biomass. This reduced resource overlap allowing intercropped potato to consume topsoil water.

The high soil temperatures and low moisture contents under sole potato had devastating effects on tuber yields, especially in the UM. This is because high temperatures promoted haulm growth at the expense of tuber growth and therefore much pool of starch available for tuber growth was directed to shoot. As soil and air temperature increase, the rate of respiration increases, while the rate of photosynthesis decreases (Chaves et al. 2002). This suggests higher respiration of starch to sustain plant growth rather than it being stored in tubers. The high soil temperatures under sole potato were extended to stolon development phase of potato growth. This greatly reduced stolon numbers thus affecting tuber size distribution and subsequently the biomass accumulation. Accordingly, when heat stress coincided with tuber maturation in the UM, this hastened leaf senescence and interrupted new leaf formation, thus resulting in tuber yield reduction.

#### 4.4.5 Nutrient uptake and use efficiency

The higher nutrient uptake due to legume intercropping was a reflection of enhanced nutrient use complementarity and facilitation contributed by legume integration. The deep root
systems by the legumes enhanced nutrient acquisition in the deep soil layers. This offered complementarity that allowed potato to take up nutrients in the surface stratum. In the UM and LH zones, dolichos exhibited deep rooting system with a large surface area that grew up to a depth exceeding 1 m. Whitbread et al. (2011) showed the ability of dolichos to capture N from the subsoil and pump it to the topsoil strata thus making it available for the shallow rooted cereals. The facilitation mechanisms that could have enhanced nutrient availability and uptake include the organic acid production and enhanced mycorrhizal colonization (Hinsinger et al. 2011). Dolichos secrets organic substances such as phosphatases which promote release of P from organic materials (Hinsinger et al. 2011; Schneider et al. 2019). The greater nutrient uptake by dolichos than lima bean in the UM and LH AEZs may be related to the ability of dolichos to develop greater RLD under high soil temperature conditions. The converse was observed in the UH AEZ that was characterized by cold intolerable by this crop (dolichos).

The increase in uptake of nitrogen, phosphorus and potassium with increasing soil moisture contents was not only due to the improved root growth leading to enhanced nutrient acquisition, but also to the increased nutrient mobility caused by soil water dissolution. Enhanced LAI development in intercropping raised the plant nutrient demand thus driving NPK uptake from the soil with subsequent translocation to the shoots. As soil water content increased under potatolegume intercropping, a greater continuity of the liquid phase occurred. This likely lowered the tortuosity of nutrient diffusion path thus allowing nutrient from larger soil volume to be absorbed by plant roots. Zeng and Brown (2000) developed a relation which showed that soil nutrient mobility increases with soil water content suggesting that more nutrients can diffuse to plant roots at sufficient soil water content. Nutrient acquisition by diffusion and mass flow become faster where the soil water available in the root-zone is optimal for plant growth (Barber, 1995).

The high phosphorus use efficiency reported in this study agrees with those reported by Gitari et al. (2020) and indicates that potato is a more efficient crop in translocation of nutrients to the tubers. The use efficiency of P was high in the UM where temperatures and rainfall were the main limitation to crop production. Sandana (2016) observed that potato utilizes P well even when it is in inadequate supply despite the low uptake of the nutrient. Such findings are in line with those obtained by Wang et al. (2015) who reported high P uptake efficiency even when the nutrient supply was limited. The higher nutrient use efficiency due to legume intercropping was largely due to the higher tuber dry matter yield and enhanced legume biomass production. Further, the decrease in NPK losses in eroded sediments, leaching and volatilization under the intercrop plots (see chapter six) enhanced nutrients available for crop uptake, thus increasing the use efficiency.

#### 4.5 Conclusions

The results from this study suggest that potato-legume intercropping has the capacity to keep root zone soil water and soil temperature optimal thus increasing nutrient uptake and potato yield. The changes in soli water storage in the topsoil reflected the net results of rainfall recharge, water consumption by ETa, output from runoff, soil evaporation and inputs from soil water redistribution. The greater root biomass contributed by legume intercrops in deeper soil layers enhanced interception of soil water. Reduction of surface temperatures was a result of increased crop cover which insulated the soil from direct solar radiation thus reducing soil evaporation. This in turn increased the soil water content and nutrient uptake. Modification of microclimatic conditions in this manner could be an effective measure to adapt potato to midland elevation agrofood systems.

#### **CHAPTER FIVE**

# SHORT-TERM DYNAMICS OF SOIL ORGANIC MATTER FRACTIONS AND MICROBIAL ACTIVITY IN SMALLHOLDER POTATO-LEGUME INTERCROPPING SYSTEMS Abstract

Continuous cultivation of potato (Solanum tuberosum L.) in monoculture systems represents the greatest factor deteriorating soil organic matter (SOM) in smallholder farms. With an aim to breaking this norm, a 2-year field trial intercropping potato with two legumes: lima bean (Phaseolus lunatus L.) and dolichos (Lablab purpureus L.), was conducted in the upper midland (1552 meters above sea level (masl.)), lower highland (1854 masl.) and upper highland (2552 masl.) agro-ecological zones of Kenya. The study was laid out in a randomized complete block design with four replications. Residues from each cropping system were quantified at the end of each season and incorporated back into the soil at start of the subsequent season. A combined physical and density fractionation was used to separate the soil in macro-aggregates  $(>250 \,\mu\text{m})$ , micro-aggregates (250–50  $\mu\text{m}$ ) and silt plus clay fractions ( $<50 \,\mu\text{m}$ ), while SOM was partitioned into labile (density of 1.65 to 1.85 g cm<sup>-3</sup>) and stable (2.60 g cm<sup>-3</sup>) fractions. Microbial biomass contents were determined by chloroform fumigation while enzymatic activities were assessed by hydrolyses of fluorescein diacetate and dehydrogenase. Compared to sole potato, intercropping increased the contents of light fraction organic matter by 12–28%, dissolved organic matter by 7–21% and microbial biomass by 15–38%, thus stimulating enzyme activities. Trends in soil microbial respiration followed those of enzyme activity and were 20-34% higher in intercropping than in sole potato. These results affirm legume intercropping as a possible entry point to restoring the impoverished soil organic matter in smallholder potato farming systems.

Keywords: Intercropping; microbial activity; microbial biomass; soil organic matter fractions.

# **5.1 Introduction**

Potato production in subtropical highlands is done mainly in monoculture systems yet the crop retains very little residues and therefore has limited capacity to return organic matter into the soil (Angers and Carter, 1996; Gitari et al. 2018a; Nyawade et al. 2019). Even with this little residue retained, most of it is used to feed livestock or burned during land preparation. This problem is particularly important among the smallholder farmers who lack the capacity to apply fertilizer or put in place adequate soil conservation measures to control soil erosion (Nyawade et al. 2018b; Gitari et al. 2019). As potato takes 40–45 days to establish full groundcover, and maintains this cover only for 20 days before senesces sets in, the crop leaves the soil exposed to high surface temperatures that accelerates oxidation of soil organic matter (SOM) making its supply to be imbalanced (Reicosky et al. 1995). An imbalanced SOM lacks the capacity to supply nutrients to the microbial population and therefore restricts their growth and activity (Haynes and Tregurtha, 1999). Consequently, the average potato yields in subtropical highlands range between 3 and 15 t ha<sup>-1</sup>, an observation that has been related to the low content of SOM. This factor if not checked, may compel farmers to abandon potato production.

The extent of the adverse effect of poor potato cropping systems on SOM depends largely on which size fraction is affected (Nyawade et al. 2018b). These fractions differ in their quality, quantity and turnover rates, and respond differently to soil management (Kapkiyai et al. 1999). The stable SOM includes well-decomposed microbial products that form mineral complexes with organic compounds and is relatively recalcitrant to microbial attack (Camberdella and Elliot, 1992; Six et al. 2002). The labile SOM fractions are derived from the newly incorporated organic materials and include the light fraction organic matter, microbial biomass and dissolved SOM (Haynes, 2005). This component of SOM has rapid turnover rate and responds fast to soil management intervention compared to the total and stable SOM (Haynes, 2005; Louis et al. 2016).

The soil microbial biomass is a living fraction of SOM consisting mainly of bacteria and fungi that decompose litter and crop residues (Naresh et al. 2017). This process releases nutrients that are made available for plant uptake (Smith and Paul, 1990). Generally, the microbial biomass makes up 5–10% of the total SOM, part of which is released to the soil in plant available forms upon death of soil microbes (Louis et al. 2016). The dissolved organic matter (DOM) results from microbial degradation of SOM and is used as substrate by the soil microbes (Roper et al. 2010). The rate at which labile fractions of SOM decompose determines the release of nutrients to plants (Kapkiyai et al. 1999). Therefore, any activity that rapidly alters the contents of soil labile organic matter are regarded to be early signals of changes in soil productivity that may occur over an extended period.

Quantification of soil microbial biomass activity is based on the rate at which respiration and enzyme hydrolyses by soil microbes occur (Naresh et al. 2017). If the rate of respiration is related to the corresponding microbial biomass size, it is defined as microbial metabolic quotient (qCO<sub>2</sub>) (Lupwayi et al. 1998). The qCO<sub>2</sub> is sensitive to short-term changes in SOM and therefore provides early warning to a degrading soil quality. The qCO<sub>2</sub> gets elevated when perturbations such as high soil temperatures and soil water deficits exacerbate SOM depletion, thus negating the functions of soil microbes (Anderson and Domsch, 1990). These processes may greatly vary with cropping systems, soil management and agro-ecological zones (Lupwayi et al. 1998; Naresh et al. 2017). This is because the microbial biomass just like any other organic matter fraction, is influenced by soil particle size distributions which has great relation with the soil management and cropping systems (Camberdella and Elliot, 1992).

The macro-aggregate provides conducive micro-environment for microbial growth and also confers physical protection to the micro-aggregates thus occluding the associated SOM from physical perturbations (Six et al. 2002). It is in this regard that simple and sustainable strategies capable of increasing aggregate formation have been proposed in smallholder potato cropping systems (Nyawade, 2015; Burke, 2017; Nyawade et al. 2018b). The proposed measures highly regard technologies that retain at least 22.4 Mg ha<sup>-1</sup> residue dry matter *in situ*, a value that corresponds to 1% increase in SOM (MacMillan and Buchanan, 1988). This large biomass can hardly be attained by pure potato crops (Angers and Carter, 1996). Thus, addition of materials with high lignin contents as mulch has been considered appropriate (Franchini et al. 2002). These materials are however characterized by nitrogen immobilization resulting in low nutrient release (Hassink, 1995). Legume intercropping due to its ability to integrate crop components with high nitrogen contents, offers an alternative that can reduce nitrogen immobilizations (Naresh et al. 2017). When returned to cropland, the fresh legume residues provide nucleation useful in formation of macro-aggregates which are the favorable habitat for soil microbes (Ghani et al. 2003; Liang, 2013.).

It is in view of this background that this study was conducted to assess the short-term effect of potato-legume intercropping on dynamics of SOM fractions (microbial biomass, dissolved SOM, light fraction SOM and heavy fraction SOM) in three agro-ecological zones of Kenya (upper midland, lower highland and upper highland). Early detection of changes in SOM is important for designing sound management practices that can restore soil quality, soil health and soil productivity. Identification of intercropping systems capable of increasing SOM contents to optimal levels under different agro-ecological conditions offers the possibility of potato integration into wide range of agro-food systems.

#### **5.2 Materials and methods**

## 5.2.1 Study sites

The trials were installed in the wet season of 2016 and were continued till 2018 dry season in three agro-ecological zones of Kenya; upper midland (1552 m above sea level (masl), lower highland (1894 masl) and upper highland (2552 masl) agro-ecological zones of Kenya. The region is representative of agricultural farming systems of Kenya where inherent low soil organic matter content greatly limits potato production. Detailed descriptions of the study sites are given in section 3.1. Measured soil properties (0–40 cm depth) before the experiment are provided in Table 5.1.

Agroecological	Soil depth	Clay	Silt	Sand	Textural class§	pН	HFOM	LFOM	DOM	MBC	
zone	(cm)		%					mg C kg <sup>-1</sup>			
Upper	0–10	24.5	33.3	42.2	CL	5.0	453	337	271	111	
midland	10-20	24.2	36.9	38.9	CL	5.0	451	190	208	36	
	20-30	28.9	29.8	41.3	CL	4.9	414	120	111	21	
	30–40	23.8	32.4	43.8	CL	4.9	388	98	108	20	
Lower	0–10	49.7	22.5	27.8	С	5.1	610	544	418	128	
highland	10-20	49.2	24.2	28.9	С	5.1	549	324	306	79	
	20-30	50.1	24.2	25.7	С	5.2	503	203	135	64	
	30–40	51.3	24.8	23.9	С	5.2	500	179	126	61	
Upper	0–10	38.3	56.1	5.6	SC	5.2	560	575	425	216	
highland	10-20	36.9	58.4	4.7	SC	5.2	454	324	325	110	
	20-30	34.6	59.5	5.9	SCL	5.3	426	205	103	41	
	30–40	33.9	57.9	8.2	SCL	5.3	421	168	104	43	

Table 5.1: Initial soil properties.

<sup>§</sup>clay (C), clay loam (CL), silt clay (SC) and silt clay loam (SCL). : Heavy fraction organic matter (HFOM), light fraction organic matter (LFOM), dissolved organic matter (DOM) and microbial biomass carbon (MBC).

#### 5.2.2 Trial design and crop husbandry

The trials were arranged in a randomized complete block design in plots measuring 4.75 m long by 3 m wide. The treatments comprised of sole stands of potato (*Solanum tuberosum* L., Unica cv-CIP 392797.22), sole lima bean (*Phaseolus lunatus* L.), sole dolichos (*Lablab purpureus* L.) and intercropping of potato with the two legumes. Each treatment was replicated 4 times. Detailed descriptions of crop husbandry and trial management is outlined in section 3.4 and 3.5.

#### 5.2.3 Soil sampling and analyses

Sampling of soils was done at inter-rows of ech plot using soil piston augers for every 10 cm interval along soil profile of 0–40 cm. Sampling was conducted at vegetative growth of potato (about 60 days after planting). The depth interval 0–40 cm was considered appropriate as it represented the rooting depth for potato crop (the depth at which >70% of the root volume was found) (Mahdian and Gallichand, 1996). In each plot, 80 soil samples were taken (5 cropping systems × 4 replicates × 4 sampling depths). The samples for each treatment were composited for each depth and frozen until analysis for soil pH (by water), soil texture (hydrometric) (Gee and Bauder, 1986), total N (Keeney and Nelson, 1982) and total organic carbon (Nelson and Sommers, 1996).

## 5.2.4 Soil moisture and soil temperature

Quantification of soil moisture and soil temperature was done between sowing and harvesting using Onset HOBO probes (UX120-006 M) installed at every 10 cm interval along soil profile of 0–120 cm. The soil moisture and soil temperature data at depth 0–30 cm was related with different SOM fractions and microbial activity.

### 5.2.5 Estimation of shoot and root biomass

Root and shoot biomass were measured at vegetative growth of potato (60 days after planting) by sampling four inner rows of each treatment (2 strips of legumes and 2 strips of potato for intercropping, and 4 strips of either legume or potato for the sole crops). Roots of each cropping system were measured for every 10 cm interval in soil profile of 0–40 cm using metal cores of 0.0015 m<sup>3</sup> by volume (Bohm, 1979). For the monoculture system of potato and legumes, root sampling was done between two strips of potato (at approximately 37.5 cm from the strips) and two strips of legumes (at approximately 25 cm from the strips) respectively. For intercropping, the

cores were driven between two potato rows in potato strips (at approximately 37.5 cm from the strips), between two legume rows in legume strips (at approximately 25 cm from the strips) and between potato and legumes in potato-legume strips (at approximately 37.5 cm from the strips). Additional root sampling was done directly at legume and potato stem base (about 5 cm from the stem) to take into account roots not extending to the inter-row space.

Samples for a given plot and sampling depth were pooled and analyzed as one. The extracted soil cores were whirled in running water to remove the embedded soil particles and dead roots, and the mixture sieved through 2 mm mesh. For shoot biomass estimations, the plants were cut at about 2 cm from the soil surface using machetes. Fresh weight of the biomass was quantified using a weighing balance. The dry mass of roots and shoots were separately determined by oven drying about 500 g samples at 65°C to a constant mass then extrapolating to kg ha<sup>-1</sup>. The rest of biomass was incorporated in respective plots at start of the subsequent season using hand hoes.

# 5.2.6 Estimations of crop biomass lignin, polyphenols, C and N contents

Dry matter subsamples of roots and shoots were analyzed for C, N, lignin and polyphenol contents. For plant lignin content, about 20 g sample of the dry matter was ground and passed through 30 mm mesh (Van Soest and Wine, 1968). The sample was placed in a crucible and an acid detergent fiber prepared on 1 g sample added. This was followed by addition of saturated potassium permanganate (5 mL) and the content stirred with a glass rod. The crucibles were allowed to stand at 25 °C for about 90 min and the content dried overnight at 100 °C and weighed. The loss in weight of the acid-detergent fiber was taken to represent the lignin content. Polyphenols were extracted according to modified Anderson and Ingram (1993) method using methanol (70%), ascorbic acid (0.05%) and formic acid (0.5%). The dry matter samples of each plant parts were digested and used for determination of total N concentrations (Bremner and Mulvaney, 1982),

while a part of it was oxidized for determination of total organic carbon content (Nelson and Sommers, 1996).

# 5.2.7 Particle size separation and sequential density partitioning of soil organic matter

A combined particle and density size fractionation was used to partition the soil organic matter (SOM) into size fractions (Camberdella and Elliot, 1992; Sohi et al. 2001) (Fig. 5.1). Fifty (50)g of the composite soil was dispersed in 50 mL of 5% sodium hexametaphosphate and subjected to overnight shaking. The content was passed through series of sizes 250-50  $\mu$ m on a mechanical sieve shaker. Sand fractions were retained on sieves >50  $\mu$ m while silt and clay fractions (<50 µm) were repeatedly siphoned off. Density separates were recovered by sodium polytungstate (SPT) following a procedure modified from Sohi et al. (2001). A deionized water was used to suspend 50 g of the soil which was then decanted to retain the organic material. The content was transferred into a centrifuge bottle (250 mL) followed by addition of 125 mL SPT solution adjusted to a density of  $1.65 \text{ g cm}^{-3}$ . The mixture was gently shaken and allowed a period of 30 min for settling and centrifuged at 2000 rpm for another 30 min. The supernatant was sieved (0.50 µm) and the residue rinsed with deionized water and referred to as light fraction organic matter (LFOM). Forty (40) ml of SPT solution of 1.85 g cm<sup>-3</sup> density plus 10 glass beads were added into the settled residue and centrifuged at 175 rpm for about 18 h. The supernatant was filtered, and the retained residue rinsed to obtain occluded particulate organic matter. The material which settled was placed in SPT of density adjusted to 2.60 g cm<sup>-3</sup> and shaken for 10 min before being centrifuged at 3000 rpm for about 30 min.



Figure 5.1: Schematic diagram of the combined aggregate size and density fractionation.

The supernatant was decanted, the residue retained and repeatedly washed with deionized water to remove the leftover SPT. This fraction was referred to as the heavy fraction organic matter (HFOM) (stable SOM). Soil samples obtained from the aggregate sizes and density size fractions were ground and used for determinations of total SOC and total N contents. Contents of dissolved organic matter (DOM) were estimated by incubating (at 20°C for 3 weeks) about 10 g soil maintained at 55% water-holding capacity (Smolander and Kitunen, 2002).

# 5.2.8 Soil microbial biomass

The microbial biomass C (MBC) content of each aggregate size and density fractions were fumigated using chloroform (Vance et al. 1987). About 20 g of the samples was subjected to ethanol-free chloroform for 48 h and kept in the dark at 25 °C for 16 h. A non-fumigated sample of the same weight was subjected to similar conditions. About 50 mL of  $K_2SO_4$  (0.5 M) was added to the two sets of samples and shaken for 30 min to obtain extracts. Carbon content of the extracts (MBC) was oxidized with Mn<sup>3+</sup> and analyzed spectrophotometrically at 600 nm (Bartlett and Ross, 1988). The MBC contents were estimated using Eq. (5.1).

$$Microbial \ biomass - carbon \ (MBC) = \frac{SOC_{fumigated} - SOC_{non-fumigated}}{0.33}$$
(5.1)

## **5.2.9** Microbial respiration

About 10 g moist soil sample obtained from each plot at depth interval of 10 cm to soil profile 40 cm were separately measured in a vessel placed in a corked glass jar. A parafilm pricked with holes was used to cover the jars to allow for aeration during 7 days pre-incubation. After this period, the content was incubated for 7 days in the dark at 20°C. The evolved  $CO_2$  was trapped in 15 mL sodium hydroxide (0.2 N) placed into an airtight glass jar. Precipitation of carbonate ions was done by adding excess barium chloride (1.5 M) to the sodium hydroxide solution and the excess sodium hydroxide titrated with hydrochloric acid (0.2 N) using phenolphthalein indicator. The  $CO_2$  absorbed during handling and titration was corrected by the blank readings (Alef, 1995). The metabolic quotient (q $CO_2$ ) was calculated using Eq. (5.2):

$$qCO_2(g \text{ biomass } C)^{-1}d^{-1} = \frac{\mu gCO_2 C \text{ evolved in 7 days } (g \text{ soil}^{-1})}{\mu g \text{ biomass } C \text{ after 7 days } (g \text{ soil}^{-1})} x \text{ 1000}$$
(5.2)

#### 5.2.10 Estimation of enzyme activities

## 5.2.10.1 Fluorescein diacetate hydrolysis

Hydrolysis of fluorescein diacetate (FDA) was performed according to procedures outlined by Adam and Duncan (2001). About 2 g fresh soil (sieved, 2 mm) obtained from each experimental plot was separately transferred into a conical flask (50 mL) followed by 20 mL of K<sub>3</sub>PO<sub>4</sub> buffered at pH 7.4. Addition of stock solution (FDA, 1000  $\mu$ g/mL) was ensured to trigger the reaction. The flasks were stoppered, transferred to an orbital incubator subjected to temperature of 30°C and shaken at 120 rpm for 15 min. After their removal from the incubator, about 15 mL chloroform (2:1 v/v) was added to the samples to end the reaction. The content was centrifuged at 1000 rpm for 10 min, the overlying liquid filtered through Whatman paper No 2 and measured on a spectrophotometer at 490 nm. A calibration graph was used to extrapolate the concentration of fluorescein released during the assay.

#### **5.2.10.2 Dehydrogenase activity**

Dehydrogenase activity (DHA) was determined following procedures outlined by García et al. (1997). Briefly, air-dry soil sample of about 1 g maintained at 60% water holding capacity (WHC) was subjected to 0.2 mL of 0.4% INT (2-p iodophenyl-3-pnitrophenyl-5-phenyltetrazolium chloride). The content was placed in distilled water maintained at 20°C and kept off from light for about 20 h. The content was shaken vigorously for 1 min in 10 mL of methanol and filtered. The extract (iodo nitro tetrazolium formazan (INTF)) was spectrophotometrically measured at 490 nm.

#### **5.2.11 Data analysis**

The data was analyzed using version 3.5.2 of R software (R Core Team, 2017). Mixed model analysis of variance was used to test the treatment effects on soil organic matter fractions

and microbial activity. Cropping system and agro-ecological zones were considered as fixed factors while the variables assessed (MBC, DOM, LFOM, HFOM, DHA, FDA, SMR, and qCO<sub>2</sub>) were considered random factors. Tukey's honest significant difference (HSD) test was used for treatment mean separations with threshold probability level set at  $p \le 0.05$ . Principal component analyses (PCA) were calculated to examine relationships between the SOM fractions (MBC, DOM, LFOM, and HFOM), microbial activity (DHA, FDA, SMR, and qCO<sub>2</sub>), soil properties (soil temperature, soil moisture, soil texture and aggregate size), cropping systems and agro-ecological zones (UM, LH and UH).

#### **5.3 Results**

## 5.3.1 Distributions of rainfall and air temperature

The amount and distribution of rainfall and air temperature measured during this study period is shown in Fig. 5.2. In the short rains of 2016, total rainfall amount of 266, 312 and 416 mm was received respectively in the upper midland (UM), lower highland (LH) and upper highland (UH) agro-ecological zones. This is compared to 355, 575 and 581 mm recorded respectively during the 2017 long rains. The 2017 short rains totaled to 318, 355 and 412 mm respectively in the UM, LH and UH agro-ecological zones. These rains occurred mainly during the vegetative growth stage irrespective of the seasons and agro-ecological zones. Across the seasons, the air temperatures were higher in the UM (18.9–31.2°C), intermediate in the LH (13.6–28.1°C) and lowest in the UH agro-ecological zones (11.7–21.3°C).



Figure 5.2: Rainfall and air temperature recorded for upper midland (a), lower highland (b) and upper highland agro-ecological zones at different potato development stages: sprout development (I), tuber initiation (II), tuber bulking (III) and tuber tuber maturation (IV).

# 5.3.2. Biomass accumulation and residue composition

Significant differences between the treatments were observed for cumulative shoot biomass which were greater in intercropping (3433–6640 kg ha<sup>-1</sup>) and lowest in sole potato (1808–5250 kg ha<sup>-1</sup>) across the agro-ecological zones (Table 5.2).

Table 5.2: Accumulative residue biomass, carbon and nitrogen supply, residue chemical composition and carbon to nitrogen ratio by different cropping systems.

Agro- ecological zone	Cropping system	Shoot DM	Root DM	Root C	Shoot C	Root N	Shoot N	Lignin	Polyphenol	Root C/N	Shoot C/N
						kg ha <sup>-1</sup>					
Upper midland	Potato + dolichos	1811c	5271d	880c	699c	38.2b	38.2bc	69.4ab	19.4a	23.8a	18.9a
	Potato + lima bean	1573b	3433b	571b	592b	32.6b	26.3b	75.0bc	25.7a	24.4a	19.9a
	Sole dolichos	2432e	7502e	1125e	989e	64.1c	44.7c	52.9a	18.3a	26.3a	16.2a
	Sole lima bean	2312d	4852c	989d	769d	46.9b	38.6bc	66.2ab	23.6a	26.6a	16.9a
	Sole potato	389a	1808a	120a	311a	10.5a	2.4a	91.2c	29.1a	28.5a	25.0b
Lower	Potato + dolichos	2980b	6611d	1282c	933c	48.8b	52.5c	62.4b	21.3a	24.4a	19.1a
highland	Potato + lima bean	2880b	4840b	799b	795b	42.4b	31.3b	70.0c	24.8a	25.5a	19.7a
	Sole dolichos	3731c	8941e	1511e	1347e	80.8d	52.5c	45.4a	19.9a	28.8a	16.9a
	Sole lima bean	3611c	6290c	1339d	1065d	60.1c	49.8c	61.5b	23.6a	26.9a	17.7a
	Sole potato	746a	2850a	164a	423a	15.9a	5.5a	86.2d	27.1a	29.8a	26.5b
Upper	Potato + dolichos	3730c	5830c	1683c	799b	38.4b	67.3c	64.6b	21.4a	25.1a	20.1a
highland	Potato + lima bean	4681d	6640d	1121b	1113c	57.9c	63.3c	65.3b	23.2a	26.6a	19.2a
	Sole dolichos	1811a	4441a	2108e	678ab	35.9b	41.2b	51.4a	19.9a	26.6a	18.9a
	Sole lima bean	5411e	8091e	1947d	1491d	89.2d	75.7d	62.2b	20.6a	25.7a	16.7a
	Sole potato	2520b	5250b	228a	592a	21.4a	7.9a	83.6c	25.1a	28.8a	27.5b

Means followed by different letters within a column indicate significant differences by Tukey's  $p \le 0.05$ .

Similar trend was observed for root biomass which ranged between 1573 and 4681 kg ha<sup>-1</sup> in intercropping and 389 to 2520 kg ha<sup>-1</sup> in sole potato. Lima bean exhibited significantly higher root and shoot biomass in the UH zone compared to dolichos. The reverse was observed in the UM and LH agro-ecological zones. The shoot and root carbon and nitrogen supplied were greatest in sole legumes, intermediate in intercropping and lowest in sole potato. Similar trend was observed for total lignin, polyphenol contents and C/N ratios.

# 5.3.3 Distributions of aggregate sizes and soil organic matter fractions

Aggregate distributions differed significantly between the treatments for the macro and micro-aggregates and were significantly greater in intercropping relative to sole potato stands (Fig. 5.3). For silt plus clay fractions, aggregate distributions showed statistical similarities among the treatments irrespective of the agro-ecological zones. Generally, for a given treatment, the distributions of aggregates were proportionally greater in silt plus clay size particles (43–53%) than in the micro (18–28%) and macro-aggregates (18–36%) irrespective of treatments.

The contents of light fraction organic matter (LFOM), microbial carbon (MBC) and dissolved organic matter (DOM) were significantly greater in macro-aggregate, intermediate in micro-aggregate and lowest in silt plus clay particles irrespective of the treatments (Fig. 5.4). In the UM and LH agro-ecological zones, potato-dolichos intercropping recorded significantly higher contents of LFOM, MBC and DOM in the macro and micro aggregates compared to sole potato and potato-lima bean intercropping. This trend was reversed in the UH where significantly higher contents of LFOM, MBC and DOM within the micro-and macro aggregates were found in potato-lima bean intercropping. This trend was reversed in the UH where significantly higher contents of LFOM, MBC and DOM within the micro-and macro aggregates accounted for 39–55% of the total SOM, this was followed by DOM within the macro-aggregates (22–38%) and MBC which comprised 5–9.6% of the total SOM.

For the HFOM, highest contents of organic matter were recorded in silt plus clay followed by micro-aggregate and lowest in the macroaggregates. This aggregate fraction exhibited statistically similar organic matter content between the treatments irrespective of agroecological zones.



Figure 5.3: Soil aggregate distribution (0–40 cm depth) at the end of the study under different cropping systems and agro-ecological zones (upper midland (UM), lower highland (LH), upper highland (UH). Bars with different letters within an aggregate class are significantly different at Tukey's  $p\leq0.05$ .



Figure 5.4: Distributions of SOM fractions (Microbial biomass carbon-MBC, light fraction organic matter-LFOM, dissolved organic matter-DOM, and heavy fraction organic matter-HFOM) within aggregate classes ( $\mu$ m) under different cropping systems and agro-ecological zones. Bars with different letters for a given aggregate-size differ significantly between treatments (Tukey's  $p \le 0.05$ ).

# 5.3.4 Effect of cropping systems and sampling depths on distributions of SOM fractions

Labile fractions of SOM (MBC, LFOM and DOM) varied significantly among the treatments and were consistently greatest in sole legumes, intermediate in intercropping and lowest in sole potato (Table 5.3).

Table 5.3: Soil organic matter fractions: heavy fraction organic matter (HFOM), light fraction organic matter (LFOM), dissolved organic matter (DOM) and microbial biomass carbon (MBC) measured in soil depth 0–40 cm under different agro-ecological zones.

Soil		Upper midland					Lower hi	ghland			Upper highland			
depth	Cropping system	HFOM	LFOM	DOM	MBC	HFOM	LFOM	DOM	MBC	HFOM	LFOM	DOM	MBC	
(cm)		mg C kg <sup>-1</sup>												
0–10	Potato + dolichos	423a	412b	335b	198c	515a	682b	501b	288b	525a	606a	635b	203a	
	Potato + lima bean	409a	487b	346b	106b	508a	646b	512b	279b	546a	922c	774c	309b	
	Sole dolichos	432a	584d	445c	242d	525a	806c	673c	315c	520a	729b	633b	228a	
	Sole lima bean	408a	509c	432c	208c	491a	813c	605c	292b	551a	983bc	750c	402c	
	Sole potato	401a	298a	240a	98a	488a	435a	334a	102a	549a	564a	417a	212a	
10-20	Potato + dolichos	432a	432b	301c	167bc	514b	566b	399b	222b	508a	529b	315c	115a	
	Potato + lima bean	443a	420b	246b	154b	500b	542b	388b	123b	522a	795c	444b	300b	
	Sole dolichos	408a	501c	330d	208d	522b	694d	482d	301d	494a	524b	434b	134a	
	Sole lima bean	431a	407b	319d	174c	443a	603c	410c	254c	501a	884c	354c	339b	
	Sole potato	399a	168a	184a	32a	439a	259a	245a	63a	445a	318a	319a	108a	
20–30	Potato + dolichos	389a	323b	169c	102b	465b	465b	234c	157b	434b	437c	247b	108b	
	Potato + lima bean	383a	312b	145b	89b	456b	471b	208b	141b	508b	591d	295b	228c	
	Sole dolichos	392a	387b	189c	167c	479b	601d	279d	240c	456a	304b	105a	98b	
	Sole lima bean	387a	356b	139b	153c	462b	504c	267d	209c	525b	604d	241b	293d	
	Sole potato	366a	106a	98a	19a	402a	162a	108a	51a	418a	201a	101a	40a	
30–40	Potato + dolichos	376a	202b	133c	76d	436a	329b	186b	102b	411a	264b	159a	52a	
	Potato + lima bean	367a	189b	121c	45b	414a	301b	169b	82b	465a	275b	210b	129bc	
	Sole dolichos	387a	245c	129bc	89d	442a	378b	199b	141c	432a	301c	146a	103b	
	Sole lima bean	354a	234c	116b	64c	406a	342b	187b	96b	467a	424d	207b	143c	
	Sole potato	343a	87a	96a	18a	400a	143a	101a	49a	448a	165a	167a	45a	

Means with different letters within a column and soil depth denote significant differences (Tukey's  $p \le 0.05$ ), ns, not significant.

Heavy fraction organic matter (HFOM) however showed statistical similarities between the treatments and ranged between 343 and  $551 \text{ mg C kg}^{-1}$  across the agro-ecological zones and sampling depths. General decrease in SOM contents with increasing soil depth was observed irrespective of the treatments and agro-ecological zones. Cropping systems, soil depth and their interactions had highly significant (p < 0.001) effect on labile fractions of SOM. The HFOM was affected by sampling depth, but not by cropping system.

# 5.3.5 Microbial biomass carbon to total SOC ratio

Carbon to nitrogen ratio of microbial biomass and total SOM (MBC/SOC) were significantly influenced by cropping systems and generally decreased with increasing soil depth (Table 5.4).

Table 5.4: Microbial biomass carbon (MBC) to total soil organic matter carbon (SOMC) ratio as influenced by cropping systems in soil profile 0–40 cm under different agro-ecological zones.

	Upper midland					ver highla	and		Upper highland			
Soil depth (cm)	0–10	10–20	20–30	30–40	0–10	10–20	20–30	30–40	0–10	10–20	20-30	30–40
	MBC to SOMC ratio											
Potato + dolichos	4.8cd	4.1cd	3.7b	3.5c	5.2bc	4.5c	4.3b	4.1b	4.4a	3.7b	3.5b	3.3b
Potato + lima bean	4.5bc	3.8b	3.6b	3.4bc	5.0b	4.2b	4.1b	3.9b	5.6b	5.1d	4.6c	4.2c
Sole dolichos	5.2e	4.5e	4.1c	3.3b	5.5c	4.8d	4.6c	4.4c	4.2a	3.1a	3.0a	2.9a
Sole lima bean	4.9de	4.3de	3.9bc	3.2b	5.0b	4.6cd	4.3bc	4.0b	5.8b	5.1d	4.8c	4.4c
Sole potato	3.8a	3.1a	2.9a	2.5a	4.1a	3.4a	3.2a	2.9a	4.3a	4.1c	3.2a	3.0ab

Means with different letters within a column denote significant differences (Tukey's  $p \le 0.05$ ).

The ratios were significantly greater in intercropping relative to sole potato irrespective of the agro-ecological zones. In the UM and LH agro-ecological zones, dolichos recorded notably higher MBC/SOC ratio compared to lima bean, a trend that was reversed in the UH agro-ecological zone. The ratios generally tended to level off at 30–40 cm soil depth.

# **5.3.6 Microbial activities**

Microbial activities measured by soil microbial respiration (SMR), metabolic quotient (qCO<sub>2</sub>), fluorescein diacetate (FDA) and

dehydrogenase (DHA) hydrolyses, differed significantly (p < 0.01) between the treatments (Table 5.5).

Table 5.5: Soil microbial respiration (SMR-mg kg<sup>-1</sup> d<sup>-1</sup>), metabolic quotient (qCO<sub>2</sub>- mg CO<sub>2</sub>-g biomass C<sup>-1</sup> d<sup>-1</sup>) and enzyme activities (fluorescein diacetate hydrolysis (FDA-g FDA/g)) and dehydrogenase (DHA-INTF/g) in response to cropping systems and soil depth under different agro-ecological zones.

Soil	Cropping	Upper midland				Lower h	ighland		Upper highland				
depth (cm)	System	FDA	DHA	SMR	qCO <sub>2</sub>	FDA	DHA	SMR	qCO <sub>2</sub>	FDA	DHA	SMR	qCO <sub>2</sub>
0–10	Potato + dolichos	111.8b	50.4b	20.3c	44.5	130.8c	68.9c	25.5bc	42.2b	101.2a	50.9b	18.2a	40.1b
	Potato + lima bean	108.8b	49.2b	16.6b	44.1b	116.9b	58.6b	22.2b	41.8b	136.6b	69.9c	24.6b	38.4a
	Sole dolichos	148.9d	58.9c	22.2c	40.4a	154.8d	79.6e	28.6c	38.9a	97.2a	40.8a	17.7a	41.1bc
	Sole lima bean	133.6c	56.8c	20.3c	43.5b	137.3c	70.4d	25.8b	39.2a	148.8b	79.9d	28.9b	38.2a
	Sole potato	88.7a	35.5a	12.2a	48.9c	89.1a	42.8a	15.9a	45.6c	105.2a	53.6b	18.9a	42.2c
10–20	Potato + dolichos	98.7b	48.6c	15.5b	40.4a	120.1b	60.8b	22.3bc	40.4b	96.7a	41.1a	15.5a	39.9b
	Potato + lima bean	90.3b	44.4b	15.1b	42.4b	114.7b	55.7b	19.5b	40.1b	131.3b	65.3b	23.4b	37.7a
	Sole dolichos	133.7c	52.8d	18.9c	39.8a	144.5c	77.6d	25.2d	37.7a	93.2a	36.1a	16.6a	40.2b
	Sole lima bean	128.2c	49.7cd	17.8c	39.2a	137.8c	68.2c	23.3cd	38.7a	139.9b	73.8b	24.7b	37.4a
	Sole potato	80.5a	30.2a	9.1a	44.4c	84.8a	36.7a	10.3a	43.9c	97.8a	45.7a	16.2a	40.4b
20–30	Potato + dolichos	88.6b	44.7b	12.4c	39.1b	113.6b	52.8bc	17.4bc	37.3b	90.1a	40.5a	15.4bc	38.2b
	Potato + lima bean	82.3b	42.6b	9.4b	40.2b	99.7b	48.9b	15.5b	38.2b	128.3b	63.3b	18.8cd	38.1b
	Sole dolichos	121.5d	48.6c	13.4c	35.5a	131.9c	63.4d	20.2d	35.8a	86.7a	33.7a	13.4a	39.3bc
	Sole lima bean	110.6c	44.4b	12.1c	36.6a	108.5b	54.8c	18.8cd	35.7a	131.9b	67.9b	20.3d	35.2a
	Sole potato	70.6a	25.7a	7.2a	43.1c	73.8a	29.7a	9.9a	40.8c	88.9a	38.8a	13.9a	40.3c
30-40	Potato + dolichos	83.7c	38.8cd	11.9c	36.6b	90.3b	45.7bc	13.3b	37.1b	75.9a	37.9b	12.8a	37.7b
	Potato + lima bean	70.0b	34.6b	8.2b	37.5b	84.7b	42.4b	11.1b	37.4b	121.7c	54.7c	16.4b	37.6b
	Sole dolichos	102.8c	40.9d	11.8c	34.4a	107.7c	55.4d	16.6b	35.1a	85.8b	27.9a	12.3a	38.2b
	Sole lima bean	94.6c	36.8bc	9.1b	36.6b	103.5c	48.7c	14.4b	36.6a	127.5c	58.8c	17.9b	34.1a
	Sole potato	55.9a	22.4a	6.3a	40.4c	72.6a	25.7a	7.8b	39.4c	74.2a	28.1a	10.4a	39.5b

Means with different letters within a column and soil depth denote significant differences (Tukey's  $p \le 0.05$ ), ns, not significant.

Contents of FDA and DHA in the soil were significantly greater in sole legume plots than in intercropping and lowest in sole potato plots irrespective of agro-ecological zones. The enzyme activities generally decreased with increasing soil depths and were variable across agro-ecological zones. In the UH and LH agro-ecological zones, the FDA and DHA hydrolyses were greatest in sole dolichos and lowest in sole potato plots across the four soil depths. This trend was reversed in the UH agro-ecological zone where sole lima bean recorded significantly higher FDA and DHA contents relative to dolichos.

Soil microbial respiration (SMR) followed similar trend to that of FDA and DHA with sole potato plots consistently showing the lowest value across sampling depths and agro-ecological zones. Metabolic quotient (qCO<sub>2</sub>) was significantly greatest in sole potato, intermediate in intercrops and lowest in sole legumes regardless of sampling depth. Across agro-ecological zones, both cropping systems and soil depth had significant effect on qCO<sub>2</sub>. All the microbial activity parameters were notably higher in the UH than in the LH and were lowest in the UM agroecological zone.

#### 5.3.7 Principal component analyses of factors influencing soil organic matter dynamics

Principal component analyses showing factors influencing SOM dynamics is presented in Fig. 5.5. The two principal components explained 74% of the variability in SOM fractions with the first component accounting for 56% of the variance and the rest (18%) being accounted for by the second component. The first component was strongly correlated with 10 of the original variables (p = 0.000) and most strongly with the microbial biomass carbon-MBC (r = 0.98). This component was therefore a measure of soil microbial biomass content and increased with increasing dissolved organic matter-DOM (r = 0.80), fluorescein diacetate-FDA (r = 0.80), dehydrogenase activity-DHA (r = 0.78), soil water content-SWC (r = 0.70), light fraction organic

matter-LFOM (r=0.65), metabolic quotient-qCO<sub>2</sub> (r=0.61) and microbial respiration-SMR (r=0.55), but negatively with clay content (r=-0.96), aggregate size (r=-0.94), agro-ecological zone (r=-0.80) and soil temperature (r=-0.57). The second component was majorly a reflection of cropping system (r=0.66), which related strongly with metabolic quotient-qCO<sub>2</sub> (r=-0.60), soil temperature (r=-0.58), soil water content-SWC (r=0.50) but weakly with FDA (r=-0.49), agro-ecological zone (r=-0.49), DOM (r=0.47) and DHA (r=-0.38).



Figure 5.5: Principal component analyses of factors influencing SOM dynamics in potato-legume intercropping. SWC, soil water content, DOM, dissolved organic matter, LFOM, light fraction organic matter, MBC, microbial biomass carbon, qCO<sub>2</sub>, metabolic quotient, FDA, fluorescein diacetate, DHA, dehydrogenase activity, SMR, microbial respiration. Vectors indicate the degree of correlation between each factor and the axes.

The interrelationship of the components revealed a direct positive association between soil FDA, DHA and MBC. Soil DHA however exhibited a weaker correlation with LFOM, MBC and DOM

compared to soil FDA. There was a significant positive association between qCO<sub>2</sub>, soil DHA and FDA hydrolysis. Further, there was a positive association between the aggregate sizes with soil labile organic matter fractions (LFOM, MBC and DOM), but a negative correlation exhibited between the clay content, microbial biomass and SMR. The activity of DHA, soil FDA hydrolysis and SMR related negatively with soil temperature, but positively with soil moisture content.

#### **5.4.** Discussions

#### 5.4.1. Distributions of aggregate sizes and soil organic matter fractions

The high variability in sizes of macro-aggregate distributions between the treatments reflected greater effect of residue retention on sand fractions relative to micro-aggregate and silt plus clay fractions. Intercropping provided readily-degradable substrates which were decomposed by the soil microbes releasing cementing agents that glued the soil particles to form micro-aggregates. The micro-aggregates were then bound with SOM and silt plus clay particles to form macro-aggregates. This effect was influenced by residue quantity that was higher in intercropping than in sole potato. When the residue supply by legume intercropping attained a threshold level, most of it remained unprotected and was highly accessible to degrading soil microbes and their enzymes. This triggered residue degradation fostering aggregate formation. This process signified improved soil physical quality such as nutrient exchange, soil water movements and aeration (Jäger et al. 2011). These conditions were favorable for microbial growth and activities.

The content of microbial biomass differed significantly among the size fractions, a reflection that different aggregate classes provided distinct microhabitats packed with organic carbon of different qualities and quantities. The macro-aggregates had the greatest contents of MBC suggesting high population of soil microorganisms. In addition, the macro-aggregates had the highest MBC/SOC ratio irrespective of the cropping system indicating high efficiency by

which the soil microbes used the carbon substrates (Anderson and Domsch, 2010). A possible reason for this observation relates to the fact that the macro-aggregate size fraction is composed primarily of partially decomposed litter that provided sources of energy to the microorganisms. Intercropping increased the contents of MBC by inputting large quantity of litter and roots thus enhancing accumulation of DOM and LFOM which provided sources of nutrition to soil microorganisms. This observation is in keeping with other studies that found high diversity, functionality and spatial distribution of fungi and bacterial populations within the macro-aggregates (Poly et al. 2001; Mummey et al. 2006). They attributed it to the fact that macro-aggregates exhibit high stability of soil water potential and are physically protected from the predators.

The microbial biomass was however significantly greater in intercropping relative to sole potato for the macro-aggregate fractions. This observation is consistent with the biomass quantity and quality which was higher in intercropping than in sole potato. The residues from legume intercrops had higher nitrogen content that increased the supply of nutrients to the soil microbes. Generally, the microbial biomass is activated by continuous addition of substrates, resulting in increased production of extra-cellular enzymes that enhance the microbial activity (Loeppmann et al. 2016). This observation was depicted in our study which showed increasing fraction of MBC which had preferential use for soil labile organic carbon as source of energy. Rasmussen and Rohde (1988) showed that low residue return to the soil results in nitrogen immobilization that adversely impacts on the microbial population density and diversity. Moreover, the soil particles obtained from intercropping had greater MBC/SOC and exhibited lower metabolic quotient (qCO<sub>2</sub>) relative to those in the sole potato. This indicates that legume intercropping fostered the growth of soil microorganisms and enhanced the efficiency by which the substrates were used as energy sources.

The silt plus clay fractions had the least MBC/SOC regardless of the treatments indicating the lowest carbon use efficiency. This observation was ascribed to the low SOM accessibility contributed by sorption of carbon on the clay colloids.

The high contents of light fraction organic matter (LFOM) within the macro-aggregates was a reflection of high residue transformation following addition of fresh residue to the soil. This fraction resided within the macro-aggregates which remained unprotected by physical mechanisms thus allowing accessibility of degrading microorganisms and their enzymes. Due to the fact that potato had relatively higher lignin and polyphenol concentrations and exhibited higher C/N ratios, we expected to find higher contents of partially decomposed litter (LFOM) in sole potato plots. This was not the case as intercropping generally exhibited higher LFOM contents. This observation is inferred to the low residue deposition by potato that stimulated breakdown of the more resistant litter transforming it into refined dissolved organic matter (DOM). Intercropping however exhibited low quantity of polyphenols and lignin compounds and had lower C/N ratios making their residues to undergo rapid decomposition with increased partitioning of organic matter to the less protected macro-aggregates.

In contrast, the LFOM and DOM content attached to clay plus silt fractions was similar between the treatments implying that these fractions were not affected by legume intercropping. For the micro-aggregates, the SOM was encapsulated within the macro-aggregates making it more protected by physical mechanisms and was thus more stabilized and resistant to further decomposition. Similarly, no response to legume intercropping was observed for the heavy fraction organic matter (HFOM) attached to the silt plus clay fractions as these particles exhibited charged surfaces that adsorbed the SOM making it physically, chemically and biochemically protected from the microbial attack (Six et al. 2002). This fraction represented >90% of the silt plus clay fractions reiterating the importance of these particles in carbon stabilization.

The contents of labile SOM fractions (LFOM, MBC and DOM) were significantly higher in the surface soil layer (0–10 cm) irrespective of the treatments. This observation was related to the high residue concentrations within the topsoil as the hand hoes had limited capacity to incorporate the residues beyond this depth. This is consistent with other authors who argued that the more labile SOM largely depend on the tillage depth and the quantity of organic residues incorporated (Somasundaram et al. 2018). Compared to sole potato, intercropping recorded significantly higher contents of labile SOM fractions (LFOM, MBC and DOM) in the 30 to 40 cm depth, an observation we related to the high root biomass contributed by legume intercrops. Generally, up to half of the labile SOM is found in the roots (Gregory and Atwell, 1991). The fact that roots are produced directly within the microbial habitat (Merino et al. 2015), may have triggered interactions that directly affected the structure and function of decomposer organism, potentially enhancing MBC biomass. In addition, roots exude labile soil carbon compounds that promote efficiency of microbial growth resulting in increased production of MBC (Kong et al. 2011).

Despite the general decrease in HFOM with depth across the treatments, only a slight variation was observed between treatments. This was a confirmation that this organic matter fraction is bio-chemically stabilized and resistant to further decomposition, irrespective of soil depth. Nyawade (2015) postulated that HFOM generally decreases with increasing depth due to the increasing contents of Al and Fe-oxides in soils dominated by kaolin minerals such as those in this study.

#### 5.4.2. Effect of legume intercropping on soil microbial activity

The fluorescein diacetate hydrolysis (FDA) was increased by legume intercropping suggesting enhanced microbial activities. This observation relates to the high residue deposition and transformation that occurred in these plots thus providing large energy source to the soil microbes. A similar result measured for dehydrogenase (DHA) revealed a higher microbial activity which likely was contributed by the high flux of root secretions and root exudates in the plant rhizosphere (Rakshit et al. 2012). Soil DHA and FDA hydrolysis decreased under sole potato stands suggesting a negative impact of low residue quantity on soil microbiological activity. This was probably a consequence of low energy sources for the microbial activity as reflected by the low contents measured for LFOM, DOM and MBC. Reddy et al. (2003) reported greater microbial activity as measured by FDA in soils planted with crimson legume clover for three years relative to soils subjected to non-legume rye cereal. Consistent findings were reported by Sicardi et al. (2004) who suggested that the labile SOM fractions provide readily source of energy to the soil microbes leading to enhanced microbial activity. It is for this reason that this fraction of organic matter is considered the most sensitive indicator of soil quality (Kapkiyai et al. 1999).

The high soil respiration under legume intercropping was an indication of increased rate of organic matter transformation by soil microbes. This observation was further reflected in the low metabolic quotient ( $qCO_2$ ) under legume intercropping pointing to the increased microbial respiration. As soil microorganisms breakdown residues, carbon becomes incorporated into the microbial biomass, and makes integral component of MBC, a process which is slowed by high C/N ratio (Chen et al. 2014). A mixture of high and low quality residues like in this study enhances microbial activity, causing a larger carbon proportion to be incorporated into microbial biomass (Müller-Stöver et al. 2012). This enhances the efficiency by which the soil microbes convert the

organic carbon into microbial biomass (Anderson and Domsch, 1990). In the UH agro-ecological zone, the unfavorable climatic conditions for dolichos growth led to lesser input of litter thus contributing to the low soil MBC content which increased the  $qCO_2$  and lowered the microbial respiration rates.

The decrease in microbial activity with increasing soil depth was consistent with that of the substrate availability that were high in the topsoil and steeply decreased with soil depth. Topsoil represented the dynamic nexus where majority of shoot biomass was deposited. For sole potato, root biomass was found entirely in the topsoil and thus maximum plant carbon allocated to rhizodeposition was expected in this zone (Canarini and Dijkstra, 2015). In other studies, fungal evenness and richness have been shown to be high in the topsoil and drastically decrease with increasing soil depth (Jumpponen et al. 2010; Vargas-Gastelum et al. 2015). This is in line with other studies that showed that subsoils accommodate only a specific type of soil microbes that are well adapted to the low substrate conditions (Hartmann et al. 2009). Further, the content of labile SOM that readily avails the substrates to the soil microbes was high in the topsoil compared to the subsoil that was dominated by HFOM that restricted the accessibility of substrates by the soil microbes.

The significant interactive effect observed between soil depth and agro-ecological zone on FDA hydrolyses, DHA activity and SMR reflected the variability in soil microbial activity contributed by agro-ecological conditions and soil types. The soils in LH agro-ecological zone were clay dominating and generally exhibited increase in clay content with increasing depth. Thus, sorption of SOM to the Al and Fe-oxides may have restricted the substrate availability, especially in the deeper soil layers. This environment disfavored the microbial growth, restricting their activity. In the UH agro-ecological zone, the high rainfall amounts repeatedly created water-

saturated conditions in the subsoils that lowered the availability of oxygen. This coupled with increase in Fe and Al oxides that formed mineral-organic associations with the SOM (Dietel et al. 2017), impaired the substrate availability for the soil microbes decreasing their activity. In the topsoil, the high biomass contents may have favored the activity of fungi contributing to increased microbial activity (Engelhardt et al. 2018).

#### 5.4.3 Principal factors influencing dynamics of SOM fractions

The fluorescein diacetate hydrolysis (FDA) and dehydrogenase activity (DHA) were directly proportional to the microbial biomass, an observation that was related to the high residue quantity contribution by legume intercrops (Schnurer and Rosswall, 1982). This is consistent with previous studies that showed that production of DHA and FDA hydrolysis increases with increase in residue deposition thus reflecting the increase in microbial enzymatic activity (Elfstrand et al. 2007). On the other hand, soil DHA and FDA hydrolysis exhibited positive association with LFOM and DOM reflecting that the total activity of microorganisms highly depended on light SOM fractions as their source of nutrition. García-Gil et al. (2000) described DHA as an intracellular enzyme useful for microbial oxidoreductase metabolism. The activity of this enzyme highly depends on the microbial biomass metabolic state which increases with increase in residue quality and quantity. Soil DHA however provided a weaker correlation with LFOM, MBC and DOM compared to soil FDA suggesting that the contribution of crop residues to microbial biomass activity affected greater mass of fungi as DHA does not reflect the fungal population (Kumar and Tarafdar, 2003).

The significant positive association between  $qCO_2$ , soil DHA and FDA hydrolysis reflected increase in efficiency of carbon assimilation by the soil microbes per unit amount of organic residue increase. Thus, the decrease in  $qCO_2$  observed under legume intercropping relative to sole potato plots signified a greater metabolic efficiency of the substrate contributed largely by increase in residue quality. The positive association between the aggregate sizes with soil labile organic matter fractions (LFOM, MBC and DOM), cropping systems and SMR indicated that the enhanced residue retention promoted the macro-aggregate formation thus enhancing microbial activity. The negative correlation exhibited between clay content and the microbial biomass as well as with SMR showed that the high clay content impeded the SOM mineralization by physically occluding the microbes in small pores (Rutherford and Juma, 1992), thus repressing their activity.

The activity of DHA, soil FDA hydrolysis and SMR related negatively with soil temperature but positively with soil moisture content implying inhibitory and stimulatory effects of high temperatures and increased soil moisture contents on microbial activity. Under extremely dry soil conditions, soil microbes are rapidly 'starved' because a dry soil does not have enough substrate supply, and is characterized by unfavorable microenvironment for microbial growth (Murphy et al. 1998). Steinweg et al. (2012) found soil temperature to be the dominant control of enzymatic activity when soil moisture was not limiting. Chavarría et al. (2016) similarly postulated that soil moisture and soil temperature are the major regulators of soil enzymatic activity at microplot scale.

Agro-ecological zone exhibited significant association with microbial biomass, aggregate sizes and SMR reflecting spatiotemporal variability caused by fluctuating soil temperatures and soil moisture contents. Temporal changes in soil moisture contents and soil temperature affect the amount of carbon allocated to rhizodeposits and secretions of rhizosphere products that cement soil particles into macro-aggregates (Bonde and Rosswall, 1987). The quantity of residue generation and nutrient transformation by soil microbes were besides highly dependent on the prevailing soil moisture and soil temperature conditions. Dolichos due to its inability to tolerate

low temperatures in the UH agro-ecological zone (Cook et al. 2005), was characterized by reduced biomass production. This crop however, accumulated about 2 times higher biomass compared to lima bean in the UM zone and up to 5 times higher compared to sole potato. Lima bean however, tolerated both low and high soil temperatures and accumulated high biomass across the agro-ecological zones. Potato being a temperate crop performed best in the UH zone where its biomass accumulation was highest. Previous work found that agro-ecological zone influences crop selection for biomass production in such a way that crop varieties are acclimatized to the zones favoring growth conditions (Jaetzold et al. 2012). Van Gestel et al. (1993) observed a parallel course of MBC content when increase in ambient temperatures caused desiccation to the soil microbes thus interrupting their growth.

## **5.5** Conclusion

This study gives an insight on the potential role of legume intercropping in increasing residue return thus enhancing the formation of labile soil organic matter fractions with little impact on the stable fractions. The increase in residue retaintion due to legume intercropping increased the proportion of macro-aggregates that signified improved nutrient release, high aeration and increased soil water infiltrations, the conditions of which are favorable for microbial growth and activities. These benefits may only be of primary importance to the smallholder farmers if the system is designed in a manner guaranteeing high yield stability. In this way, the ultimate goal of enhanced soil productivity will be masked, but earned by the farmer as a bonus.

#### CHAPTER SIX

# OPTIMIZING SOIL NITROGEN BALANCE IN A POTATO CROPPING SYSTEM THROUGH LEGUME INTERCROPPING

### Abstract

Negative nitrogen balance represents a major factor causing low potato yield in potato growing areas of Kenya while its excessive surplus poses a significant environmental concern. In order to synchronize this tradeoff, a field trial integrating potato (Solanum tuberosum L.) with lima bean (Phaseolus lunatus L.) or dolichos (Lablab purpureus L.) in intercropping system was conducted in the upper midland (1552 m above sea level (masl), lower highland (1894 masl) and upper highland (2552 masl)) agro-ecological zones of Kenya. The study was laid out in a randomized complete block design with four replications. Soil samples (0-120 cm depths) were taken from each plot at key stages of potato growth. Nitrogen gains from mineralization, fertilization, biological fixation, and outputs from biomass accumulation, leaching, volatilization, runoff and soil erosion were quantified using standard procedures. Soil N balance ranged from -10.7 to -18.1 kg N ha<sup>-1</sup> for sole potato, 4.1 to 6.6 kg N ha<sup>-1</sup> for intercropping and 2.9 to 22.3 kg N ha<sup>-1</sup> for sole legumes. The intermediate range of polyphenol and lignin contents in intercropping enhanced N mineralization with peak N release showing a close synchrony of N supply and potato crop N demand. Nitrate being highly mobile, was quickly leached below the active root zone in sole potato (4.2 to 46.6 kg N ha<sup>-1</sup>), a process that was diverged by the deep root systems of legume intercrops (amount of N leached, 1.1 to 24.0 kg N ha<sup>-1</sup>). These results suggest that legume intercropping can provide a means of balancing the nitrogen retained in the biomass and soil, thus offering a mechanism for optimizing the soil N balance in smallholder potato farming systems.

Keywords: Legume intercropping; nitrogen balance; nitrogen mineralization; nitrogen uptake.

# **6.1 Introduction**

Nitrogen is one of the most important nutrients constraining potato production in the subtropical smallholder farming systems (Sanchez 2002; Henao and Baanante 2006; Burke 2017). The N balances in these systems are on average negative as the N outflows often exceed the N inflows. The N inflows in a potato production system are majorly from inorganic fertilizers, organic manures or additions through biological nitrogen fixation where legumes are integrated in rotation or intercropping systems (Jégo et al. 2008; Shekofteh et al. 2013). The outflow processes include N removal by soil erosion and in harvested biomass, denitrification during prolonged periods of wet soil, and volatilization of ammonia (NH<sub>3</sub>) in high pH soils after surface granular urea application (Loomis and Connor, 1992). Leaching occurs mainly where nitrates leave the soil in drainage water (Arregui and Quemada, 2006).

Defining an appropriate fertilizer application rate that matches N supply with crop N demand is thus necessary in smallholder potato farming systems to avoid N mining. This is only possible if knowledge of the quantity of N that will be supplied by the crop residues is known. This process is defined by the soil temperature and soil water contents, residue chemical composition: lignin (L), polyphenol (P), carbon (C), nitrogen (N) and their ratios such as C-to-N, L-to-N, and L + P-to-N (Palm and Sanchez, 1991; Tian et al. 1993; Becker and Ladha, 1997; Mitchell et al. 2000). Where decomposed organic matter exhibits high C-to-N, lignin-to-N, and polyphenol-to-N ratios, N supply will not be sufficient to satisfy N requirements of soil microorganisms, resulting in net N immobilization (Lupwayi and Haque 1999; Lupwayi et al. 2006). The production and retention of available N often increases with increasing soil moisture and soil temperature to a threshold and thereafter declines (Baldock et al. 2018).

In Kenya, the low soil N content is aggravated by the poor potato cropping systems that majorly entail sole cropping with limited capacity to provide adequate soil cover and protect soil from erosion (Muthoni et al. 2013; Nyawade et al. 2018b; Gitari et al. 2018b). The nitrogen input not assimilated by potato crop accumulates in the soil as nitrate that is leached through irrigation and rainwater or lost through volatilization or denitrification (Gentile et al. 2009). To compensate for these N losses, a number of commercial farms have opted to do heavy application of inorganic N fertilizer (Kimetu et al. 2006; Muthoni et al. 2013). This strategy is too expensive for the smallholder farmers and may contribute to increased environmental pollution through eutrophication. In addition, majority of potato produced in Kenya is rainfed and grown under conditions where availability of water defines potential productivity. In such conditions, provision of an appropriate supply of N should be synchronized with temporal demand of the crop over the growing season. This is true as N makes up 1–4% of potato dry matter while too much N results in vigorous early vegetative growth leading to haying-off under water-limited conditions (van Herwaarden et al. 1998).

There is therefore a growing need to design low-cost and efficient integrated nitrogen management systems compatible with smallholder farmers' socioeconomic status. In this context, legumes or nitrogen-rich plant residues present promising alternatives to the mineral fertilizers. These crops can be simultaneously introduced into potato cropping systems. The organic residues from these plants must decompose fast enough for N to become available to potato crop. In this way, synchronization of N supply and crop N demand for a given target yield level may reduce N losses and increase N use efficiency. Intercropping that includes deep rooted crops can reduce nitrate leaching due to their ability to effectively take up residual nitrogen, converting it into crop
biomass which is later mineralized for nitrogen release (Constantin et al. 2012; Plaza-Bonilla et al. 2015).

Intercropping potato with high nitrogen fixing legumes can alleviate the high cost of inorganic soil nitrogen inputs (Burke, 2017). Sanginga and Woomer (2009) reported that velvet bean (*Mucuna pruriens* L.) accumulated about 160 kg N ha<sup>-1</sup> in 12 weeks when it was intercropped with maize, while Eaglesham et al. (1981) found that the nitrogen fixed by companion cowpea was about 41 kg N ha<sup>-1</sup> in maize-cowpea intercropping system. Roots of legumes can decompose and release nitrogen into the soil where it is made available for subsequent crops (Giller et al. 2015). Legume cover crop integration into potato cropping systems can appreciably reduce nitrogen losses due to soil erosion thereby increasing efficiency of nitrogen use (Xing et al. 2011).

Intercropping is however practiced by only about 5–12% of the smallholder farmers in Kenya and mostly with maize (*Zea mays* L.) intercrop (Muthoni et al. 2013). We therefore conducted a two-year field experiment to compare the soil nitrogen content of sole potato and potato intercropped with lima bean (*Phaseolus lunatus* L.) or dolichos (*Labalab purpureus* L.) in three AEZs of Kenya (upper midland, lower highland and upper highland). We hypothesized that (i) legume intercropping increases the amount of dry matter available for nitrogen mineralization thus increasing the soil nitrogen content available for uptake by potato crop and (b) that legume intercropping minimizes the amount of N lost to leaching and soil erosion thus increasing the soil N balance.

#### 6.2 Materials and methods

#### **6.2.1 Experimental site and design**

The trials were carried out during the four rainy seasons of 2017 and 2018 in three agroecological zones (AEZs) of Kenya; upper midland (1552 m above sea level (masl)), lower highland (1894 masl) and upper highland (2553 masl). Detailed site descriptions are given in section 3.1. During this study period, total rainfall amount of 345, 184, 328 and 216 mm was received respectively during the 2017 long rains, 2017 short rains, 2018 long rains and 2018 short rains in the upper highland AEZs. This is compared to rainfall amount of 388, 211, 372 and 269 mm received in the lower highland AEZ, and 399, 209, 369 and 234 mm respectively in the upper highland AEZ. Across the seasons, the mean air temperatures were higher in the upper midland (15.7–27.9°C), intermediate in the lower highland (13.1–25.4°C) and lowest in the upper highland (11.1–22.3°C) AEZs. Soil chemical and physical properties measured at start of the experiment are shown in Table 6.1.

Table 6.1: Initial soil chemical properties measured at different soil depth: pH; soil organic carbon (SOC); total nitrogen (N); nitrate N (NO<sub>3</sub><sup>-</sup>–N), ammonium N (NH<sub>4</sub><sup>+</sup>–N).

Agro-ecological	Soil	pH	SOC	Ν	NO <sub>3</sub> -N	$NH_4^+-N$
zone	depth (cm)			g l	кg <sup>-1</sup>	
Upper	0–30	5.0	18.2	1.3	5.8	4.7
midland	30–60	5.0	15.5	1.1	5.4	4.2
	60–90	4.9	12.3	1.1	3.8	2.9
	90–120	4.9	9.8	0.9	3.3	2.0
Lower	0–30	5.1	25.5	1.8	8.4	5.1
highland	30–60	5.1	15.6	1.1	7.7	4.6
	60–90	5.2	7.8	0.6	7.2	3.3
	90–120	5.2	2.4	0.2	6.6	2.0
Upper	0–30	5.2	36.6	2.8	9.2	5.7
highland	30–60	5.2	25.5	2.1	8.2	4.3
	60–90	5.3	13.1	1.1	8.3	3.1
	90–120	5.3	9.8	0.9	7.3	2.1

#### 6.2.2 Experimental design and crop husbandry

The trials were laid out in a randomized complete block design with four replications. The treatments comprised of sole potato (*Solanum tuberosum* L.), sole lima bean (*Phaseolus lunatus* L.), sole dolichos (*Lablab purpureus* L.) and intercrop of potato with either lima bean or dolichos. Intercropping was done in 2 rows of potato alternating with 2 rows of legumes. Fertilizer application was based on soil analysis and was adjusted seasonally taking into account the amount

of mineral N in the soil prior to planting. On average, this activity consisted of basal application of 50 kg ha<sup>-1</sup> N, 90 kg P ha<sup>-1</sup>, 100 kg K ha<sup>-1</sup> and single topdressing with 50 kg N ha<sup>-1</sup> in the form of calcium ammonium nitrate (26-0-0). Topdressing was done 15–25 days after potato emergence depending on the general soil moisture conditions. Legumes received only basal phosphorus applications (triple super phosphate, 0-46-0) at rates averaging 20 kg P ha<sup>-1</sup>. Detailed cultural activities are described in section 3.5.

#### 6.2.3 Soil sampling and analyses

Soil samples were taken in 10 replicates from inter-rows of each plot with a 15 mm innerdiameter soil piston auger at soil depth 0–30, 30–60, 60–90 and 90–120 cm. The soil samples were mixed into a composite for each depth, transported in cooler boxes and stored at 4°C until analysis. Subsamples of 20 g fresh soil weight were extracted in 1M KCl for 1 h (1 soil: 2 solution) and centrifuged. The supernatant was analyzed for  $NH_4^+$ –N and  $NO_3^-$ –N by standard colorimetric methods using Auto Analyzer 3 (Bran + Luebbe, Germany). The N content was converted to kg ha<sup>-1</sup> basis using soil bulk density values for each sampling depth.

#### 6.2.4 Soil moisture, rainfall and ambient temperature measurements

Soil moisture content was measured every 30 cm to depth of 120 cm using HOBO soil moisture sensors installed in each plot. The soil moisture in the top 30 cm soil layer was measured gravimetrically in all replications before planting to calibrate the sensor probes and to estimate the amount of water required for irrigation. Rainfall amount of each experimental site was recorded using onsite rain gauges while air temperatures were obtained from HOBO temperature sensors installed within the experimental sites.

# 6.2.5 Root sampling and estimation of root length density

Root length density was estimated as outlined in section 4.2.9 and related with the amount of N leached to discern the effect of root heterogeneity on N capture.

#### 6.2.6 Computation of nitrogen balance

Soil nitrogen balance was computed as the difference between N input and output pools within the crop root zone (0-120 cm depth) (OECD, 2001). Taking into considerations the major nitrogen input and output pathways in smallholder potato cropping systems, soil nitrogen balance was computed using Eq. (6.1):

$$N_{bal} = \left[N_{fert} + N_{initial} + N_{residual} + N_{min} + N_{fix}\right] - \left[N_{uptake} + N_{ero} + N_{leach} + N_{vol}\right] \quad (6.1)$$

where inputs:  $N_{fert} = N$  gains from fertilizer;  $N_{initial} = Soil N$  available at start of the season;  $N_{residual}$ = available soil N after harvest;  $N_{min} = N$  mineralization from crop residues, and the outputs;  $N_{uptake}$ = nitrogen accumulated in the plant;  $N_{ero} = N$  lost in runoff and N carried by eroded sediment;  $N_{leach} = N$  leached beyond the active root zone;  $N_{vol} = N$  lost to volatilization.

The contributions of biological nitrogen fixation to total N accumulation was estimated by the N difference method (Peoples et al. 2002) which is based on the fact that sources of N for fixing crops are soil, fertilizer and the atmosphere (Eq. 6.2).

$$\% N \, dfa = 100 - (\% N \, dfs + \% N \, dff) \tag{6.2}$$

Where, Ndff is nitrogen derived from the fertilizer; Ndfs, nitrogen derived from the soil and Ndfa nitrogen derived from the atmosphere.

The amount of N leached was analyzed from the soil solution extracted using SoluSAMPLER located at vertical depths of 0–150 cm at increasing interval of 30 cm (Biswas and Schrale, 2007). In this study, soil N beyond 120 cm depth was considered leached, as no active roots were found past this depth. Nitrogen content from the sampled soil solution was determined

by UV/vis spectrophotometer and indigotic colorimetric method. The N contents were converted to kg N ha<sup>-1</sup> basis using soil bulk density values measured for each sampling depth.

Nitrogen uptake by potato and legumes was computed at potato vegetative growth, tuber initiation, bulking, and maturation stages. For potatoes, samples were separated into tubers, roots and shoot while the legumes were partitioned into shoot, roots, and into grains for sampling conducted at maturity. The samples were dried in an oven at 70°C for 48 h, weighed, and subsamples analyzed for N content using Kjeldhal digestion method (Keeney and Nelson, 1982). The total N uptake per plant was calculated as the product between plant part N concentration and biomass weight.

The litterbag method (Anderson and Ingram, 1993; Verhoef, 1995) was used to measure N mineralization from crop residues. The bags measured 0.3 m by 0.3 m and were made of plastic with 1 mm mesh. At harvest of each crop, 8 bags corresponding to the number of sampling times were filled with about 250 g of crop residues from each treatment after they had been chopped into pieces 50 mm long. About 250 g soils without crop residue amendments were also added in litterbags as controls. The bags were buried at a depth of 30 cm at a distance of 30 cm between the bags along the potato-legume rows. The bags were retrieved and sampled at 2, 4, 6, 8, 10, 12, 14 and 16 weeks after residue incorporation. At each sampling time, plant materials remaining in the litter bags were oven dried at 60°C to a constant weight. The oven-dried samples were weighed separately to determine dry matter (DM) losses using Eq. (6.3).

$$Weight \ loss = 100x(M0 - Mt)/M0 \tag{6.3}$$

where

M0 is the initial plant dry matter (DM) (g) in the litterbag; Mt is the plant DM mass (g) in the bag at time t when the bags were retrieved.

At each litter-sampling time, soil samples were taken at 0–30 cm on the spots where the litterbags were laid and in control plots and analyzed for mineral N. The samples were put in airtight polythene bags, stored in the laboratory at 4°C, and analyzed within 2 days of sampling. Briefly, 20 g soil samples (passed through 0.5 mm sieve) were extracted in 60 ml 1 M KCl. The mixture was shaken on an end-over-end shaker for 1 h, centrifuged and the clear supernatant transferred into 100 ml flasks. The supernatant was analyzed for total N using Kjeldahl method. The N content was converted to kg ha<sup>-1</sup> basis using soil bulk density values measured with soil cores for each sampling interval. Net N mineralization was calculated as the difference between the amounts of mineral N released between the amended soil and the control soil.

The amount of N mineralized over time of sampling was calculated by the first-order model (Eq. (6.4)) (Jones, 1984).

$$N_m = N_0 (1 - e^{-kt}) + N_1 \tag{6.4}$$

where

 $N_{\rm m}$  (mg kg<sup>-1</sup>) is the amount of N mineralized at time *t* (d);  $N_0$ , the initial mineralizable N content (mg kg<sup>-1</sup>); *k*, the first-order rate constant (d<sup>-1</sup>) and  $N_1$  (mg kg<sup>-1</sup>), the zero-order constant (mg kg<sup>-1</sup>), i.e., N mineralized at time 0.

A subsample of 25 g soil was dried in an oven at 105°C for 24 h to determine the soil moisture content at the time of sampling. In addition, the initial materials added to the bags (0 week sampling time) were analyzed for organic C using the wet digestion procedure of Walkley and Black (Nelson and Sommers, 1996). Lignin was assessed by the acid detergent fibre method (Goering and van Soest, 1970), while the water-soluble polyphenols were quantified using procedures outlined by Folin Denis (King and Heath, 1967).

Ammonia volatilization was measured using ventilation chambers placed at inter-rows of each plot (Jantalia et al. 2012). In 2017 SR, samples were collected at 4, 11, 18, 26, 34, 50, and 64 days after planting (DAP) and at 6, 13, 23, 30, 40, 50, 59, 70 and 82 DAP in 2017 LR. In 2018 SR, the data were collected at 8, 22, 29, 36, 44, 51, 58, 71, and 86 DAP and at 6, 13, 20, 28, 40, 52, 61, 70, and 83 DAP in 2018 LR. At each sampling date, the polyfoam strips and acid solution placed in plastic cups from each chamber were collected in 125 mL of 2 M KCl solution. Fifty (50) mL of this solution was analyzed using an automated ammonia analyzer (TL 2800, Timberline Instruments). Nitrogen loss through NH<sub>3</sub> volatilization during consecutive sampling dates (mg NH<sub>3</sub>–N m<sup>-2</sup>) was determined using Eq. 6.5.

$$NH_3 \ volatilization = \frac{NH_3 - N \ conc \ (mg \ mL^{-1})* \ total \ vol \ of \ solution \ (250 \ mL)}{Surface \ area \ of \ the \ soil \ covered \ by \ the \ trap \ (0.90 \ m^2)}$$
(6.5)

Cumulative  $NH_3$  volatilization (kg N ha<sup>-1</sup>) was determined by summing up the amount of  $NH_3$  volatilized during each sampling period throughout the growing season using Eq. 6.6.

Cumulative NH<sub>3</sub> emission = 
$$\sum_{i}^{n} \frac{X_i + X_{i+1}}{2} (X_{i+1} - t_i)$$
 (6.6)

where Xi is the NH<sub>3</sub>–N flux measurement on day t, Xi + 1 is the succeeding NH<sub>3</sub>–N flux measurement on day ti+1, and n is the final date of NH<sub>3</sub>–N flux measurement.

Runoff and sediment samples were collected using soil sedimentation containers installed at the lowest edge of each experimental plot. The runoff water collected was overflowed into a second container with a weir. The water level in this container was monitored at 30 min intervals by a GY-type sensor (UIZ-ECH20, UIZIN, Tokyo, Japan) and converted into water discharge to determine water runoff. The runoff water samples (about 15 mL) were transferred into 20-mL vials followed by a drop of 0.5M sulfuric acid and transported in ice water (4°C) within 24 h to the laboratory and chilled at -4°C till analyses (Eq. (6.7)). The sediments were weighed after drying in an oven at 70°C. A sample of air-dried sediment from each plot (about 100 g) was collected in 100 ml can and analyzed for total N (Eq. (6.8)).

$$N_{runoff} (kg ha^{-1}) = Runoff N(mgl^{-1}) * runoff depth(mm) * 0.253$$
(6.7)

 $N_{eroded sediment}(kg ha^{-1}) = \text{soil loss}(kg ha^{-1}) * \text{ sediment } N$ (6.8)

Residual soil nitrogen was determined by collecting soil from the center of each plot at 0– 30 cm m at potato harvest. The samples were transferred to the laboratory at 4°C, stored at -18°C, and analyzed for N within 1 week, following Kjedlhal method (Nelson and Sommers, 1982). Corresponding bulk density of soil was measured from an intact soil core and used to convert mg of N kg<sup>-1</sup> soil to kg N ha<sup>-1</sup>.

# 6.2.7 Data analysis

The data was analyzed using R software, version 3.5.2 (R Core Team, 2017). The treatment effects on soil water content, soil temperature and N balance were tested using general linear model analysis. Whenever interaction of cropping system and season was found significant, data were analyzed in separate seasons. Tukey's honest significant difference test was applied for multiple mean comparisons between treatments and tests with p < 0.05 were considered statistically significant. To derive the mineralization rate constants (k), N mineralization data was fitted to the kinetic model by data processing software KyPlot (KyensLab Inc., Tokyo, Japan) using nonlinear least-square technique. An exponential function was fitted to give the relative predictive power of N mineralization as a function of root length density with an equation of the form y = abx where  $a \neq 0$ .

# 6.3 Results

# 6.3.1 Runoff and sediment yield

Mean seasonal soil loss and runoff differed significantly among the treatments and were highest in sole potato, intermediate in intercropping and lowest in sole legumes (Fig 6.1).



Figure 6.1: Mean seasonal soil loss and runoff measured under different cropping systems in the upper midland (UM), lower highland (LH) and upper highland agro-ecological zones. Means with different letters show significant differences between treatments at Tukey's  $p \le 0.05$ . Data presented as averages of 4 replicates.

Compared to sole potato, cumulative soil loss in the UH AEZ reduced by 25–43% in potato-lima bean intercropping. Similarly, soil loss in the upper midland and lower highland AEZs reduced by 16–47% in potato-dolichos intercropping. Cumulative runoff showed a similar trend reducing by 22 to 43% respectively in potato-legume intercropping compared with the sole potato.

#### 6.3.2 Residue yield and chemical composition

Biomass yield differed significantly among the cropping systems and was greatest in pure legumes (range of 4.4 to 7.5 t DM ha<sup>-1</sup>) followed by intercropping (3.4 to 6.6 t DM ha<sup>-1</sup>) and lowest in sole potato cropping (1.8 to 3.9 t DM ha<sup>-1</sup>) (Table 6.2).

Table 6.2: Dry matter yield, nitrogen concentrations and chemical composition of crop residues at time of field incorporation. Values expressed as averages over four seasons.

Agro- ecological zone	Cropping system	DM (t ha <sup>-1</sup> )	OC (%)	Total N (%)	Lignin (L) (%)	Polyphenol (P) (%)	C:N	L:N	P:N	L+P : N
Upper	Potato + dolichos	5.3d	37.8b	2.0b	5.4bc	1.5a	18.9a	2.7b	0.8ab	1.7b
midland	Potato + lima bean	3.4b	35.8b	1.8b	5.8bc	2.0a	19.9a	3.2b	1.1b	2.4c
	Sole dolichos	7.5e	51.8c	3.2c	4.1a	1.4a	16.2a	1.2a	0.4a	0.5a
	Sole lima bean	4.9c	47.3c	2.8c	5.1ab	1.8a	16.9a	1.9a	0.7a	0.9a
	Sole potato	1.8a	27.5a	1.1a	7.1c	2.2a	25.0b	6.4c	2.1c	7.8d
Lower	Potato + dolichos	6.6d	49.6bc	2.6b	4.8b	1.6a	19.1a	1.9ab	0.6a	1.0b
highland	Potato + lima bean	4.8b	45.3b	2.3b	5.4c	1.9a	19.7a	2.4b	0.8a	1.4b
	Sole dolichos	8.9e	55.7d	3.3c	3.5a	1.5a	16.9a	1.1a	0.5a	0.5a
	Sole lima bean	6.3c	53.1cd	3.0c	4.7b	1.8a	17.7a	1.6a	0.6a	0.7a
	Sole potato	2.9a	34.4a	1.3a	6.7d	2.1a	26.5b	5.2c	1.6b	5.2c
Upper	Potato + dolichos	5.8b	44.2b	2.2bc	5.0b	1.6a	20.1a	2.3b	0.8a	1.4b
highland	Potato + lima bean	6.6c	48.0c	2.5cd	5.0b	1.8a	19.2a	2.0b	0.7a	1.1b
	Sole dolichos	4.4a	52.9d	2.8d	4.0a	1.5a	18.9a	1.4a	0.6a	0.7a
	Sole lima bean	8.1d	55.1d	3.3e	4.8b	1.6a	16.7a	1.5a	0.5a	0.6a
	Sole potato	3.9a	38.5a	1.4a	6.5c	1.9a	27.5b	4.7c	1.4b	4.3c

Means followed by the same letter down the column within an agro-ecological zone do not differ significantly by Tukey's HSD at  $p \le 0.05$ .

Residue total organic carbon content varied from 27.5–38.5% in pure potato stands to 47.3–55.7% in pure legume stands. Intercropping showed intermediate range of residue OC composition ranging between 35.8–49.6% across the AEZs. The highest concentration of total N was observed in pure legumes (2.8–3.3%) followed by intercropping (1.8–2.6%) and lowest in pure potato (1.1–

1.4%). Lignin contents were significantly higher in pure potato (6.5-7.1%) relative to intercropping (4.1-5%) and lowest in pure legumes (3.5-5.1%). Similar trends were observed for the polyphenol contents that ranged between 1.4–1.8% in pure legumes, 1.5–2.0% in intercropping and 1.9–2.2% in pure potato. Compared to sole potato cropping, intercropping potato with legumes lowered the C-to-N ratio by 20–28%, lignin-to-N ratio by 50–63%, polyphenol-to-N ratio by 47–63% and lignin + polyphenol-to-N ratio by 69–80%.

# 6.3.3 Nitrogen mineralization and uptake

Net N mineralized ranged between  $6.9-9.0 \text{ kg N ha}^{-1}$  in sole potato,  $39.3-56.0 \text{ kg N ha}^{-1}$  in intercropping and  $39.9-64.9 \text{ kg N ha}^{-1}$  in pure legumes (Fig. 6.2a). Peak N release in intercropping system was  $5-9 \text{ kg N ha}^{-1}$  within 8-10 weeks of residue decomposition and matched with peak N uptake by potato ( $19.9-31.2 \text{ kg N ha}^{-1}$ ) that occurred at this time. This is in contrast to pure potato which obtained peak net mineralization of  $2.6-3.6 \text{ kg N ha}^{-1}$  at 12-14 weeks of residue decomposition. Nitrogen uptake at this period was relatively low ranging between  $4.3-25.4 \text{ kg N ha}^{-1}$ . Sole legumes exhibited peak mineralization at 6-8 weeks after residue incorporation. Nitrogen uptake by potato, though high at this period, was not at its peak. Cumulatively, intercropped potato showed significantly higher N uptake per plant compared with the sole potato and was lowest in the sole legumes irrespective of AEZs (Fig. 6.2b). This N uptake ranged from  $87.5 \text{ to } 171.2 \text{ kg N ha}^{-1}$  in intercropping and  $53.1-79.5 \text{ kg N ha}^{-1}$  in sole legumes. In the upper highland AEZ, N uptake by dolichos grown either in pure stand or in intercropping was significantly lower than that of lima bean.



Figure 6.2: Accumulations of residue N mineralized (a) and N taken up by the crops (b) after 16 weeks of residue decomposition and potato planting respectively in the upper midland, lower highland and upper highland agro-ecological zones. Soil control treatments without residue amendments were included during decomposition for comparisons and for computation of net N mineralization.

# 6.3.4 Soil nitrogen balance

Soil N depletion occurred in pure potato stands (-38 to -24.5 kg N ha<sup>-1</sup>) while surplus occurred in monocultures of dolichos (5.5– 58.2 kg N ha<sup>-1</sup>) and lima bean (19.7–55.5 kg N ha<sup>-1</sup>) (Table 6.3). Intercropping showed intermediate soil N balance (3.1–9.5 kg N ha<sup>-1</sup>). The main component of soil N input was mineralization followed by residual N and N<sub>2</sub> fixation across the cropping systems and AEZs. Table 6.3: Nitrogen balance and its components under different cropping systems in the upper midland (UM), lower highland (LH) and upper highland (UH) agro-ecological zones.

			Nitrogen outputs									
	Cropping system	Fertilizer	Mineralization	N <sub>2</sub> fixation	Residual N	Initial N	Crop N uptake	Leaching	Runoff	Erosion	Volatilization	ΔΝ
							Kg N ha <sup>-1</sup>					
	Pure potato	90.0c	9.3a	-	1.6a	1.0a	94.3b	6.6c	1.4d	2.2c	34.4a	-37.0a
(UM)	Pure lima bean	0.0a	34.7b	2.7a	21.5b	1.0a	36.1a	2.4b	0.5a	0.3a	0.9d	19.7c
	Pure dolichos	0.0a	39.2b	12.3c	24.5b	0.8a	40.9a	1.7ab	0.2a	0.1a	0.7d	33.2d
	Potato + lima bean	52.0b	48.7c	6.8b	19.8b	0.9a	92.4b	1.3a	1.3cd	1.2b	28.8b	3.1b
	Potato + dolichos	52.0b	51.6c	21.9d	23.7b	1.1a	116.9c	1.2a	1.0bc	0.9ab	21.7c	8.6b
	Pure potato	90.0c	25.7a	-	6.9a	1.1a	106.3b	28.8b	3.8c	2.2c	20.6a	-38.0a
(LH)	Pure lima bean	0.0a	62.0c	4.2a	29.1bc	1.1a	38.6a	6.6a	0.2a	0.3a	0.4c	50.4c
	Pure dolichos	0.0a	73.7d	14.9c	33.2c	0.9a	58.4a	5.2a	0.2a	0.2a	0.4c	58.2c
	Potato + lima bean	45.0b	55.4bc	8.9b	24.3b	0.8a	115.7bc	4.4a	1.6b	2.0c	7.7b	3.0b
	Potato + dolichos	45.0b	67.4cd	16.5c	28.2bc	0.9a	135.8c	3.9a	0.6ab	1.3b	5.7b	10.7b
	Pure potato	100.0c	26.7a	-	16.5b	2.8a	116.9d	36.9	4.0b	3.8c	8.8a	-24.5a
(UH)	Pure lima bean	0.0a	80.1c	5.2c	25.4c	2.6a	46.0b	10.9a	0.5a	0.3a	0.1d	55.5c
	Pure dolichos	0.0a	43.3b	1.8a	8.8a	2.5a	24.8a	24.3	1.0a	0.3a	0.5d	5.5b
	Potato + lima bean	50.0b	82.1c	7.9d	18.7b	2.1a	134.1e	11.7a	1.6a	1.7b	2.2c	9.5b
	Potato + dolichos	50.0b	45.1b	3.3b	9.9a	2.5a	71.0c	20.9b	4.0b	2.1b	5.5b	7.3b

 $\Delta$ -N indicates the soil nitrogen balance. Values followed by different letters within a column denote significant differences at Tukey's  $p \le 0.05$ . Values are 4 replicates expressed as averages over four seasons.

Leaching, erosion and runoff accounted for 14–40% of the N output in the upper highland and were highest in sole potato stands, intermediate in intercropping and lowest in sole legumes. Volatilization contributed 12–23% of N output in the upper midland AEZ and was highest in pure potato stands and lowest in sole legumes.

The amount of N leached decreased with increasing root length density  $(0.53 \le R^2 \le 0.90)$ (Fig. 6.3). The relationship was strongest in pure potato plots ( $R^2 = 0.90$ ) followed by potato + lima bean ( $R^2 = 0.65$ ), pure dolichos ( $R^2 = 0.63$ ) and lowest in sole lima bean ( $R^2 = 0.53$ ).



Figure 6.3: Amount of nitrogen leached below the active root zone as a function of root length density under different cropping systems. Each point is treatment average of 4 replicates across the sampling sites.

#### **6.4 Discussion**

#### 6.4.1 Influence of potato-legume intercropping on soil erosion

Legume intercropping increased leaf area index as legumes exhibited rapid growth which enhanced canopy overlap thus covering the empty spaces in the inter-rows of potato and legumes (see Fig 4.2 and 7.3). This reduced the raindrop hitting force and slowed down the velocity of runoff. While the low canopy of dolichos reduced the raindrop falling height and minimized the raindrop hitting force, the effective canopy overlap by lima bean enhanced the capability of leaves to resist bending thus augmenting stem flow. This increased the effective rain-receiving area which eliminated erosion and runoff generation. The canopy of dolichos was extended after potato harvest to the subsequent season and was thus effective in controlling post-harvest runoff and soil erosion. The canopy closure of potato was effective only after 40 to 45 days and diminished at physiological maturity when plants began to senesce. Jego et al. (2008) noted that the shorter period of growth characterizing potato intensifies soil erosion especially after the plants begin to senesce.

The low sediment yield and runoff generation under potato-dolichos intercropping in the upper midland and lower highland AEZs resonates with the high LAI observed in this treatment. This observation suggests a greater soil stabilizing effect of dolichos which may be asserted to the continuity of groundcover throughout the year compared with the other treatments which were left bare soon after potato harvest. In the upper highland AEZ, growth of dolichos was suppressed by the low ambient temperatures (Cook et al. 2005). In similar studies, reduction of sediment yield, runoff generation and associated N loss under potato-dolichos intercropping has been adduced to the canopy heterogeneity contributed by differences in plant heights (Nyawade et al. 2019). The first raindrops that hit the canopy is intercepted and dispersed by dolichos which exhibits greater height

relative to potato. Pure potato stands due to their low uniform canopy height, converge the smaller raindrops into bigger drops renewing their erosion potential.

#### 6.4.2 Residue N mineralization and N uptake

The N mineralization process was influenced by residue biochemical composition, the effect of which was greatest under sole legumes indicating that legumes enhanced supply of N to the soil. This observation was indicated by the ratios of residue C-to-N, lignin-to-N, polyphenol-to-N and lignin + polyphenol-to-N which were low in intercropping relative to pure potato stands. This observation agrees with previous findings which showed that legumes provide high amounts of Nrich residues thus enhancing N mineralization and supply (Palm and Sanchez, 1991; Tian et. 1992; Handayamoto et al. 1994).

Generally, N mineralization was lowest within the first 2 weeks of residue decomposition under sole potato residue but increased sharply at the sixth week. During this period (sixth week), the potato tissues had less than 10 g kg<sup>-1</sup> N with C-to-N ratio of approximately 30, indicating that this concentration is a reasonable critical level for initial net mineralization of potato residue. With legume intercropping, immobilization was observed within the first 2 weeks of residue incorporation probably due to the low build-up of microbial population which are involved in residue degradation (Ruijter et al. 2010).

The increase in residue N mineralization after the 4<sup>th</sup> week of residue decomposition irrespective of treatments could be due to release of soluble and readily decomposable N-containing biochemical compounds like amino sugars, nucleic acids and proteins (Paul and Clark, 1989). The peak N release at 6–10 weeks across the treatments after residue incorporation was consistent with the residue mass loss and signified intense organic matter degradation at these times. At later stages

of decomposition, little extra N was released, an observation we attribute to the decrease in residue mass which was the source of microbial substrates.

The differences in N mineralization patterns between the sites was attributed to seasonal soil moisture and temperature variations. Moisture and temperature control decomposer activity and thus influence plant residue decomposition (Tian et al. 1993; Mitchell et al. 2000). The lower temperatures in the upper highland AEZ could have lowered the respiration rates thus delaying net mineralization (Tian et al. 1997). The low mineral N during the wet seasons in this study could therefore be attributed to the generally low soil temperatures which lowered the decomposition rate of crop residues.

The occurrence of peak residue N mineralization and N uptake in intercropping indicated synchrony of N supply and demand. At week 4 after potato planting, pure legumes showed higher cumulative N uptake due to their ability to grow fast and take up the N mineralized. A comparatively slow mineralization from sole potato residues resulted in low initial N supply as greater amount of N was immobilized in potato residue at this time. The increase in N uptake by intercropped potato could have also been due to root facilitative interactions such as rhizodeposition and N-transfer (Jensen, 1996). Whitbread et al. (2011) showed the ability of dolichos to capture N from the subsoil and pump it to the surface soil strata thus making it available for the shallow rooted crops.

#### 6.4.3 Soil N balance and its components

A significantly higher N balance was recorded in intercropping relative to sole potato reflecting the effect of legumes on N dynamics. Intercropping increased groundcover across the seasons and agro-ecological zones which greatly conferred shading to the soil. This lowered the soil temperature and reduced ammonia volatilization while minimizing N losses to soil erosion and runoff. Nitrogen being highly mobile in its nitrate form was quickly leached below the potato-root zone whereas the legume intercrops developed deep root systems that enhanced the recovery of the N leached from the topsoil.

The greater N fixation by intercropped legumes was perhaps in response to increased soil N competition by potato grown in association. This, according to previous studies permits legumes' greater reliance on symbiotic N<sub>2</sub>-fixation (Inal et al. 2007). Alonso-Ayuso et al. (2014) found that high competition with barley for soil N forced vetch to rely on  $N_2$  fixation for its N requirement, and in return accumulated large amount of N compared to sole vetch. Therefore, in conditions that enabled the growth of both companion species, the increased uptake of N by potato caused a temporary decrease in soil N thereby increasing the proportion of N<sub>2</sub> fixed by the legume. Hauggaard-Nielsen et al. (2009) argued that the high N<sub>2</sub> fixation potential of legumes grown in intercropping is due to the high degree of complementarities arising from species interaction that enables natural regulation mechanisms between intercrop components. Such mechanisms include improved N capture in mixed crops as a result of complementarity in foraging strategies in space (soil profile) and time (growth period of the crop) (Lithourgidis et al. 2011). Snoeck et al. (2000) showed that up to 30% of N<sub>2</sub> fixed by legumes was transferred to the associated non-legume under field conditions. They attributed this to the interactions of roots and to the secretion of root exudates. Similarly, the legumes in this study had deeper rooting depth which enabled them to explore greater soil volume and diverge the N leached beyond the potato root zone.

The generally higher  $N_2$  fixation by dolichos over lima bean could be ascribed to the genetic differences among the crop cultivars as interactions between legume microsymbionts are highly specific (Qiang et al. 2003). In addition, the continued accumulation of biomass by dolichos days after lima bean senesced may explain the differences in relative contribution of fixed  $N_2$  between the two legumes. Cook et al. (2005) noted that dolichos is able to take up N from soil layers up to

1.5–2.4 m depth. Studies conducted in central Kenya indicated that the N contribution of dolichos is in the range of 8 to 25 kg N ha<sup>-1</sup> while that of lima bean is in the range of 1.6 to 7 kg N ha<sup>-1</sup> (Chemeningwa et al. 2007).

The greater N uptake by dolichos than lima bean in the lower highland and upper midland agro-ecological zones may be related to the enhanced  $N_2$ -fixation. This mechanism directly affects N uptake and metabolism (Wyeh and Rains, 1978). The converse was observed in the upper highland AEZ due to the low temperatures that dolichos could not tolerate. Chemeningwa et al. (2007) recorded higher nodule numbers per plant for dolichos than for lima bean under elevated temperature conditions. This was attributed to dolichos' high adaptation to water stress. The notably lower crop nitrogen uptake in the upper midland AEZ relates to the low soil moisture and high ambient temperatures.

Nitrogen volatilization was lowest in sole legumes because these treatments were not fertilized with N, and thus lacked the N emission triggered by N fertilization. The differences in cumulative ammonia volatilization between the study sites were attributed to the differences in temperature and rainfall. The lower temperatures coupled with higher rainfall amounts in the upper highland AEZ may have decreased ammonia volatilization (Whitehead et al. 1988) while the high ambient temperatures and low soil moisture in the upper midland AEZ favored net volatilization.

# 6.5 Conclusion

The results from this study give an insight on the potential role of potato-legume intercropping in enhancing N inputs into the soil while minimizing the output pathways, thus offering a mechanism for optimizing soil nitrogen content in smallholder farming systems. Reduction of nitrogen loss in this manner could be an effective measure to improve soil productivity and increase the efficiency of nutrient use while dampening nitrogen losses to water bodies. Even though the observed biological  $N_2$  fixation contribution by the legumes in this study seemed insufficient to cover the total potato N need, this amount of nutrient remains important to the resource-constrained farmers. These contributions will better be understood if linkages between litter fauna and residue biochemical composition is assessed and established under different temperature and moisture regimes.

#### **CHAPTER SEVEN**

# INTERCROPPING OPTIMIZES SOIL TEMPERATURE AND INCREASES CROP WATER PRODUCTIVITY AND RADIATION USE EFFICIENCY OF POTATO

# Abstract

Integrating crop species with different photosynthetic pathways has a great potential to increase efficiency in the use of scarce resources. In order to tap the resource complementarity emanating from this mix, this study intercropped potato (Solanum tuberosum L.) with lima bean (Phaseolus lunatas L.) and dolichos (Lablab purpureus L), and related soil temperature with radiation use efficiency and crop water productivity of rainfed potato in the upper midland (1552 meters above sea level (masl), lower highland (1894 masl) and upper highland (2552 masl)) agro-ecological zones of Kenya. The study was laid out in a randomized complete block design with four replications. Soil samples (0–120 cm depths) were taken from each plot at key stages of potato growth. Leaf area index (LAI), light interception, soil temperature and soil water contents (SWC) were quantified at different stages of potato growth and related with the radiation use efficiency (RUE) and crop water productivity (CWP). Intercropping increased LAI by 26–57% relative to sole potato stands and significantly lowered the soil temperatures in the 0–30 cm depth by up to 7°C. This caused an increase in SWC by up to 38%, thus increasing RUE by 56–78% and CWP by 45–67%. Land equivalent ratios of the intercrops were generally greater than unity indicating more efficient and productive use of land resource. Intercropping potato with legumes is coupled with optimum rootzone soil temperature and soil water content, thus potentially exerting additive relations in radiation interception and subsequent conversion into crop biomass.

**Keywords:** Intercropping; radiation interception; radiation use efficiency; soil water content; soil temperature.

111

# 7.1. Introduction

Sustainable potato production requires careful optimization of the use of resources to improve soil fertility and crop productivity. This does not preclude the use of solar radiation and soil moisture resources. Biomass productivity of potato crops grown under optimum growth conditions has accordingly been described by the amount of soil water utilized per unit biomass production and the efficiency by which solar radiation is converted into plant biomass (Monteith, 1965, 1977). These processes are partly determined by the leaf area index (LAI), and an index of radiation interception referred to as the extinction coefficient (Lizaso et al. 2003). As LAI increases, more radiation is intercepted per unit ground area resulting in higher assimilation rates (Ewert, 2004).

The efficiency by which the intercepted radiation is converted into dry matter by potato crop has great relation with the soil temperature and soil moisture contents. Under water deficit conditions, potatoes curl their leaves to lower transpiration rates (Struik, 1989). This mechanism reduces the radiation interception and in turn negatively impacts plant water uptake. Optimal foliage growth and therefore light interception of potato occurs at soil temperature range of 15–20°C (Rykaczewska, 2015). Elevated soil temperatures within the potato rhizosphere induce moisture stress that increases total dry matter allocation to roots and stems at expense of tubers thus reducing crop water productivity (Thornton et al. 1996). Raising night temperatures over the range of 0–20°C increases potato root length while temperature increase to above 25°C induces sharp reduction in the number and weight of tubers (Wilkinson and Davies, 2002).

The adverse effects of elevated soil temperatures on potato growth can be optimized by cropping systems capable of enhancing persistence of leaf area coverage. Intercropping is one of such systems well recommended due to its multiple benefits (Muthoni et al. 2013; Nyawade et al. 2018a; Gitari et al. 2018a, 2018b, 2019). Intercropping achieves greater complementarity of light

use per unit area of land with crop mixtures having different rooting ability, canopy structure, height, and nutrient requirements (Nyawade et al. 2019). This complementarity in the use of resources depends on microclimate modification created by the two crops grown in companion.

Nevertheless, little information is available on crop water productivity and radiation use efficiency of potato in relation to soil thermal regimes generated by legume intercrops. This information is needed for identification of management practices that may optimize the high soil temperature conditions prevalent in the tropical and sub-tropical potato growing areas. Therefore, the present work was designed to test the hypothesis that potato yield, crop water productivity and radiation use efficiency of potato grown in intercropping systems exhibit positive relationship with the soil temperature and soil moisture contents.

#### 7.2 Materials and methods

#### 7.2.1 Experimental site and trial design

The trial was carried out during the rainy seasons of 2017 and 2018 in three agro-ecological zones of Kenya; upper midland-Kirinyaga (1552 m above sea level (masl)), lower highland-Kabete (1894 masl) and upper highland-Nyanadarua (2552 masl)). Nyandarua receives mean annual temperature of 18.2°C with an annual rainfall amount of 1500 mm. Kabete receives average temperature of 21.2°C and annual rainfall of 1000 mm. Kirinyaga exhibits relatively lower annual rainfall amount ranging between 600 and 1000 mm and mean annual temperature of 24.4°C. The low rainfall couple with high ambient temperatures are the major hindrances to potato production in the UM and LH AEZs. The initial soil physical and soil chemical characteristics are presented in Chapter 4 (Table 4.1).

A heat and water stress tolerant potato (*Solanum tuberosum* L.) cultivar, Unica (CIP 392797.22) (CIP, 2018), was grown in sole stands and in intercropping with either lima bean

113

(*Phaseolus lunatus* L.) or dolichos (*Lablab purpureus* L.) and the respective single crop of the legumes. The treatments were laid out in a randomized complete block design with four replications. Intercropping arrangement constituted 2 rows of potato alternating with 2 rows of legumes. Presprouted tubers were planted at a uniform depth of 10 cm on pre-hilled ridges spaced at 75 cm with an inter-seed spacing of 30 cm. Two legume bean seeds were planted per hole at within row space of 20 cm and inter-row space of 75 cm between potato and legume strips and 50 cm between two legume strips. The crop management and cultural practices are described in section 3.5.

#### 7.2.2 Determination of soil water content and soil temperature

Soil water content (SWC) of each plot was measured at 0–30 cm, 30–60, 60–90 and 90–120 cm depth intervals using tensiometers (0–100 kPa) installed in the inter-row of potato and legumes as described in chapter 4, section 4.2.5. Prior to their installation, the tensiometers were calibrated using gravimetric soil moisture measurements taken at soil depth intervals similar to that of tensiometers.

The soil temperature was measured using temperature sensor probes (Onset HOBO USeries, UX120-006 M). The probes were installed at the depth of 0–30 cm in each experimental plot. Soil temperature measurements were taken between sowing and harvesting and recorded by automatic data-logging equipment at every 1 h common step.

# 7.2.3 Determination of leaf area index, light interception, crop yield, light extinction and radiation use efficiency

Radiation interception of photosynthetically active radiation (PAR) and leaf area index (LAI) were measured from 14 days after potato emergence and progressively at 14 days interval until physiological maturity using a Sunfleck Ceptometer-LP-80 (Decagon Devices, Pullman, WA, USA). Measurements were taken only under the sky-blue conditions with no or minimum clouds,

between 1130 and 0130 h (local time), and during a period of constant incident solar radiation. This was meant to eliminate the effect of solar elevation on light interception. Data collection interval thus deviated by about 1–3 days depending on the sky conditions. Dead and brown plant materials were removed from the experimental plots before each radiation measurement. For each measurement, nine above and below canopy readings were taken perpendicularly to the crop rows to ensure that more leaf area was exposed to the light sensors. Corresponding LAI values were read upon averaging the above and below canopy readings. Plot values were computed from the average of four successive middle row readings. The PAR intercepted was calculated using Eq. (7.1–7.3) (Koocheki et al. 2016).

$$PAR_{intercrop} = PAR_o \left[ 1 - \exp\left( \left( -\lambda_{potato} * LAI_{potato} \right) + \left( -\lambda_{legume} * LAI_{legume} \right) \right) \right]$$
(7.1)

$$PAR_{potato} = \frac{\lambda_{potato} * LAI_{potato}}{(\lambda_{potato} * LAI_{potato}) + (\lambda_{legume} * LAI_{legume})}$$
(7.2)

$$PAR_{legume} = PAR_{intercrop} - PAR_{potato}$$
(7.3)

Where PAR is the photosynthetically active radiation (400–700 nm); LAI, the leaf area index,  $\lambda$ , light extinction coefficient; PAR<sub>o</sub>, photosynthetically active radiation incident which equals to half the daily global radiation (Monteith and Unsworth, 1990). Daily global radiation was estimated from the daily sunny hours recorded from the adjacent meteorological station. The light extinction coefficient ( $\lambda$ ) was determined from the slope of the linear regression between the natural logarithm of radiation transmission and leaf area index (Monteith, 1965).

The tuber and legume yields were estimated from the central 1.2 m<sup>2</sup> area of each plot. Tubers were dug out using fork hoe at 85–95 days after planting when the stems were completely dry, brushed and fresh weight taken. About 500 g of the samples from each plot were sliced and dried in an oven at 65°C for 72 h and reweighed to determine tuber DW. Harvesting for lima bean and dolichos was done at 100 and 120 days respectively. The shoot biomass estimations was done by

cutting the plants at the soil surface using machetes. The dry mass was determined by oven-drying about 500 g samples at 65°C to a constant mass. The yields (tubers and legumes) were converted into potato equivalents (PEY) using Eq. (7.4). For dolichos, the estimations considered grain and shoot biomass separately for this legume is used both as pulse and forage.

$$PEY(kg ha^{-1}) = PY(kg ha^{-1}) + \frac{LY(kg ha^{-1})*LP(US$kg^{-1})}{PP(US$kg^{-1})}$$
(7.4)

Where PEY = potato equivalent yield, PY = potato yield, LY = legume yield, PP = market price of potato (0.38 US\$ kg<sup>-1</sup>) and LP = market price of legumes (0.21, 0.05 and 1.15 US\$ kg<sup>-1</sup> for lima bean grain, dolichos forage and dolichos grain respectively).

Radiation use efficiency (RUE) (g MJ<sup>-1</sup>) of each cropping system was estimated by fitting a linear regression (least square) to the cumulative amount of radiation absorption (MJ m<sup>-2</sup>) and dry matter accumulation from successive harvests (g m<sup>-2</sup>) (Monteith, 1994). The slope of each regression was taken as the RUE for each treatment. Data taken at crop physiological maturity were excluded from RUE calculations due to its negligible role in biomass accumulation (Black and Ong, 2000).

#### 7.2.4 Estimation of crop water productivity

Crop water productivity (CWP) was computed from the soil water balance equation and potato equivalent yield (Allen et al. 1998, Eq. (7.5)), as:

$$CWP = \frac{PEY}{P+CR+\Delta SW+I-E-RO}$$
(7.5)

Where PEY = potato equivalent yield obtained; P = precipitation; CR = capillary rise of water;  $\Delta$ SW = change in soil water storage in root zone between planting and harvesting period, I = irrigation; E = surface evaporation; R = runoff. Capillary rise was assumed negligible because the groundwater table was more than 25 m below the soil surface (Karuku et al. 2014). Micro-lysimeters installed in each plot to a depth of 30 cm below the soil surface were used to estimate the soil surface evaporation. Deep percolation was estimated by tensiometers installed to 150 cm soil depth.

# 7.2.5 Assessment of intercrop productivity

The Land equivalent ratio (LER) was used to evaluate the productivity of intercrops over sole cropping (Eq. (7.6) (Mead and Willey, 1980).

$$LER = \frac{Y_{il}}{Y_{sl}} + \frac{Y_{ip}}{Y_{sp}}$$
(7.6)

Where Yil and Yip are intercrop yields of legume and potato respectively while Ysl and Ysp are the sole yields of legume and potato. The LER values above unity indicate that intercropping is more productive and efficient in using land resource than sole cropping, and values less than unity that sole crops are more productive.

#### 7.2.6 Statistical analyses

The data was analyzed using R software, version 3.5.2 (R Core Team, 2017). The treatment effects on soil water content, soil temperature, crop water productivity and radiation use efficiency were tested using a mixed model analysis of variance (ANOVA) with cropping system, season and agro-ecological zone considered as fixed factors and block as random factor. Whenever the interaction of cropping system and season was found significant, data were analyzed in separate seasons. The homogeneity of variances was tested by Bartlett test and where the variances were not homogeneous, data were transformed by the function y = X1/2 or y = log(x). Tukey's honest significant difference test was applied for multiple mean comparisons between treatments, and tests with p < 0.05 were considered statistically significant.

# 7.3 Results

# 7.3.1 Climatic variables measured during the study period

Averaged across the seasons, the mean rainfall amounts received during the study period were higher in the 2017 long rains followed by 2018 long rains, 2018 short rains and were lowest in the 2017 long rains (Table 7.1). These rains were 30–60% lower than the long-term averages and occurred mainly at tuber emergence and vegetative growth regardless of the site.

Table 7.1: Mean rainfall amount, diurnal and night temperatures, solar radiation, saturation vapor deficit and evaporation for the period between potato planting and harvesting.

	2	2017 LR		2	2017 SR		2018 LR			2018 SR				
Parameter/Agro-ecology	UM	LH	UH	-	UM	LH	UH		UM	LH	UH	 UM	LH	UH
Rainfall amount (mm)	290	308	388	•	184	221	299		243	298	379	 190	237	321
Diurnal temperature (°C)	28.5	25.4	22.2		29.8	25.4	21.0		27.3	24.9	21.4	27.9	25.3	22.5
Night temperature (°C)	20.3	17.8	14.5		21.3	18.5	16.0		20.8	18.1	16.5	17.3	16.5	17.3
Solar radiation (MJ m <sup>-2</sup> )	21.2	19.8	18.8		21.5	19.3	19.8		20.6	20.4	18.1	21.8	19.1	19.3
Vapor deficit (mbar)	8.0	6.1	5.2		7.9	6.5	5.9		8.0	6.1	5.2	7.1	6.1	5.0
Soil evaporation (mm day-1)	11.2	9.7	4.5		12.5	9.6	4.8		10.9	9.2	4.1	13.6	10.0	4.8

LR, long rains, SR, short rains, UM, LH, UH, upper midland, lower highland, upper highland agro ecological zones respectively. Values expressed as averages over four seasons.

The diurnal and night temperatures were highest in the UM zone and lowest in the UH zones. Mean global solar radiation were greater in the UM zone compared to the LH and UH AEzs. Soil evaporation ranged between 4.5 mm day<sup>-1</sup> in the UH zone and 11.2 mm day<sup>-1</sup> in the UM, and saturation water deficit between 5.2 and 8.0 mbars across the sites.

#### 7.3.2 Leaf area index development

The leaf area index (LAI) differed significantly between treatments and was greatest in legume intercropping (1.24–3.78) relative to sole potato cropping (Fig. 7.1). Intercrop of potato and dolichos attained maximum LAI in the UM (1.25–2.68) and in the LH (1.45–3.13) irrespective of the season. This was reversed in the UH zone where intercrop of potato and lima bean recorded the greatest LAI (1.78–3.78). Leaf area index was significantly greater in the long rains (0.45–3.78) than

in the short rains (0.06–2.29) irrespective of the agro-ecological zone and peaked one week earlier in the UM zone compared to the UH zone.



Figure 7.1: Development of leaf area index by different treatments in upper midland (UM), lower highland (LH) and upper highland (UH) agro-ecological zones. Each point is treatment average of 4 replicates. Vertical bars indicate standard error of means. LR=long rains, SR= short rains.

# 7.3.3 Soil water content and soil temperature

There were significant differences (p < 0.05) in the mean volumetric soil water contents (SWCs) measured between the different cropping systems and growing seasons (Table 7.2). Compared with the sole potato stands, highest significant SWC in the surface soil layer was measured under potato-dolichos in the UM (0.07–0.19 cm cm<sup>-3</sup>) and in the LH AEZs (0.18–0.31 cm cm<sup>-3</sup>), but under potato-lima bean in the UH zone (0.24–0.36 cm cm<sup>-3</sup>). Generally, SWC was consistently highest in the UH AEZ (0.17–0.36 cm cm<sup>-3</sup>) and lowest in the UM AEZ (0.02–0.26 cm cm<sup>-3</sup>) regardless of the sampling stage. Mean soil temperature was significantly higher under sole potato (22.6–28.1°C) than under legume intercropping (18.1–23.7°C) irrespective of agro-ecological

zone and seasons. Seasons, agro-ecological zones, and cropping systems had significant effect on soil temperature and soil water content.

Table 7.2: Intercropping effect on soil moisture content and soil temperatures across the four seasons of study.

		2017	2017	2018	2018	2017	2017	2018	2018			
Agro-		LR	SR	LR	SR	LR	SR	LR	SR			
ecologica		Vol	lumetric w	ater conter	nt <sup>1</sup>		Soil tem	perature <sup>1</sup>				
l zone			(cm <sup>3</sup>	cm <sup>-3</sup> )			(°C)					
Upper	Sole Potato	0.13a	0.02a	0.10a	0.08a	27.1c	28.1d	26.2d	27.5c			
midland	Sole Dolichos	0.26d	0.10c	0.24e	0.22d	20.6a	20.9a	20.3a	21.4a			
	Sole Lima bean	0.21c	0.06b	0.21de	0.18c	22.9b	21.5b	21.8ab	21.6ab			
	Potato + Dolichos	0.19bc	0.07b	0.18cd	0.16bc	22.5b	22.4bc	22.2bc	21.9ab			
	Potato + Lima bean	0.17b	0.09bc	0.14b	0.15b	23.1b	23.6c	23.7c	22.3b			
Lower	Sole Potato	0.22a	0.14a	0.20a	0.15a	24.7c	25.8d	24.8d	25.4c			
highland	Sole Dolichos	0.37e	0.22d	0.32c	0.31d	17.1a	20.4a	17.3a	21.4a			
	Sole Lima bean	0.32d	0.16b	0.31c	0.28c	19.5b	21.2b	19.6b	21.3a			
	Potato + Dolichos	0.28c	0.18bc	0.31c	0.24b	19.5b	21.1b	20.4bc	21.8ab			
	Potato + Lima bean	0.25bc	0.19c	0.25b	0.26bc	20.7b	23.9c	21.8c	22.4b			
Upper	Sole Potato	0.26a	0.17a	0.19a	0.18a	22.6d	23.5d	21.7d	22.3e			
highland	Sole Dolichos	0.30b	0.27c	0.34d	0.24bc	15.3a	20.6c	19.3c	18.7bd			
	Sole Lima bean	0.38c	0.24b	0.27c	0.25c	17.8b	17.4a	17.8a	15.4a			
	Potato + Dolichos	0.31b	0.27c	0.27c	0.22b	17.9bc	19.8bc	17.9a	19.1cd			
	Potato + Lima bean	0.36c	0.24b	0.23b	0.26d	18.1c	18.3ab	18.9bc	17.5b			

Letters indicate comparisons for means between the cropping systems at  $p \le 0.05$  by Tukey's HSD test. SR, short rains season, LR, long rains season. <sup>1</sup>Significant interaction between cropping system, agro-ecological zone and season was found for volumetric soil moisture content and soil temperature at the 0.05 probability level. Values expressed as averages over four seasons.

# 7.3.4 Light interception and leaf extinction coefficient

Fractions of light intercepted were consistently greatest in intercropping of potato and dolichos (0.54–0.59) in the UM and in the LH zones (0.61–0.85), but in potato-lima bean (0.76–0.82) in the UH zone (Table 7.3). These values were consistently lowest in the sole potato stands and decreased as crop growth advanced to tuber bulking. Intercropping significantly lowered the leaf extinction coefficient of both legumes and potato with the values ranging between 0.35 and 0.65 across the sites. Cropping systems, seasons and agro-ecological zone had significant effect on fractions of light intercepted.

Agro-		2017	2017	2018	2018	2017 I P	2017	2018	2018			
ecological	Cropping system	LR	SR	LR	SR	2017 LK	SR	LR	SR			
zone		Fract	ion of lig	ht interco	epted <sup>1</sup>	Lea	Leaf extinction coefficient					
Upper	Pure potato	0.29a	0.22a	0.27a	0.25a	0.61c	0.65c	0.62d	0.63d			
midland	Pure lima bean	0.44b	0.47b	0.48b	0.48b	0.38b	0.39b	0.41bc	0.37bc			
	Pure dolichos	0.58c	0.56c	0.59c	0.54c	0.39ь	0.36a	0.42c	0.34a			
	Potato + lima bean	0.59c	0.61c	0.65d	0.53c	0.37a	0.38ab	0.41bc	0.36a			
	Potato + dolichos	0.73d	0.71d	0.75e	0.69d	0.36a	0.37ab	0.38a	0.38c			
Lower	Pure potato	0.35a	0.42a	0.44a	0.30a	0.54c	0.5c	0.57c	0.57c			
highland	Pure lima bean	0.65b	0.71b	0.76b	0.61b	0.42b	0.46b	0.41b	0.44b			
	Pure dolichos	0.70c	0.79bc	0.74b	0.64b	0.35a	0.37a	0.42b	0.37a			
	Potato + lima bean	0.71c	0.82c	0.85c	0.61b	0.38ab	0.38a	0.42b	0.38a			
	Potato + dolichos	0.82d	0.87d	0.88c	0.78c	0.36a	0.37a	0.37a	0.38a			
Upper	Pure potato	0.43a	0.45a	0.54c	0.38a	0.51c	0.54c	0.53c	0.54d			
highland	Pure lima bean	0.72c	0.71c	0.79d	0.68c	0.36a	0.37ab	0.38a	0.39bc			
C	Pure dolichos	0.45a	0.41a	0.46b	0.41a	0.37a	0.36a	0.37a	0.38a			
	Potato + lima bean	0.81d	0.82d	0.76d	0.79d	0.39a	0.35a	0.38a	0.37a			
	Potato + dolichos	0.54b	0.52b	0.50b	0.50ь	0.45b	0.40b	0.44b	0.41c			

Table 7.3: Intercropping effect on light interception and leaf orientation.

Letters indicate comparisons for means between the cropping systems at  $p \le 0.05$  by Tukey's HSD test. SR, short rains season, LR, long rains season. <sup>1</sup>Significant interaction between cropping system, agro-ecological zone and season was found for fraction of light intercepted at the p<0.05. Values expressed as averages over four seasons.

# 7.3.5 Intercropping effect on potato yield equivalent

Highest potato equivalent yield were recorded by intercropping  $(7.2-38.2 \text{ t ha}^{-1})$  relative to sole potato  $(1.3-16.3 \text{ t ha}^{-1})$  and were significantly greater in the long rains  $(5.4-38.2 \text{ t ha}^{-1})$  than in the short rains season  $(1.3-27.5 \text{ t ha}^{-1})$  (Table 7.4). Potato-dolichos intercropping attained significantly lower potato equivalent yield  $(12.8-21.7 \text{ t ha}^{-1})$  compared with potato-lima bean  $(19.4-33.8 \text{ t ha}^{-1})$  in the UH zone. Potato equivalent yield responded significantly to the effect of cropping system, seasons and agro-ecological zones.

		2017 LR	2017 SR	2018 LR	2018 SR
Agro-ecological zone	Cropping system		Potato equivale	ent yield (t ha <sup>-1</sup> ) <sup>1</sup>	1
Upper midland	Pure potato	8.9a	1.3a	5.4a	3.2a
	Pure lima bean	12.3b	4.7b	9.9b	6.8b
	Pure dolichos	19.7c	7.3c	12.7bc	9.9c
	Potato + dolichos	25.1d	9.5d	16.9d	18.9d
	Potato + lima bean	17.3c	7.2c	13.2cd	9.8c
Lower highland	Pure potato	13.9a	2.9a	7.4a	6.5a
-	Pure lima bean	18.9b	5.8b	17.1b	9.8b
	Pure dolichos	25.2c	9.6c	22.6c	13.8c
	Potato + dolichos	38.2d	14.5d	34.6d	24.7d
	Potato + lima bean	24.9c	9.4c	23.8c	13.5c
Upper highland	Pure potato	16.3b	7.9a	9.4a	6.5a
	Pure lima bean	22.2c	19.6c	22.6cd	19.8c
	Pure dolichos	13.1ab	15.8b	11.1a	14.8b
	Potato + lima bean	33.8d	19.4c	26.8d	27.5d
	Potato + dolichos	12.8a	14.5b	19.6b	21.7c

Table 7.4: Potato yield as affected by intercropping across the four study seasons.

Letters indicate comparisons for means between the cropping systems at  $p \le 0.05$  by Tukey's HSD test. SR, short rains season, LR, long rains season. Values expressed as averages over four seasons. <sup>1</sup>Significant interaction between cropping system, agro-ecological zone and season was found for the potato equivalent yield at  $p \le 0.05$ .

#### 7.3.6 Intercropping effect on crop water productivity and radiation use efficiency of potato

Intercropping potato with legumes showed significantly higher crop water productivity ranging between 4.04 and 9.67 kg ha<sup>-1</sup> m<sup>-3</sup> relative to sole potato (1.02–3.23 kg ha<sup>-1</sup> m<sup>-3</sup>) (Table 7.5). Crop water productivity was significantly greater in potato-dolichos intercropping than in potatolima bean intercropping in the UM and LH AEZs (4.04–9.67 kg ha<sup>-1</sup> m<sup>-3</sup>). This trend was reversed in the UH AEZ where potato-lima bean exhibited highest CWP (3.21–5.98 kg ha<sup>-1</sup> m<sup>-3</sup>). Radiation use efficiency (RUE) of potato varied significantly among the treatment and ranged between 0.18 and 0.98 g MJ PAR<sup>-1</sup> in sole potato and 1.23–2.89 MJ PAR<sup>-1</sup> in potato-legume intercropping.

		2017	2017	2018	2018	2017	2017	2018	2018		
Agro-		LR	SR	LR	SR	LR	SR	LR	SR		
ecological		Crop wa	ter produ	ctivity <sup>1</sup>		Radiation use efficiency <sup>1</sup>					
zone	Cropping system		kg h	a <sup>-1</sup> m <sup>-3</sup>		MJ PAR <sup>-1</sup>					
Upper	Sole Potato	1.13a	1.02a	1.57a	1.32a	0.77a	0.18a	0.28a	0.30a		
midland	Sole Dolichos	4.58b	3.21c	3.04b	4.23c	1.89b	1.09b	1.72b	1.34c		
	Sole Lima bean	4.34b	2.02b	3.11b	2.13b	1.78bc	1.21b	1.56b	1.05b		
	Potato + Dolichos	6.23c	4.04d	6.96d	5.88d	2.09d	1.38c	2.12c	2.01d		
	Potato + Lima bean	5.99d	4.02d	4.02bc	4.01c	1.98cd	1.23b	2.08c	1.32c		
Lower	Sole Potato	2.43a	2.12a	2.45a	2.25a	0.87a	0.98a	0.59a	0.65a		
highland	Sole Dolichos	4.63b	4.34c	4.25b	5.33c	1.99b	1.28b	1.84b	1.67c		
-	Sole Lima bean	5.31c	3.21b	5.19c	3.23b	1.89b	1.27b	1.87b	1.35b		
	Potato + Dolichos	5.84c	5.53c	9.67d	6.81d	2.89d	1.43d	2.87d	2.31d		
	Potato + Lima bean	4.53b	5.22c	5.16c	5.12c	2.04c	1.33c	2.13c	1.56bc		
Upper	Sole Potato	3.23a	2.21a	3.02a	2.87a	0.98a	0.36a	0.54a	0.53a		
Highland	Sole Dolichos	3.78a	4.21b	3.23a	5.98c	2.03b	1.76cd	1.98c	1.98c		
	Sole Lima bean	5.98b	3.21b	5.12b	4.56bc	1.94b	1.64c	1.84c	1.97c		
	Potato + Dolichos	9.09d	6.54c	8.00d	4.55b	0.97a	1.96d	0.89b	0.98b		
	Potato + Lima bean	7.38c	9.20d	7.02c	9.02d	2.56c	1.48b	2.49d	2.36d		

Table 7.5: Crop water productivity and radiation use efficiency in response to intercropping.

Letters indicate comparisons for means between the cropping systems at  $p \le 0.05$  by Tukey's HSD test. SR, short rains season, LR, long rains season. Values expressed as averages over four seasons.<sup>1</sup> Significant interaction between cropping system, agro-ecological zone and season was found for crop water productivity and radiation use efficiency at the 0.05 probability level.

# 7.3.7 Intercropping effect on land productivity

Land equivalent ratios (LER) were above unity with intercropping in the UM and LH AEZs

(1.09–1.29) but lower than unity with sole potato (0.42–0.74) and sole legumes (0.91–0.98) (Fig.

7.2). The highest LER in this study (1.31) was obtained in potato-dolichos system in the UM AEZ.

Intercropping potato with dolichos obtained a LER value below unity in the UH AEZ (0.17), the

value of which was much lower than that of the sole potato (0.98).



Figure 7.2: Land equivalent ratios (LER) measured in the upper midland (UM), lower highland (LH) and upper highland (UH) agroecological zones. Data presented as pooled averages across the four study seasons. The LER values above unity indicate that intercropping is more productive and efficient in using land resource than sole cropping, and values less than unity implies that sole crops were more productive.

# 7.3.8 Relationships among potato yield, soil water content, soil temperature, crop water productivity and radiation use efficiency

Potato equivalent yield, crop water productivity and radiation use efficiency responded significantly to the effect of soil water content, soil temperature, night temperatures and to the interaction of soil water content and soil temperature (Table 7.6).

Dependent variable	Independent variables	Coefficient	Standard error	t Stat	р
Potato	Soil water content (SWC)	2.340	0.936	2.500	0.002***
equivalent	Soil temperature	-6.710	3.010	-2.230	0.000**
yield	Night temperature	-0.642	0.274	-2.346	0.031*
	SWC*Soil temperature	2.100	2.170	0.970	0.043***
	SWC*night temperature	0.420	0.190	2.180	0.019*
	Soil temperature*night temperature	-2.360	0.440	-5.310	0.760
Crop water	Soil water content (SWC)	1.360	7.570	0.180	0.000***
productivity	Soil temperature	-0.137	0.039	-3.513	0.009**
	Night temperature	-0.080	0.044	-1.818	0.023*
	SWC*Soil temperature	7.500	0.560	13.330	0.000 ***
	SWC*night temperature	0.160	0.060	2.820	0.065
	Soil temperature*night temperature	-0.110	9.300	-0.010	0.078
Radiation use	Soil water content (SWC)	4.130	2.360	1.750	0.006**
efficiency	Soil temperature	-0.137	0.039	-3.513	0.009**
	Night temperature	-1.710	0.300	-5.690	0.042*
	SWC*Soil temperature	-7.580	2.810	-2.690	0.007**
	SWC*night temperature	2.750	5.210	0.530	0.069
	Soil temperature*night temperature	-0.140	0.080	-1.630	0.010

Table 7.6: Response of potato yield, crop water productivity and radiation use efficiency to soil moisture and temperature.

Significance codes: 0 '\*\*\*' 0.001 '\*\*' 0.01 '\*' 0.05 '.' 0.1 '' 1; Adjusted R square-Potato equivalent yield = 0.83; Crop water productivity = 0.78; radiation use efficiency = 0.82.

The interaction of soil temperature and soil water content had the highest significant positive effect on both the potato equivalent yield, crop water productivity and radiation use efficiency. Soil water content showed stronger effect on all the three parameters (potato equivalent yield, crop water productivity and radiation use efficiency) compared to soil temperature and night temperatures.

# 7.4 Discussions

# 7.4.1 Variability of crop leaf area index

The greater leaf area index (LAI) under intercropping could be explained by the differences that existed in the vertical foliage arrangement and canopy architecture of the different crop varieties. As indicated by the significantly lower leaf extinction coefficient, the larger and nearly vertical leaves exhibited by dolichos provided complementarity to the slender leaves of potato. Ma et al. (2015) observed that potato canopy is characterized by leaf bending, creating bare surfaces between the crop rows which greatly hamper the LAI. Dolichos provided a low dense canopy which closed these bare spaces while the effective canopy overlap by lima bean bridged the interrow spacing thus enhancing LAI development (see Fig. 7.3). The canopy closure of potato was effective only after 40 to 45 days followed by rapid decline after physiological maturity. The legume intercrops conferred a complimentary canopy that kept LAI relatively high during these periods.



Figure 7.3: Canopy overlap by potato grown alone (a) and potato intercropped with lima bean (b) and dolichos at vegetative growth of potato. Photos taken in the upper midland agro-ecological zone at vegetative growth of potato.

Even though the higher soil temperature conditions early in the season favored tuber sprout and early canopy development in the UM and LH AEZs relative to the UH AEZ, the restricted peak leaf area index and limited persistence led to no advantage. While evaluating the effect of three temperature regimes (16, 22 and 27°C) on potato leaf formation, Shah et al. (2004) recorded leaf number values of 32, 20 and 14, respectively per mainstem indicating depressing effect of high soil temperatures. Potatoes grown under high soil temperature conditions often grow taller with longer internodes, reduced leaf numbers, and are characterized with leaves which are shorter and narrow (Struik et al. 1989). Elevated soil temperatures have further been shown to decrease
the number of potato leaves emerging from lateral branches (Nyawade et al. 2018a). All these affect the potential of LAI development and subsequently, radiation interception and use efficiency.

Attainment of peak LAI at different times in different sites suggests that the spatial distribution of LAI was affected by agro-ecological zones. This variability could be explained by changes in the timing of leaf senescence in response to water availability and ambient temperatures. Low nutrient availability under low soil moisture levels in the UM zone may have accelerated leaf senescence to satisfy the nutrient demand of sink organs leading to earlier attainment of peak LAI. Consistent results were found by Sinclair (2000) who argued that water deficit considerably reduces the stomatal conductance leading to an interruption of biomass accumulation. Under such conditions, water reserves of the plants may be consumed which can lead to termination of LAI development. Zahoor et al. (2010) demonstrated that high ambient temperatures might cause loss of cell wall plastid integrity. This coupled with stomatal closure in response to heat stress generally lowers the photosynthetic rates. Stomatal closure occurs due to decreased leaf turgor under low atmospheric vapor pressure along with root-generated chemical signals (Chaves et al. 2002). These processes might have shortened the crop leaf longevity leading to decrease in LAI. A better LAI development in the UH AEZ than in the LH AEZ could be because of the higher and well distributed rainfall amount which led to better nutrient assimilation.

The observed values of leaf extinction coefficient are in agreement with those reported by other authors for potato in tropical Africa (Allen and Scott, 1980). The higher leaf extinction coefficient (above 0.5) observed in pure potato stand was an indication of few vertical leaves in these treatments (Sinclair and Muchow, 1999). It thus implies that monocropped potato had more bent leaves with increased its propensity to converge solar radiation.

## 7.4.2 Soil temperature and soil water content

The high soil moisture content in the 0–30 cm in intercropping was related to the shading conferred by the greater canopy that minimized the soil evaporation losses. In sole potato, heat from the topsoil was more easily lost to evaporation through the intervening bare soil surfaces. Gitari et al. (2018b) found that intercropping potato with dolichos increased crop water productivity by 20% due to the creation of canopy shades which lowered the soil surface evaporation. The reduced ability of dolichos to confer protective shade in the UH AEZ was attributed to its low adaptation to low temperature conditions (Cook et al. 2005). The deep root systems by the legumes further minimized soil water extraction within the topsoil as they had increased capacity to extract water from the subsoils.

## 7.4.3 Effect of legume intercropping on radiation interception

The increased light interception by intercrops relative to sole potato stands could be asserted to the increased canopy size. The intercrops generally attained maximum canopy above 3, a value that corresponds to full groundcover by a typical potato cropping system (Allen and Scott, 1980). This was partly attributed to the increased number of leaves forming on lateral branches of legumes. Plants in intercropping systems were thus able to occupy all the empty niches thus contributing strongly to canopy size and radiation interception.

Because dolichos and lima bean leaves have similar leaf characteristics on their adaxial surfaces, corresponding light absorption would be expected to be identical. This was not the case in this study as sole stands of dolichos intercepted more light than lima bean in the UM and LH AEZs. Dolichos put short, dense canopy with few interior leaves relative to lima bean which established tall, broad dense crown with many interior leaves which allowed very little light to penetrate through the canopy.

The reversal of radiation interception by dolichos relative to lima bean in the UH AEZ indicated that differences in radiation interception among the cropping systems were influenced by agro-ecological zones. This observation could be explained by the fact that different crops have different thermotolerance limits (Wahid et al. 2007). Unlike legumes which indicated progressive growth with little response to prevailing heat stress, potato crop responded by developing leaves showing downward curvatures, similar to finding by Romero (2017). This observation was confirmed by the consistently higher leaf extinction coefficient in the sole potato. This mechanism greatly reduced leaf area exposure to solar radiation and thus reduced radiation interception. The mechanism was probably meant to avoid water loss by potato under extreme temperature conditions. In the UM AEZ where crops suffered longer heat stress, potato leaves drooped followed by wilting that started from the lower strata leaves. Only leaves that exhibited some level of greenness recovered turgor and finalized their production cycle. These leaves however had limited capacity to absorb solar radiation, an observation affirmed by the proportional decrease of LAI with increasing soil and ambient temperatures.

## 7.4.4 Intercropping effect on biomass accumulation and land productivity

Intercropping accumulated higher biomass and was more light use efficient compared to potato monoculture system due to their more vertical leaves as was indicated by the relatively low leaf extinction coefficient. The invariably lower dry matter yield of potato intercropped with lima bean compared to potato intercropped with dolichos in the UM and LH AEZs is primarily due to the shading effect caused by the bushy canopy of lima bean. The quality of light in terms of the ratio of light intercepted to total solar radiation reaching the potato crop was thus compromised by the understory canopy of lima bean. Burke (2017) noted that shading prolongs the stolon elongation period and delays tuberisation. When shading reduced radiation by approximately 50%

during the period of tuber initiation, tuber numbers decreased by 20%. These results strongly suggest that the amount of solar radiation intercepted by potato was indeed causal in determining dry matter accumulation.

Land productivity measured by land equivalent ratio was greater than unity indicating more efficient and productive use of land resource by intercrops. It was therefore more productive to intercrop potato with legumes than grow each component in monocultures as legume intercrops converted the intercepted radiation more efficiently into biomass. Intercropping in addition, increased the efficiency by which the soil nutrients and soil water were converted into biomass. It may therefore be inferred that intercropping is a resilient cropping system capable of stabilizing yields while guaranteeing efficient use of land resource.

# 7.4.5 Relation of potato yield, crop water productivity and radiation use efficiency to temperature and soil water content

Based on our observations, canopy and tuber growth were generally slowed during the periods of heat stress and increased thereafter when the soil temperature and soil moisture conditions improved to nearly optimal levels for potato growth. A direct consequence of this effect was shortening of the tuber bulking period. Burke (2017) noted that high soil temperatures coupled with high ambient temperatures caused premature senescence in potato with yield reduction of 20–30%. In a study conducted by Radeni and Caesar (1986), heating soil to 28°C reduced the flow of assimilates to tubers. Similarly, Krauss and Marschner (1984) observed cessation of starch accumulation when developing tubers were subjected to soil temperature of 30°C. It could also be possible that allocation of assimilated carbon into non-structural and structural carbon was altered by the high soil temperatures (Arai-Sanoh, 2010).

For the legumes, the effect of high soil temperatures were mediated in part by the deep roots. While potato crop was characterized by roots that rarely exceeded a vertical depth of 30 cm, dolichos roots were traced to 120 cm depth while lima bean roots penetrated to about 90 cm depth. Legumes could therefore access cooler subsoil layers. This is in agreement with the previous studies which have established that potato has shallow fibrous root systems which are concentrated in the topsoil layer making the crop highly sensitive to fluctuating soil moisture contents and high surface soil temperatures (Aliche et al. 2018; Gitari et al. 2018b).

Moisture deficits during root initiation period induce lignification of adventitious roots and hampers potato growth (Belehu and Hammes, 2004). This process is exacerbated under high soil temperatures. It may therefore be concluded that the consequence of aiming for the highest tuber yields in potato production is a water demand in excess of that which can consistently be met by the plant from natural sources. This view was evident in the UM zone that despite it receiving great irradiance, could not attain the high biomass yield primarily due to the low and poorly distributed rainfall.

Soil temperature interacted with soil moisture to influence the potato yield, crop water productivity and radiation use efficiency. High soil temperatures have been shown to increase the soil water flow resistance through the soil-plant-atmosphere continuum and hence reducing the plant root water uptake (Schwarz et al. 1997). Therefore, keeping soil water contents high at extreme soil temperature conditions would minimize the impact of high soil temperatures and increase the efficiency by which the intercepted light is converted into plant biomass. Prolonged heat and water stress results in root clumping, root deformations and shrinkage, and causes weak root-soil contact that limits root water uptake and transport (Trebejo and Midmore, 1990). This effect is especially true if heat stress is coupled with soil water deficit and/or an increase in soil temperature (Irmak, 2016).

Night temperatures also showed a significant effect on potato yield, crop water productivity and radiation use efficiency. Intercropping lowered the night soil temperatures to an optimal range of 15–20°C for tuber formation thus increasing tuber weight and yield. High night soil temperatures above the latter range may induce high gibberellin acid concentrations in the stolon tip with consequent disruption of tuber growth (Wilkinson and Davies, 2002). Onset of tuberization was generally delayed with night temperatures greater than 25°C, an observation associated with accelerated metabolism and growth due to induction of specific inhibitory effects. Radiation use efficiency of potato has been shown to decrease during high temperatures only if the partial stomatal closure is coupled with high night temperatures that lower net respiration (Burke, 2017). The greater specific heat capacity of a moist soil could have contributed to the yield increase observed in the UH AEZ (Hunt et al. 2010).

Radiation use efficiency was significantly lower in the UM relative to LH AEZ partly because of reduced partitioning of dry matter to tubers under heat and water stress conditions. This effect was offset to some extent in intercrops due to legumes' ability to confer shade that optimized soil temperature and soil moisture conditions. The reduced efficiency by which the solar radiation was converted into tuber dry matter in the LH AEZ appeared to be associated with the delay in the onset of tuber bulking and increased stem growth caused by the higher temperatures, an observation well explained by Bodlaender (1963). The lower soil temperatures within the optimal range for potato growth in the UH AEZ favored rapid tuber initiation resulting in greater number of tubers formed. Prolongation and greater persistence of leaf area index in the UH AEZ further contributed to extended radiation interception thus compensating for the radiation interception sacrificed earlier in the season when the low soil temperatures delayed tuber sprout.

The observed effect of season on radiation use efficiency is primarily due to variations in the amount and distribution of rainfall in relation to the potential demand for water. The larger amount and better distribution of rainfall during the long rains season increased the soil water content which in turn favored early establishment and growth of crops. On the contrary, potato suffered from severe moisture stress conditions during flowering and tuber filling stages. This greatly contributed to low vegetative growth and yield decreases during the short rains. Extremely dry soil not only restricts respiration, but also facilitates wilting due to depressed xylem and leaf tissues (Falah et al. 2010). In potatoes, this condition is generally manifest first in the newly formed leaves with consequent death if this condition continues throughout the plant.

## 7.5 Conclusion

This study demonstrate the potential role of legume intercropping in enhancing radiation interception and use efficiency of potato grown under heat and water stress conditions. Legume intercrops reduced soil temperature, hastened foliage development and canopy cover of the soil which in turn reduced soil temperature and favored tuber initiation. By extracting water from the deep soil layers and lifting it to the leaves via evapotranspiration, legumes enhanced leaf area index development under heat and water stress conditions. This enhanced radiation interception and subsequent conversion into biomass. These findings have implications with respect to potato production in tropical lowlands where high temperatures are amongst the major problems limiting the potato production. Farmers are likely to stabilize potato yields and accrue high returns from systematic legume integration into potato cropping system. These benefits can only be realized if crop diversification is based on agro-ecological and system compatibility.

## **CHAPTER EIGHT**

## GENERAL DISCUSSIONS, CONCLUSIONS AND RECOMMENDATIONS

#### 8.1 General discussions

Potato production in the lowlands and midlands is limited mainly by erratic rainfall and high ambient and soil temperatures. As soil temperatures increase, soil dries up much faster and water movement in the soil becomes limited for the shallow rooted potato crop. Chapter four of this study revealed the capacity of potato-legume intercropping to moderate the soil temperatures and increase the soil water contents. Legume intercrops developed increased root length density at greater depths which progressively extracted the free water from deeper soil layers. The deeper root system exhibited by legume intercrops provided a mechanism for continued nutrient uptake leading to divergence of the nutrients leached below the potato root zone. Since there was lack of protective canopy cover in the bare regions between the double rows of potato stands, these portions were directly exposed to solar radiation and exhibited higher soil evaporation. The resultant high soil temperatures reduced potato root growth, potentially restricting the uptake of water and soil nutrients.

As intercropping had greater positive influence on nutrient uptake at vegetative phase and water consumption at tuber bulking phase of potato growth, it becomes evident that significant water and nutrient savings can be achieved by tailoring the timing of legume incorporation to specific stages of the potato growth cycle. In general, water and nutrient deficits during stolonization and tuber initiation and bulking, had adverse effect on tuber yield.

As evident in chapter five, the high formation of SOM under legume intercropping relative to sole potato cropping was due to the high crop residue retention. When intercropped with legumes, the total biomass accumulation increased by up to 10 fold suggesting increased source

134

of SOM. The SOM content was particularly high in the third season of residue incorporation suggesting cumulative buildup of SOM. The fact that roots were in close proximity to soil microbes and soil mineral surfaces, induced interactions that directly affected the decomposer community structure and function, potentially enhancing microbial biomass and activity. It is thus apparent that progressive legume inclusion in consociation with potato crop would in the long-term restore the impoverished soil fertility in smallholder potato farms. Thus, improvement of SOM in smalholder farming systems not only requires the application of chemical fertilizers but also strategies that integrate legumes into cropping systems.

The greater microbial activity in plots with dolichos relative to those with lima bean in the UM and LH AEZs was attributed to the greater biomass generated by dolichos. This trend was reversed in the UH AEZ where low temperatures hindered dolichos growth and therefore biomass accumulation. This observation points the need to promote legume varieties adapted to the range of existing climatic conditions so as to ensure wide adaptability and stable supply of SOM. Further, the low soil organic matter contents under sole potato cropping indicated that this system might contribute to faster deterioration of soil health and quality. This was partly due to the low residue return and the high residue content of lignin and polyphenols. This caused nitrogen to be immobilized and become unavailable for subsequent crops thus restricting the microbial growth and activity.

Chapter six of this study demonstrated the capacity of potato-legume intercropping to optimize the amount of nitrogen retained in the biomass and soil, thus offering a mechanism for optimizing the soil N balance. The high acquisition of nitrogen in the surface soil layer by potato crop created depletion zones thus generating a heterogeneous pattern of resource distribution. This probably compelled the legume components to develop extensive root systems to enhance N acquisition from deeper soil layers. When legumes were grown in intercropping systems, the ratios of incorporated residue C-to-N, lignin-to-N, polyphenol-to-N and lignin + polyphenol-to-N decreased, creating favorable conditions for SOM decomposition thus resulting in net N mineralization. Most of N release occurred at a time when N uptake by potato crop was at its peak, an indication that synchrony of nutrient supply and demand was attained thus reducing excessive N accumulation. Reduction of nitrogen loss in this manner could be an effective measure to increase the efficiency of nutrient use. Thus, potato-legume intercropping is an option to help reduce N-fertilization as compared to sole potato production.

As demonstrated in chapter seven, intercropping increased biomass yield, an observation that was associated with either increase in numbers of tubers per hill or increase in individual tuber weight. The choice of legume was found to be very important and should be based on the AEZ. The yields expressed as potato equivalents were greatest in potato-lablab intercropping in the upper and lower highland AEZs and in potato-lima bean intercropping in the upper highland AEZ indicating that these systems may be preferred by smallholder potato farmers in these zones. Dolichos was better adapted to the midlands and lowlands because of its high tolerance to drought once established. The increased solar interception and use efficiency by intercropping relative to potato monocropping was asserted to the increased canopy size and duration. Factors related to genotype selection and populations were found important in the distribution of photosynthetically active radiation within the intercrop canopy.

Land equivalent ratios of intercrops were above unity indicating more efficient and productive use of land resource. In the upper midland zone, the high soil temperatures recorded in sole potato plots adversely affected potato root development making much of starch to be directed to stems and leaves at the expense of tuber expansion. It could also be possible that allocation of assimilated carbon into non-structural and structural carbon was altered by the high soil temperatures. For the legumes, the effects of high soil temperatures were mediated in part by their deep roots coupled with their good adaptations to high temperatures. In addition, the legume intercrops exhibited low leaf extinction coefficient making the system more light use efficient compared to potato monoculture system.

## 8.2 Conclusions

Systematic incorporation of legume intercrops in smallholder potato farming systems is evidently a novel technology to conserve soils in the hilly terrains and optimize soil temperatures and moisture contents in the drier lowlands. This practice achieves sustainability through increased resource productivity, enhanced risk aversion and improved profitability. Consequently, potatolegume intercropping has a high potential to adapt farmers to the adverse effects of climate change. Even though we could not directly relate the increase in potato equivalent yields to increased SOM and soil nitrogen contents, it was apparent that this advantage would occur in the long-term with legume intercropping. Thus, potato-legume intercropping system is a resilient and effective practice capable of adapting potato to the changing climatic conditions and expand production to the midland elevation agro-food systems.

## 8.3 Implications of the study

## 8.3.1 Implications to potato and legume breeders

While intercropping potato with legumes is an option to enhance climate resilience, this is only possible if the genotypes used are well adapted to the system. To the breeders, the selection of potato varieties for high yield in closely planted sole-crop stands is likely to result in varieties which are also optimal for intercropping with legumes. In cases where reductions in irradiance receipts by potato grown in intercrops is experienced, only potato genotypes possessing shade tolerance exhibits the remedy to increase biomass accumulation (Schulz et al. 2019). Selection for high initial vigor may be important for potato relayed with late maturing legumes to avoid excessive growth retardation which persists later on. In this case, indeterminate legumes are appropriate. The legumes to be be bered should be shade tolerant, have short stature with low vigor, and preferably mature in less than 90 days. Legume varieties with erect leaves would allow more light to reach the associated potato. These legumes should have a high harvest index to lower vegetative vigour, and probably have adequate compensatory growth after potato harvest. In addition, early vigorous root proliferation by legume may be a useful selection trait for maintaining yield of potato under restricted water level (Puértolas et al. 2014).

## **8.3.2 Agronomic implications**

Potato-legume intercropping seems less suitable for high nitrogen input agriculture or where soil is highly fertile, unless potato is intercropped at low density. Under very fertile soil conditions, N fixation potential is greatly reduced and the rapid growth of the legumes depresses potato growth. In such cases, low harvest index of both potato and legume lowers the nutrient use efficiency. Preferably, the legume breeds that can exude carboxylates and phosphate solubilizing enzymes are appropriate for P deficient soils while those that are high nodulating are suitable for N depleted soils. The potential synergy of legume intercropping and potato cultivars was observed in the drier midland and lowlands. Therefore, this synergy could represent an additive interaction if abiotic factors are the main limitations to potato yield.

## **8.3.3** Policy implications and recommendations

Systematic incorporation of legume intercrops in smallholder potato farming systems is evidently a novel technology to conserve soils in the highlands and optimize soil temperatures in the lowland and midlands. Consequently, this innovation has a high potential to adapt farmers to effects of climate change. Bearing this in mind, a proposal is made for a policy framework that should pay attention to the following components:

- (i) **Develop agro-ecological specific intercropping:** potato-legume intercropping must be adapted to location-specific agro-ecological zones and target multipurpose legumes providing human food, livestock feed and enhancing soil fertility.
- (ii) Strengthen local seed system: poor access to high-quality fodder and grain legumes are potential barriers to adoption of potato-legume intercropping. A functional seed production and dissemination system entailing proper assessment of seed demand must be ensured. Programs that allow integration of formal and informal seed sectors may be necessary to ensure enhanced production.
- (iii) Create attractive markets for fodder and grain legumes: farmers will adopt potato-legume intercropping if the derived produce has a ready market and the practice ensures good returns on land, capital and labor.
- (iv) Develop capacity at all levels: potato-legume intercropping should be integrated into formal and informal learning and extension systems, and implemented through field demonstrations, farmer trainings, exhibition visits and farmer field schools. Capacity building should include provision of essential public services, which facilitates access to markets and knowledge, and improves skills needed to innovate and improve resource use efficiency. Make the results/derived benefits of potato-legume intercropping available to the farmers to enhance public education and contribute to more effective management of resources in the future.
- (v) Support local research on the environmental and economic benefits of best management practices: there is need to integrate potato-legume intercropping into climate change and agricultural innovation policies. This can be achieved through enhanced research. Generally,

farmers tend to be sceptical about practices implying national standards when there is no local history of use. Alternative practices new to an area need to be locally tested in the field so that farmers can see the environmental and economic benefits first-hand. The government role in the agricultural innovation system should be focused in providing governance, funds for innovation activities, and incentives for private investment and adoption of innovation. The research approach should aim at crop diversification and must ensure transition of agriculture and food production towards agroecology.

- (vi) Provide incentives for intensification: as for the long-term benefits of potato-legume intercropping (such as improved soil fertility), there is no evidence of a rapid productivity gain. It is however certain that these gains will occur over time, but the direct benefits to productivity are probably small. Incentives such as provision of farm machineries and legume seeds to the farmers implementing potato-legume intercropping will undoubtedly intensify the adoption of this technology.
- (vii) Design and validate a participatory rural appraisal decision-support tool: the tool is needed to guide the prioritization protocols of custom tailored intercropping system, and should be evaluated through endline surveys. One aspect of the participatory approach would relate either to subsidies/direct support to farmers who would adopt potato-legume intercropping or to state environmental regulations if the problems would not be solved through voluntary participation. The farmers should be convinced that by voluntary action, they would more likely achieve satisfactory locally acceptable solutions to impoverished soil fertility and low potato productivity. Further, there should be a policy evaluation model that evaluates the impact of different potato-leume intercropping policy measures on trade and markets, through their effects on output prices, inputs prices or income.

## 8.4 Research gaps

In order to fully unravel the mechanisms underlying the observed benefits of potato-legume intercropping and unlock the technology to the smallholder farmers, the following research gaps need to be addressed:

- (i) While the residue transformation into labile organic matter fractions reflected the role of soil microbes in nutrient cycling, the role of fauna e.g. the earthworm casts, bacterial strains and mycorrhiza under potato-legume mixture remains elusive. In addition, linkages between litter fauna, age of biomass cutting, residue biochemical composition and SOM dynamics should be assessed under different temperature and moisture regimes.
- (ii) The increase in soil nitrogen balance was undoubtedly due to increased  $N_2$  fixation. Nevertheless, whether the nitrogen fixed is directly transferred to potato intercrop remains a speculation. These concerns need further exploration if resource complementarity and facilitation mechanisms are to be fully understood.
- (iii) The increase in phosphorus uptake by potato due to legume intercropping was alluded to microbial rhizospheric phosphorus facilitation and organic acid exudation. To validate such postulation, there is need to induce the activity of phosphorus solubilizing bacteria and organic acid production by different legumes in a microcosm set up. The phosphorus released should be mobilized to companion potato crop by endophytic bacterial strains under controlled environment before the experiment is furthered under field conditions.
- (iv) It is expected that when crops are intercropped, there would be benefits from root interaction or avoidance, or even allellopathic exchange in the rhizosphere of the companion crops. Only little knowledge has been unraveled in this area under potao-legume intercropping systems.

141

#### REFERENCES

- Acreche, M.M., Briceno-Felix, G., Sanchez, J.M., Slafer, G.A., 2009. Radiation interception and use efficiency as affected by breeding in Mediterranean wheat. Field Crop Res. 110, 91–97.
- Adam, G., Duncan, H., 2001. Development of a sensitive and rapid method for the measurement of total microbial activity using fluorescein diacetate (FDA) in a range of soils. Soil Biol Biochem. 33, 943–951.
- Addo-Quaye A.A., Darkwa, A.A., Ocloo, G.K., 2011. Yield and productivity of component crops in a maize-soybean intercropping system as affected by time of planting and spatial arrangement. J Agric Biol Sci. 6, 50–57.
- Alef, K., 1995. Estimation of soil respiration. In: Alef, K., Nannipieri, P., (Eds). Methods in soil microbiology and biochemistry. Academic Press, NY. pp. 464–470.
- Aliche, E.B., M. Oortwijn, Theeuwen, T.P., Bachem, C.B., Visser, R.G.F., Linden, C.G., 2018. Drought response in field-grown potatoes and the interactions between canopy growth and yield. Agric Water Manage. 206, 20–30.
- Al-Kaisi, M.M., Yin, X., Licht, M.A., 2005. Soil carbon and nitrogen changes as influenced by tillage and cropping systems in some Iowa soils. Agric, Ecosyst Environ. 105, 635–647.
- Allen, B.J., R.K., Scott, 1980. An analysis of growth of the potato crop. J Agric Sci Cam. 94, 583– 606.
- Allen, R.G., Pereira L.S., Raes, D., Smith, M., 1998. Crop evapotranspiration. Guidelines for computing crop water requirements. FAO Irrig. Drain Paper 56.FAO, Rome, 300 pp.
- Alonso-Ayuso, M., Gabriel, J.L., Quemada, M., 2014. The kill date as a management tool for cover cropping success. PLoS ONE 9(10): e109587.

- Alva, A., Moore, A.D., Collins, H.P., 2012. Impact of deficit irrigation on tuber yield and quality of potato cultivars. J. Crop Improv. 26, 1–17.
- Anderson J.M., Ingram, J.S., 1993. Tropical soil biology and fertility: a handbook of methods, 2<sup>nd</sup> edn. CAB International, Wallingford, UK.
- Anderson, T., Domsch, K.H., 1990. Application of eco-physiological quotients (qCO<sub>2</sub> and qD) on microbial biomasses from soils of different cropping histories. Soil. Biol. Biochem. 22, 251–255.
- Anderson, T.H., Domsch, K.H., 2010. Soil microbial biomass: the eco-physiological approach. Soil Biol. Biochem. 42, 2039–2043.
- Angers, D.A, Caron J., 1998. Plant-induced changes in soil structure: processes and feedbacks. Biogeochemistry. 42, 55–72.
- Arai-Sanoh, Y., Tsutomu, I., Ohsumi, O., Kondo, M., 2010. Effects of Soil Temperature on Growth and Root Function in Rice. Plant Prod Sci. 13, 235–242.
- Arregui, L.M., Quemada, M., 2006. Drainage and nitrate leaching in a crop rotation under different N-fertilizer strategies: application of capacitance probes. Plant Soil. 288, 57–69.
- Asbjornsen, H., Goldsmith, G.R., Alvarado-Barrientos, M.S., Rebel, K., Van Osch, F.P., Rietkerk,
  M., Chen, J., Gotsch, S., Tobo´n, C., Geissert, D.R., Go´mez-Tagle, A., Vache, K.,
  Dawson, T. E., 2011. Ecohydrological advances and applications in plant–water relations:
  a review. *J Plant Ecol.* 4, 3–22.
- Auerswald, K., Gerl, G., Kainz, M., 2006. Influence of cropping system on harvest erosion under potato. Soil Till. Res. 89, 22–34.

- Bachand, P.A.M., Bachand, S., Fleck, J., Anderson, F., Windham-Myers, L., 2013. Differentiating transpiration from evaporation in seasonal agriculturalwetlands and the link to advective fluxes in the root zone. Sci. Total Environ. 484, 232–248.
- Baldock, J., Macdonald, L., Farrell, M., Welti, N., Monjardino, M., 2018. Nitrogen dynamics in modern cropping systems. CSIRO Agriculture and Food, Locked Bag, Glen Osmond, SA 5064.
- Balesdent, J., Chenu, C., Balabane, M., 2000. Relationship of soil organic matter dynamics to physical protection and tillage. Soil Till Res. 53,215–230.
- Baligar, V., Fageria, N.K., He, Z.L., 2001. Nutrient use efficiency in plants, communications in soil science and plant analysis. 32, 921–950.
- Barber, S.A., 1995. Soil nutrient bioavailability- A mechanistic approach. 2nd edn. John Willey and Sons, Inc., New York.
- Bartlett, R.J., Ross, D.N., 1988. Colorimetric determination of oxidizable carbon in acid soil solutions. Soil Sci Soc Am J. 52, 1191–1192.
- Bates, D., M€achler, M., Bolker, B., Walker, S., 2015. Fitting linear mixed-effects models using lme4. J Stat Software. 67 (1), 1–48.
- Becker, M., Ladha, J.K., 1997. Synchronizing residue N mineralization with rice N demand in flooded conditions. In: Driven by nature: plant litter quality and decomposition (Cadisch G, Giller KE, eds.), pp. 231–238, CAB International, Wallingford, U.K.
- Belehu, T., Hammes, P.S., 2004. Effect of temperature, soil moisture content and type of cutting on establishment of sweet potato cuttings. S Afr J Plant Soil. 212, 85–89.
- Bethune, M.G., Selle, B., Wang, Q.J., 2008. Understanding and predicting deep percolation under surface irrigation, Water Resour. Res., 44, W12430.

- Bichel, A., Maren, O., Voroney, P., Laura, E., 2016. Sequestration of native soil organic carbon and residue carbon in complex agroecosystems, Carbon Management, 7, 261–270.
- Biswas T, Schrale, G., 2007. Sentek SoluSAMPLER: a tool for managing salt and nutrient movement in the root zone. In: Instruction Manual Version 2.0. Sentek Pty Ltd., Stepney, SA, Australia.
- Black, C., Ong., 2000. Utilization of light and water in tropical agriculture. Agri Forest Meteorol 104, 25–47.
- Blok, D., Heijmans, M.M.P.D., Schaepman-Strub, G., Kononov, A.V., Maximov, T.C., Berendse,
  F., 2010. Shrub expansion may reduce summer permafrost thaw in Siberian tundra. *Global Change Biol.* 16, 1296–1305.
- Bodlaender, K.B., 1963. Influence of temperature, radiation and photoperiod on development and yield. In: The growth of the potato (ed. J. D. Ivins and F. L. Milthorpe). London: Butterworths.
- Bohm, W., 1979. Methods of studying root systems. Springer-Verlag, Heldelberg.
- Bonde, T.A., Rosswall, T., 1987. Seasonal variation of potentially mineralizable nitrogen in four cropping systems. Soil Sci Soc Am J. 51, 1508–1514.
- Bremner, J.M., Mulvaney, C.S., 1982. Nitrogen total, In: Page, A.L., Miller, R.H., Keeney, D.R., (Eds.). Methods of soil analysis, Part 2, Chemical and microbiological Properties. 2<sup>nd</sup> edn., Agron. 9, ASA, SSSA, Madison, WI, pp. 595–624.
- Brintha, I., Seran, T.H., 2009. Effect of paired row planting of radish (*Raphanussativus* L.) intercropped with vegetable amaranthus (*Amaranthus tricolor* L.) on yield components of radish in sandy. Regosol.J Agric Sci. 4, 19–28.

- Brutsaert W., Chen, D., 1995. Desorption and the two stages of drying of natural tallgrass prairie. Water Resour Res. 31, 1305–13.
- Brye, K. R., Norman, J. M., Gower, S. T., and L. G. Bundy. 2003. Effects of management practices on annual net N-mineralization in a restored prairie and maize agroecosystems. Biogeochemistry 63(2), 135–160.
- Burke, J.J., 2017. Growing the potato crop. Vita, Equity House, Upper Ormond Quay, Dublin 7, Ireland.
- Burton, A., 2018. Influence of solution management techniques on nutrient use efficiency in hydroponically grown salad-type plants. Proquest 4.
- Caldwell, M.M., Dawson, T.E., Richards, J.H., 1998. Hydraulic lift: consequences of water efflux from the roots of plants. Oecologia. 113, 151–161.
- Camberdella, C.A., Elliot, E.T., 1992. Particulate soil organic-matter changes across a grassland cultivation sequence. Soil Sci Soc Am. J. 56, 777–783.
- Canarini, A., Dijkstra, F.A., 2015. Dry-rewetting cycles regulate wheat carbon rhizodeposition, stabilization and nitrogen cycling. Soil Biol Biochem. 81, 195–203.
- Cassman, K.G., Munns, D.N., 1980. Nitrogen mineralization as affected by soil moisture, temperature and depth. Soil Sci Soc Am J. 44, 1233–1237.
- Chabbi, A., Rumpel, C., 2009. Organic matter dynamics in agroecosystems-the knowledge gaps. Eur J Soil Sci. 60, 153–167.
- Chantigny, M.H., Angers, D.A., Rochette, P., 2002. The fate of carbon and nitrogen from animal manure and crop residues in wet and cold soils. Soil Biol Biochem. 34, 509–517.
- Chavarría, D.N., Verdenelli, R.A., Muñoz, E.J., Silvina, C.C., Restovich, B., Andriulo, A.E., Meriles, J.M., Vargas-Gil, S., 2016. Soil microbial functionality in response to the

inclusion of cover crop mixtures in agricultural systems. Span J Agric Res. 14(2), 304–316.

- Chaves, M., Pereira, S., Maroco, J., Rodrigues, L., Ricardo, P., Osorio, L., 2002. How plants cope with water stress in the field. Ann Bot. 89, 907–916.
- Chemining'wa, J.W., George, N., Muthomi, S.W., Theuri, 2007. Effect of Rhizobia inoculation on and starter-N on nodulation, shoot biomass and yield of grain legumes. Asian J Plant Sci. 6 (7), 1113–1118.
- Chen, H.Q., Hou, R.X., Gong, Y.S., 2009. Effects of 11 years of conservation tillage on soil organic matter fractions in wheat monoculture in Loess Plateau of China. Soil Till Res. 106, 85–94.
- Chen, R., Senbayram, M., Blagodatsky, S., Myachina, O., Dittert, K., Lin, X., Blagodatskaya, E., Kuzyakov, Y., 2014. Soil C and N availability determine the priming effect: Microbial N mining and stoichiometric decomposition theories. Glob Chang Biol. 20, 2356–2367.
- CIP, 2008. Catalogue of CIP potato varieties. https://research.cip.cgiar.org/germplasmistribution/red\_varie/brochure.php?variedad=392797.22. Accessed 14 June 2019.
- CIP, 2017. Evaluation and selection of heat and drought tolerance of CIP potato germplasm, International Potato Center, Lima, Peru.
- Claudia, M., Giletto, H., Echeverrı'a E., 2013. Nitrogen balance for potato crops in the southeast pampas region, Argentina. Nutr Cycl Agroecosyst. 95,73–86.
- Constantin, J., Beaudoin, N., Launay, M., Duval, J., Mary, B., 2012. Long-term nitrogen dynamics in various catch crop scenarios: test and simulations with STICS model in a temperate climate Agric Ecosyst Environ. 147, 36–46.

- Cook, B.G., Pengelly, Brown, B.C., Donnelly, J.L., Eagles, D.A., Franco, M.A., Hanson, J., Mullen, B.F., Partridge, I.J., Schultze-Kraft, R.,2005. Tropical forages: an interactive selection tool. *Lablab purpureus*. CSIRO, DPI and F (Qld), CIAT, and ILRI, Brisbane, Australia. http://www.tropicalforages. Accessed 13 June 2019.
- Courtney, J. Greg, A., Quirine, K., Jen, B., Kristen, S., 2005. Nitrogen basics-the nitrogen cycle. <u>http://cceonondaga.org/resources/nitrogen-basics-the-nitrogen-cycle</u>.
- Coxson, D.S., Looney, J.H.H., 1986. Point frame method of crop cover determination. Can. J. Bot. 64, 2464–2475.
- Daly, E., Porporato, A., 2005. A review of soil moisture dynamics: from rainfall infiltration to ecosystem response. *Environ Engineering Sci.* 22, 9–24.
- Demo, P., Lemaga, B., Kakuhenzire, R., Schulz, S., Borus, D., Barker, I., Woldegiorgis, G., Parker M.L., Schulte-Geldermann, E., 2015. Strategies to improve seed potato quality and supply in Sub-Saharan Africa: experience from interventions in five countries. In: Potato and sweetpotato in Africa: transforming the value chains for food and nutrition security, DABI, Wallingford, pp 155–67.
- Dereje, G., Tigist, A., Megersa, M., Tesfa, B., 2016. The Influence of Intercropping Sorghum with Legumes for Managementand Control of Striga in Sorghum at Assosa Zone, Benshangul GumuzRegion, Western Ethiopia, East Africa. Adv. Crop Sci. Tech. 4(5), 238.
- Dietel, J., Dohrmann, R., G., Meyer-Stüve, G.S., Turner, S., Schippers, A., Kaufhold, S., Butz-Braun, R., Condron, L.M., Mikutta R., 2017. Complexity of clay mineral formation during 120,000 years of soil development along the Franz Josef chronosequence, New Zealand. Geol Geophys. 60, 23–35.

- Dinesh, R.S., Ghoshal, C., Sheeja, T.E., Shiva, K.N., 2009. Soil microbial activity and biomass is stimulated by leguminous cover crops. J Plant Nutr Soil Sci. 172, 288–296.
- Dong, Z.B., Wang, X.B., Liu, L.Y., 2000. Wind erosion in arid and semiarid China: an overview. J. Soil Water Conserv. 55, 439–444.
- Eaglesham, A.R., Ayanaba, J.A., Rao, V.R., Eskew, D.L., 1981. Improving the nitrogen nutrition of maize by intercropping with cowpea. Soil Biol Biochem. 13,169–171.
- Egbe, O.M., 2010. Effects of plant density of intercropped soybean with tall sorghum on competitive ability of soybean and economic yield at Otobi, Benue State, Nigeria. J Cereals Oilseeds. 1(1), 1–10.
- Elfstrand, S., Baath, B., Martersson, A., 2007. Influence of various forms of green manure amendment on soil microbial community composition, enzyme activity and nutrient levels in leek. App Soil Ecol. 36, 70–82.
- Engelhardt, I.C., Welty, A., Blazewicz, S.J., Bru, D., Rouard, N. Breuil, M., Gessler, A., Galiano, L., Miranda, J., Spor, A., Barnard, R.L., 2018. Depth matters: effects of precipitation regime on soil microbial activity upon rewetting of a plant-soil system. ISME. https://doi.org/10.1038/s41396-018-0079-z.
- Eshel, G., R. Egozi, Y., Goldwasser, Y., Kashti, P., Fine, E., Hayut, H., Kazukro, B., Rubin, Z., Dar, O., Keisar, D.M., Segni, D., 2015. Benefits of growing potatoes under cover crops in a Mediterranean Climate. Agr. Ecosys. Environ. 211,1–9.
- Ewert, F., 2004. Modelling Plant Responses to Elevated CO<sub>2</sub>: How Important is Leaf Area Index? Annal Bot. 93, 619–627.
- Fabeiro, C., Olalla, M., de Juan, J., 2001. Yield and size of deficit irrigated potatoes. Agric. Water Manag. 48(3), 255–266.

- Fageria, N. K. and V. C. Baligar, 2005. Enhancing nitrogen use efficiency in crop plants. Adv Agron. 88, 97–185.
- Falah, M.A.F., Wajima, T., Yasutake, D., Sago, Y., Kitano, M., 2010. Responses of root uptake to high temperature of tomato plants (*Lycopersicon esculentum* Mill.) in soil-less culture. J Agric Tech. 6(3), 543–558.
- FAO, 1996. Agro-ecological zoning guidelines. FAO Soils Bulletin 76. Soil Resources, Management and Conservation Service. FAO Land and Water Development Division.
- FAO, 2012. Harmonized World Soil Database (version 1.2). Food Agriculture Organization, Rome, Italy and IIASA, Laxenburg, Austria (http://webarchive.iiasa.ac.at/Research/LUC/External-World-soil-database/HTML/). Accessed on 14<sup>th</sup> September, 2018.

FAOSTAT, 2017. http://www.fao.org/faostat/en/#data/QC. Accessed on 26<sup>th</sup> June, 2019.

Feddes, R.A., Kowalik, P.J., Zaradny, H., 1978. Simulation of Field Water Use and Crop Yield, 189 pp., Simulation Monograph, Pudoc, Wageningen, Netherlands.

Feije, F., Anger, V., 1972. Spot test in inorganic analysis. Anal Chem Acta. 149,363–367.

- Ferreira, T.C., Gonc, Alves, D.A., 2007. Crop-yield/water-use production functions ofpotatoes (*Solanum tuberosum* L.) grown under differential nitrogen andirrigation treatments in a hot dry climate. Agric Water Manage. 90, 45–55.
- Franchini, J.C., Gonzalez-Vila, F.J., Rodriguez, J., 2002. Decomposition of plant residues used in no-tillage systems as revealed by flash pyrolysis. J Anal Appl Pyrol. 62, 35–43.
- Gachene, C.K.K., Jarvis, N.J., Mbuvi, J.P., 1997. Soil erosion effects on soil properties in a highland area of Central Kenya. Soil Sci Soc Am J. 61, 559–564.

- Gachene, C.K.K., Nyawade, S.O., Karanja, N.N., 2019. Soil and Water Conservation: An Overview. In Zero hunger. Encyclopedia of the UN sustainable development goals, 1–15, eds. Leal, FW, Azul A, Brandli L, Ozuyar P, Wall T. Cham: Springer.
- Gachene, C.K.K., Wanjogu, S.N., Gicheru, P.T., 1989. The distribution, characteristics and some management aspects of sandy soils in Kenya. Paper presented at the International Symposium on Managing Sandy Soils, Jodhpur, India.
- García, C., Hernández, M.T., Costa, F., 1997. Potential use of dehydrogenase activity as an index of microbial activity in degraded soils. Commun Soil Sci. Plant Anal. 28, 123–134.
- Garcia, C.A., B.J., Andraski, D.A., Stonestrom, C.A., Cooper, J. Šimunek, S.W., Wheatcraf, 2011. Interacting Vegetative and Thermal Contributions to Water Movement in Desert Soil Vadose Zone J. 10, 552–564.
- García-Gil, J.C., Plaza, C., Soler-Rovira, P., Polo, A., 2000. Long-term effects of municipal solid waste compost application on soil enzyme activities and microbial biomass. Soil Biol Biochem. 32, 1907–1913.
- Gebremichael, A., Bekele, B., Tadesse, B., 2020. Evaluation of the effect of sorghum-legume intercropping and its residual effect on yield of sorghum in yeki woreda, sheka zone, Ethiopia. Int J Agric Innov Res. 9(2), 62–66.
- Gee, G.W., Bauder J.W., 1986. Particle-size analysis, In: Klute, A.: Methods of soil analysis, Part 1. Agron. 9, 2nd edn., ASA, Madison, WI, pp. 383–411.
- Generose, N., Roel, M., Cheryl A.P., 2002. Soil phosphorus dynamics and maize response to different rates of phosphorus fertilizer applied to an Acrisol in western Kenya. Plant Soil. 243, 1–10.

- Gentile, R., Vanlauwe, B., van Kessel, C., Six, J., 2009. Managing N availability and losses by combining fertilizer-N with different quality residues in Kenya. Agric Ecosyst Environ. 131,308–314.
- Ghani, A., Dexter, M., Perrott, K.W., 2003. Hot-water extractable carbon in soils: a sensitive measurement for determining impacts of fertilization, grazing and cultivation. Soil Biol Biochem. 35(9) 1231–1243.
- Giller, K.E., 2015. Nitrogen Fixation in Tropical Cropping Systems, 2<sup>nd</sup> Ed., CABI, Wallingford, p. 423.
- Gillian, A., Duncan, H., 2001. Development of a sensitive and rapid method for the measurement of total microbial activity using fluorescein diacetate (FDA) in a range of soils. Soil Biol Biochem. 33, 943–951.
- Gitari, H.I., 2018. Potato-legume intercrop effects on water and nutrients use efficiency, crop productivity and soil fertility in a Humic Nitisol, Kenya. PhD Thesis, University of Nairobi.
- Gitari, H.I., Gachene, C.K.K., Karanja, N.N., Kamau, S., Nyawade, S., Sharma, K., Schulte-Geldermann, E., 2018a. Optimizing yield and economic returns of rain-fed potato (*Solanum tuberosum* L.) through water conservation under potato-legume intercropping systems. Agric Water Manage 208, 59–66.
- Gitari, H.I., Gachene, C.K.K., Karanja, N.N., Kamau, S., Nyawade, S.O., Schulte-Geldermann, E., 2019. Potato-legume intercropping on a sloping terrain and its effects on soil physicochemical properties. Plant Soil. 438–447.
- Gitari, H.I., Karanja, N.N., Gachene, C.K.K., Kamau, S., Sharma, K., Schulte-Geldermann, E., 2018b. Nitrogen and phosphorous uptake by potato (*Solanum tuberosum* L.) and their use efficiency under potato-legume intercropping systems. Field Crops Res. 222, 78–84.

- Gitari, H.I., Shadrack, N., Kamau, S., Gachene, C.K.K., Karanja, N.N., Schulte-Geldermann, E., 2020. Agronomic assessment of phosphorus efficacy for potato (*Solanum tuberosum L*) under legume intercrops. J Plant Nutr. 43(6), 864–878.
- Goering, H.K., van Soest, P.J., 1970. Forage fibre analyses: apparatus reagents, procedures and some applications. USDA Handb. No. 379. U.S. Govt. Printing Offices, Washington, DC.
- Gough, R.E., Wolf, B., 1996. Diagnostic techniques for improving crop production, Volume 1. CRC Press.
- Graham, E.A., Rundel, P.W., Kaiser, W., Yeung, L., Stealey, M., Eric, E.M., 2012. Fine-scale patterns of soil and plant surface temperatures in an alpine fellfield habitat, White Mountains, California. Arctic, Antarctic, Alpine *Res.* 44, 288–295.
- Gregory, P.J., Atwell, B.J., 1991. The fate of carbon in pulse labeled crops of barley and wheat. Plant Soil. 136, 205–213.
- Grossnickle, S.C., 2000. Ecophysiology of northern spruce species: the performance of planted seedlings. NRC–CNRC, NRC Research Press, Ottawa, ON, Canada, 407 p.
- Gupta, V., Germida, J.J., 1988. Distribution of microbial biomass and its activity in different soil aggregate size classes as affected by cultivation. Soil Biol Biochem. 20, 777–786.
- Gustave, N., Mushagalusa, J., Ledent, X., Draye, 2008. Shoot and root competition in potato/maize intercropping: effects on growth and yield.
- Haiming, T., Xiaoping, X., Wenguang, T., Ke, W., Chao, L., Kaikai, C., 2017. Returning winter cover crop residue influences soil aggregation and humic substances under double-cropped rice fields. Rev Bras Cienc Solo.41:e0160488.

- Handayamoto, E., Cadisch, G., Giller, K.E., 1994. Nitrogen release from prunings of legume hedgerow trees in relation to quality of the prunings and incubation method. Plant Soil 160,237–248.
- Harahagazwe, D., Condori, B., Barreda, C., Bararyenya, A., Byarugaba, A.A., Kude, D.A., Lung'aho, C., Martinho, C., Mbiri, D., Nasona, B., Ochieng, B., Onditi, J., Randrianaivoarivony, J.M., Tankou, C.M., Worku, A., Schulte-Geldermann, E., Mares, V., de Mendiburu, F., Quiroz, R., 2018. How big is the potato (*Solanum tuberosum* L.) yield gap in Sub-Saharan Africa and why? A participatory approach. Open Agric. 3, 180–189.
- Hartmann, M., Lee, S., Hallam, S.J., Mohn, W.W., 2009. Bacterial, archaeal and eukaryal community structures throughout soil horizons of harvested and naturally disturbed forest stands. Environ Microbiol. 11, 3045–3062.
- Hassink, J., 1995. Decomposition rate constants of size and density fractions of soil organic matter. Soil Sci Soc Am J. 59, 1631–1635.
- Hauggaard-Nielsen, H., Gooding, M., Ambus, P., Corre-Hellou, P., Crozat, Y., Dahlmann, C., Dibet, A., von Fragstein, P., Pristeri, A., Monti, M., Jensen, E.S., 2009. Pea-barley intercropping for efficient symbiotic N-2-fixation, soil N acquisition and use of other nutrients in European organic cropping systems. Field Crop Res 113, 64–71.
- Haxeltine, A., Prentice, I.C., 1996. A general model for the light-use efficiency of primary production. Funct Ecol. 5, 551-561.
- Hayhoe, H.N., De Jong, R., 1988. Comparison of two soil water models for soybeans. Can Agric Eng. 30, 5–11.
- Haynes, R.J., 2005. Labile organic matter fractions as central components of the quality of agricultural soils: an overview. Adv Agron. 85, 221–268.

- Haynes, R.J., Tregurtha R., 1999. Effects of increasing periods under intensive arable vegetable production on biological, chemical and physical indices of soil quality. Biol Fertil Soils. 28, 259–266.
- Hellmuth, H., 2019. Effects of potato-legume intercropping and variety on potato performance in different agroecosystems of Kenya. Master's thesis. International Agricultural Development, Department of Plant Science, University of California, Davis.
- Henao, J., Baanante, C., 2006. Agricultural Production and Soil Nutrient Mining in AfricaImplications for Resource Conservation and Policy Development: Summary anInternational Center for Soil Fertility and Agricultural Development; IFDC.
- Henriksen, T.M., Breland, T.A., 1999. Decomposition of crop residues in the field: evaluation of a simulation model developed from microcosm studies. Soil Biol Biochem. 31, 1423–1434.
- Hinsinger, P., 1998. How do plant acquire mineral nutrients? Chemical processes involved in the rhizosphere. Adv Agron. 64, 225–265.
- Hinsinger, P., 2001. Bioavailability of soil inorganic P in the rhizosphere as affected by rootinduced chemical change: a review. Plant Soil. 237, 173–195.
- Hughes, G., Keatinge, P.J.M., Copper, N.F., 1987. Solar radiation interception and utilization by chickpea crops in northern Syria. J Agric Sci Cam, 108,419–424.
- Hulugalle, N.R., Lal, R., 1986. Soil water balance of intercropped maize and cowpea grown in a tropical hydromorpliic soil in Western Nigeria'. Agron J. 77, 86–90.
- Hunt, H.W., Fountain, A.G., Doran, P.T., Basagic, H., 2010. A dynamic physical modelfor soil temperature and water in Taylor Valley, Antarctica. Antarctic Sc. 22(4), 414–419.
- Inal, A., Gunes, A., Zhang, F., Cakmak, I., 2007. Peanut/maize intercropping induced changes in rhizosphere and nutrient concentrations in shoots. Plant Physiol Biochem 45, 350–356.

- Inman-Bamber, N.G., 2004. Sugarcane water stress criteria for irrigation and drying off. Field Crops Res. 89,107–122.
- Inselsbacher, E., Umana, N., Stange, R., Gorfer, M., Schüller, E., Ripka, K., Zechmeister-Boltenstern, S., Hood-Novotny, R., Strauss, J., Wanek, W., 2010. Short-term competition between crop plants and soil microbes for inorganic N fertilizer. Soil Biol Biochem. 42(2), 360–372.
- Irmak, S., 2016. Impacts of extreme heat stress and increased soil temperature on plant growth and development. UNL Extension Water Resources.
- IUSS Working Group WRB, 2015. World Reference Base for Soil Resources, 2015. International Soil Classification System for Naming Soils and Creating Legends for Soil Maps. World Soil Resources Reports No. 106. FAO, Rome.
- Jackson, M.L., 1967. Soil chemical analysis. Prentice Hall of India Private Limited, New Delhi.
- Jackson, R.B., Moore, L.A., Hoffmann, W.A., Pockman, W.T., Linder, C.R., 1999. Ecosystem rooting depth determined with caves and DNA. Proc Natl Acad Sci. 96, 11387–11392.
- Jaetzold, R., Hornetz, B., Shisanya, C.A, Schmidt, H (Eds) 2012. Farm Management Handbook of Kenya. Vol. I-IV (Western, Central, Eastern, Nyanza, Southern Rift Valley, Northern Rift Valley, Coast), Nairobi. Available at: https:// www.uni-trier.de/index.php?id=58581.
- Jaetzold, R., Schmidt, H., Hornetz, B., Shisanya, C.A., 2006. Farm Management Handbook of Kenya. Natural Conditions and Farm Information Volume II, Part 2/B2, second edition. Ministry of Agriculture/GTZ, Nairobi, Kenya.
- Jagadamma, S., Lal, R., Hoeft, R.G., Nafziger, E.D., Adee, E.A., 2007. Nitrogen fertilization and cropping systems effects on soil organic carbon and total nitrogen pools under chisel-plow tillage in Illinois. Soil Till Res. 95, 348–356.

- Jäger, N., Stange, C.F., Ludwig, B., Flessa, H., 2011. Emission rates of N2O and CO<sub>2</sub> from soils with different organic matter content from three long-term fertilization experiments-a laboratory study. Biol Fertil Soils. 47, 483–494.
- Jajang, S.H., Suradinata, Y.R., 2015. Effects of row intercropping system of corn and potato and Row spacing of corn on the growth and yields of Atlantic potato cultivar planted in medium altitude. Asian J Agric Res. 9, 104–112.
- Jantalia, C.P., Halvorson, A.D., Follett, R.F., Bruno, J.R.A., Jose, C.P., Segundo, U., 2012. Nitrogen source effects on ammonia volatilization as measured with semi-Static chambers. Agron J 104, 1595–1603.
- Jarecki, M.K., Lal, R., 2003. Crop management for soil carbon sequestration. Crit Rev Plant Sci. 22, 471–503.
- Jégo, G., Martínez, M., Antigüedad, I., Launay, M., Sanchez-Pérez, J.M., Justes, E., 2008. Evaluation of the impact of various agricultural practices on nitrate leaching under the root zone of potato and sugar beet using the STICS soil-crop model. Sci Total Environ. 394, 207–21.
- Jemison, J.M., 2019. Use of Nurse Crops in Potato Production to Protect Soils from Erosion. Am. J. Potato Res. 96, 13–20.
- Jensen, E.S., 1996. Barley uptake of N deposited in the rhizosphere of associated field pea. Soil Biol Biochem. 28, 159–168.
- Jimenez, C., Tejedor, M., Rodriguez, M., 2007. Influence of land use changes on soil temperature regime of andosols on Tenerife, Canary Island, Spain. Europ J Soil Sci. 58(2), 445–449.
- Jones, C.A., 1984. Estimation of an active fraction of soil nitrogen. Commun Soil Sci Plant Anal 15, 23–32.

- Jones, D.L., Shannon, D., Murphy, D.V., Farrar, J., 2004. Role of dissolved organic nitrogen (DON) in soil N cycling in grassland soils. Soil Biol Biochem 36, 749–756.
- Jumpponen, A., Jones, K.L., Blair, J., 2010. Vertical distribution of fungal communities in tallgrass prairie soil. Mycologia. 102, 1027–1041.
- Kaguora, E., Osena, E., Rono, M., Nyongesa, M., 2015. Potato variety catalogue. National Potato Council of Kenya.
- Kapkiyai, J., N.K., Karanja, J.N., Qureshi, P.C., Smithson, P.L., Woomer, 1999. Soil organic matter and nutrient dynamics in a Kenyan nitisols under long-term fertilizer and organic input management. Soil Biol Biochem. 3, 1773–1782.
- Karanja, M., 2013. Analysis of rainfall variability on Irish potato production in Ol-joro-orok division, Nyandarua County, Kenya. Thesis, Kenyatta University.
- Karuku, G.N., Gachene, C.K.K., Karanja, N.N., Cornelisb, N., Verplacke, H., 2014. Effect of different cover crop residue management practices on soil moisture content under a tomato crop (*Lycopersicon esculentum*). Trop Subtrop Agroecosyst 17, 509–523.
- Keeney, D.R., Nelson, D.W., 1982. Nitrogen in organic forms. Pages 643–698 in A. L. Page et al. Eds. Methods of soil analysis. Part 2. Agronomy No. 9, American Society of Agronomy, Madison, WI.
- Kelly, A., Zarka, K., Donna, C., Douches, D., Robin, S., Buell, C., 2017. A Guide to Growing Potatoes in Your Home Garden. Mitchigan State University. http://potatobg.css.msu.edu/Outreach%20and%20Education/Garden%20Guides/PotatoG uideFinal.pdf.
- Kemanian, A.R., Stockle, C.O., Huggins, D.R., 2004. Variability of radiation-use efficiency of different tropical crops. Crop Sci 44, 1662–1672.

- Kermaha, M., Angelinus, C., Frankeb, S.A., Benjamin, D.K., Ahiabord, R.C., Giller, K.E., 2017.Maize-grain legume intercropping for enhanced resource use efficiency and crop productivity in the Guinea savanna of northern Ghana. Field Crops Res. 213, 38–50.
- Kibunja, C., Keziah, N., Ndungu-Magiroi, W., Dixon, K., Wamae, J., Mwangi, L.N., Mary, N., Koech, J. Erastus, M.K., 2018. Optimizing fertilizer use within the context of integrated soil fertility management in Kenya. <u>http://africasoilhealth.cabi.org/wpcms/wp-</u> content/uploads/2017/03/Optimizing-Fertilizer-ISFM-in-Kenya-Ch-7-high-res.pdf.
- Kimetu, J.M., Mugendi, D.N., Bationo, A., Palm, C.A., Mutuo, P.K., Kihara, J., Nandwa, S., Giller, K.E., 2006. Partial balance of nitrogen in a maize cropping system in humic nitisol of Central Kenya. Nutr Cycl Agroecosyst. 76, 261–270.
- Kinama, J.M., Ong, C.K., Stigter, C.J., Ng'ang'a, J.K., Gichuki, F.N., 2011. Hedgerow intercropping maize or cowpea/senna for dry matter production in Semi-Arid Eastern Kenya. J Agric Sci Technol. 372–384.
- King, C., 2017. A novel option to combat soil erosion. <u>https://spudsmart.com/novel-option-</u> combat-soil-erosion/. Accessed on 31<sup>st</sup> July 2020.
- King, H.G.C., Heath, G.W., 1967. The chemical analysis small samples of leaf material and the relationship between the disappearance and composition of leaves. Pedobiologia 7, 192–197.
- Kirkegaard, J.A., Hunt, J.R., McBeath, T.M., 2014. Improving water productivity in the Australian Grains industry – a nationally coordinated approach. Crop Pasture Sci. 65, 583–601.
- Kong, A.Y.Y., Scow, K.M., Córdova-Kreylos, A.L., Holmes, W.E., Six, J., 2011. Microbial community composition and carbon cycling within soil microenvironments of conventional, low-input, and organic cropping systems. Soil Biol Biochem. 43, 20–30.

- Koocheki, A., Mahallati, M.N., Solouki, H., Karbor, S., 2016. Evaluation of Radiation Absorption and use Efficiency in Substitution Intercropping of Sesame (*Sesamum indicum* L.) and Mung Bean (*Vigna radiata* L.). Adv Plants Agric Res 3(5), 001–09.
- Krauss, A., Marschner, H., 1984. Growth rate and carbohydrate metabolism of potato tubers exposed to high temperatures. Potato Res 27, 297–303.
- Kumar, K., Goh, K.M., 2002. Management practices of antecedent leguminous and non– leguminous crop residues in relation to winter wheat yields, nitrogen uptake, soil nitrogen mineralization and simple nitrogen balance. Europ J Agron. 16, 295–308.
- Kumar, P., Tarafdar, J.C., 2003. 2, 3, 5-triphenyl tetrazolum chloride (TTC) as electron acceptor of culturable soil bacteria, fungi and actinomycetes. Biol Fertil Soils. 38, 186–189.
- Lafta, A.M., Lorenzen, J.H., 1995. Effect of high temperature on plant growth and carbohydrate metabolism in potato. Plant Physiol, 109, 637–643.
- Lahti, M., Aphalo, P.J., Finér, L., Lehto, T., Leinonen, I., Mannerkoski, H., Ryyppö, A., 2002.
  Soil temperature, gas exchange and nitrogen status of 5-year-old Norway spruce seedlings.
  Tree Physiol. 22, 1311–1316.
- Lal, R., Kimble, J., Follett, R., 1997. Soil quality management for carbon sequestration. p. 1–8. In:
  R. Lal et al. (ed.) Soil properties and their management for carbon sequestration. United
  States Department of Agriculture, Natural Resources Conservation Ser vices, National Soil
  Survey Center, Lincoln, NE.
- Levy, D., Coleman, W.K., Veilleux, R.E., 2013. Adaptation of potato to watershortage: irrigation management and enhancement of tolerance to droughtand salinity. Am J Potato Res. 90, 186–206.

- Li, X., Jiri, S., Haibin, S., Jianwen, Y., Zunyuan, P., Xuewen, G., 2017. Spatial distribution of soil water, soil temperature, and plant roots in adrip-irrigated intercropping field with plastic mulch. Europ J Agron 83, 47–56.
- Li, X., Ming'an, S., Chunlei, Z., Xiaoxu, J.,2018. Spatial variability of soil water content and related factors across the Hexi Corridor of China. Journal of Arid Land, <u>https://doi.org/10.1007/s40333-018-0123-x</u>.
- Liang, S., 2013. Short-term effects of legume winter cover crop management on soil microbial activity and particulate organic matter. (Under the direction of Dr. Wei Shi). Thesis. North Carolina State University. pp 89.
- Lithourgidis, A.S., Dordas, C.A., Damalas, C.A., Vlachostergios, D.N., 2011. Annual intercrops: an alternative pathway for sustainable agriculture. Australian Journal of Crop Science 5(4), 396–410.
- Lizaso, J.I., Batchelor, W.D., Westgate, M.E., Echart, L., Echarte, L., 2003. Enhancing the ability of CERES-Maize to compute light capture. Agric Syst. 76, 293–311.
- Loecke, T.D., Cambardella, C.A., Liebman, M., 2012. Synchrony of net nitrogen mineralization and maize nitrogen uptake following applications of composted and fresh swine manure in the Midwest U.S. Nutr Cycl Agroecosyst. 93, 65–74.
- Loeppmann, S., Blagodatskaya, E., Pausch, J., Kuzyakov, Y., 2016. Substrate quality affects kinetics and catalytic efficiency of exo-enzymes in rhizosphere and detritusphere. Soil Biol Biochem. 92, 111–118.
- Loi, L.S., 2000. Intercropping cassava with legume crops in the northern mountainous region of Vietnam. In: Cassava Research and Extension in Vietnam. Proc. National Workshop, held in Ho Chi Minh city, Vietnam. March 16-18. pp. 160-170. (in Vietnamese).

- Loomis, R.S., Connor, D.J., 1992. Crop ecology: productivity and management in agricultural systems. Cambridge University Press, Cambridge. 10, 8–538.
- Louis, B.P., Maron, P., Viaud, V., Leterme, P., Menasseri-Aubry, S., 2016. Soil C and N models that integrate microbial diversity. Environ Chem Lett. 14(3), 331–344.
- Lu, N., Chen, S., Wilske, B., Sun, G., Chen, J.,2011. Evapotranspiration and soil water relationships in a range of disturbed and undisturbed ecosystems in the semi-arid Inner Mongolia, China. J Plant Ecol. 4 (1–2), 49–60.
- Ludwig, F., Dawson, T.E., Kroon, H., Berendse, F., Prins, H.H.T., 2003. Hydraulic lift in *Acacia tortilis* trees on an East African savanna. Oecologia. 134, 293–300.
- Lupwayi, N.Z., Clayton, G.W., O'Donovan, J.T., Harker, K.N., Turkington, T.K., Soon, Y.K., 2006. Nitrogen release during decomposition of crop residues under conventional and zero tillage. Can J Soil Sci. 86, 11–19.
- Lupwayi, N.Z., Haque, I., 1999. Leucaena hedgerow intercropping and cattle manure application in the Ethiopian highlands I. Decomposition and nutrient release. Biol Fertil Soils. 28,182– 195.
- Lupwayi, N.Z., Rice, W.A., Clayton, G.W., 1998. Soil microbial diversity and community structure under wheat as influenced by tillage and crop sequence. Soil Biol Biochem. 30, 1733–1741.
- Ma, B., Yuxin, L., Xiaojun, L., Fan, M., Faqi, W., Zhanbin, L., 2015. Soil splash detachment and its spatial distribution under corn and soybean cover. Catena. 127, 142–151.
- MacMillan, J.K., Buchanan, R., 1988. Soil organic matter. Publication no 88–012, agdex 536/540.
   Plant Industry Branch, New Brunswick Department of Agriculture and Rural Development, Fredericton, NB.
- Magdoff, F., 1991. Understanding the Magdoff pre-sidedress nitrate test for corn. J Prod Agric 4(3), 297–305.
- Mahdian, Gallichand, 1996. Modeling soil water content and pressure head with SWACROP in potato fields. Departement de Genie Rural. FSAA. Universite Laval. Quebec, QC. Callada. G1K 7P4.
- Maitra, S., Tanmoy, S., Pradipta, B., 2020. Potential and Advantages of Maize-Legume Intercropping System. DOI: 10.5772/intechopen.91722.
- Massignamet, A.M., Chapman, S.C., Hammer, G.L., Fukai, S., (2011). Effects of nitrogen supply on canopy development of maize and sunflower. Crop Pasture Sci. 62, 1045–1055.
- McDaniel, M.D., L.K., Tiemann, A.S., Grandy, 2014. Does agricultural crop diversity enhance soil microbial biomass and organic matter dynamics? A meta-analysis. Ecological Applications 24(3), 560–70.
- McMichael, B.L., Quisenberry, J.E., 1993. The impact of the soil environment on the growth of root systems. Environ Expt Bot. 33, 53–61.
- Merino, C., Nannipieri, P., Matus, F., 2015. Soil carbon controlled by plant, microorganism and mineralogy interactions. Soil Sci Plant Nutri. 15(2), 321–332.
- Mitchell, R.D.J., Harrison, R., Webb, J., 2000. The effect of crop residue incorporation date on soil inorganic nitrogen, nitrate leaching and nitrogen mineralization. Biol Fertil Soils. 32, 294–301.
- MoALF, 2016. Climate risk profile for Nyandarua. Kenya County climate risk profile series. The Kenya Ministry of Agriculture, Livestock and Fisheries (MoALF), Nairobi, Kenya.

- Mohammed, I.B., Olufajo, O.O., Singh, B.B., Miko, S., Mohammed, S.G., 2008. Evaluation of yield of components of sorghum/cowpea intercrops in the Sudan savannah ecological zone. Agric. Biol. Sci. 3, 30–37.
- Monteith, J.L., 1965. Radiation and crops. Experimen Agric. 1(4), 241–251.
- Monteith, J.L., 1977. Climate and the efficiency of crop production in Britain. Philosoph Transact Royal Soc Lond. 281, 277–294.
- Monteith, J.L., 1994. Validity of the correlation between intercepted radiation and biomass. Agric Forest Meteorol. 68, 213–220.
- Monteith, J.L., Unsworth, M., 1990. Principles of environmental physics (2nd Ed.). Edward. Arnold, London.
- Monti, M.A.P. Giuseppe, B.G., Antonio, G., 2019. Cereal/grain legume intercropping in rotation with durum wheat in crop/livestock production systems for Mediterranean farming system. Field Crops Res. 240, 23–33.
- Muchena, F.N., Gachene, C.K.K., 1988. Soils of the highland and mountainous areas of Kenya with special emphasis on Agricultural soils. Mt Res Dev. 8, 183–191.
- Müller-Stöver, D., Hauggaard-Nielsen, H., Eriksen, J., Ambus, P., Johansen, A., 2012. Microbial biomass, microbial diversity, soil carbon storage, and stability after incubation of soil from grass–clover pastures of different age. Biol Fert Soils. 48,371–383.
- Mummey, D.L., Holben, W.E., Six, J., Stahl, P.D., 2006. Spatial stratification of soil microbial populations in diverse soils: Rubrobacteria and Gemmatimonads are abundant in water-stable microaggregate interiors while Acidobacteria are primarily associated with macroaggregates. Microb Ecol. 51, 404–411.

- Murakami, H., Vecchi, G.A. Villarini, G., Delworth, T., Gudgel, R., Underwood, S., Yiang, X., Zhang, W., Lin, S. 2016. Seasonal forecasts of major hurricanes and landfalling tropical cyclones using a high-resolution GFDL coupled climate model, J Climate, 29, 7977– 7989.
- Murphy, D.V., Sparling, G.P., Fillery, I.R.P., 1998. Stratification of microbial biomass C and N and gross N mineralization with soil depth in two contrasting Western Australian Agricultural soils. Aust J Soil Res. 36, 45–56.
- Murphy, J., Riley, J.P., 1962. A modified single solution method for the determination of phosphate in natural waters. Anal Chim Acta. 27, 31–36.
- Murungu, F.S., Chiduza, C., Muchaonyerwa, P., Mnkeni, P.N.S., 2011. Mulch effects on soil moisture and nitrogen, weed growth and irrigated maize productivity in a warm-temperate climate of South Africa. Soil Till Res. 112, 58–65.
- Muthoni, J., 2016. Soil fertility situation in potato producing Kenyan highlands -case of KALRO-Tigoni. Int J Hortic. 25.
- Muthoni, J., Shimelis, H., Melis, R., 2013. Potato production in Kenya: farming systems and production constraints. J Agric Sci. 5, 182–197.
- Mylavarapu, R.S., Sanchez, J.F., Nguyen, J.H., Bartos, J.M., 2002. Evaluation of Mehlich-1 and Mehlich-3 extraction procedures for plant nutrients in acid mineral soils of Florida."
  Communications in Soil Science and Plant Analysis. 33(5–6), 807–200.
- Naresh, R.K., Timsina, J., Bhaskar, V., Gupta, R.K., Singh, A.K., Dhaliwa S.S., Rathore, R.S., Kumar, V., Singh, P., Singh, S.P., Tyagi, S., Kumar, Sunil, Chandra, N., Mahajan, 2017. Effects of tillage, residue and nutrient management on soil organic carbon dynamics and

its fractions, soil aggregate stability and soil carbon sequestration: a review. EC Nutrition. 12(2), 53–80.

- Natarajan, M., Willey, R.W., 1980. Sorghum-pegion pea intercropping and the effects of plant population density. 2. Resource use. J Agric Sci. 95, 5–65.
- Nelson, D.W., Sommers, L.E., 1996. Carbon and organic matter, in Sparks, D.L., Page, A.L.,
  Helmke, P.A., Loeppert, R.H., Soltanpour, P.N., Tabatabai, M.A., Johnston, C.T., Sumner,
  M.E.: Methods of soil analysis, Part 3, Chemical Analysis. SSSA, ASA, Madison, WI, pp. 961–1010.
- NPCK, 2017. National Potato Council of Kenya, Potato variety catalogue. <u>https://www.google.com/search?client=firefox-b-&q=POTATO+VARIETYCATALOGUE2017</u>. Accessed 19 Feb 2019.
- Nyawade, O.S., 2015. Effect of potato (Solanum tuberosum L.) cropping systems on soil and nutrient losses through run-off in a humic nitisol. MSc. Thesis. University of Nairobi.
- Nyawade, S., N., Karanja, C.K.K., Gachene, E., Schulte-Geldermann, M., Parker 2018a. Effect of potato hilling on soil temperature, soil moisture distribution and sediment yield on a sloping terrain. Soil Till Res. 184, 24–36.
- Nyawade, S., Vandamme, E., Friedmann, M., Parker, M. 2020. Potato-legume intercropping enhances climate resilience and adaptive capacity of smallholder farmers. Research Brief 03. Lima, Peru: International Potato Center. 4 p. <u>https://hdl.handle.net/10568/108045.</u>
- Nyawade, S.O., Gachene, C.K.K., Karanja, N.N., Gitari, H.I., Schulte-Geldermann, Parker, P., 2019. Controlling soil erosion in smallholder potato farming systems using legume intercrops. Geoderma Reg 15 e00225.
- Nyawade, S.O., Karanja, N.N., Gachene, C.K.K., Gitari, H.I., Schulte-Geldermann, E., Parker, M., 2020a. Optimizing soil nitrogen balance in a potato cropping system through legume intercropping. Nutr Cycl Agroecosyst, 117: 43–59.

- Nyawade, S.O., Karanja, N.N., Gachene, C.K.K., Schulte-Geldermann, E., Parker, L.M., 2018b Susceptibility of soil organic matter fractions to soil erosion under potato-legume intercropping systems in central Central Kenya. J Soil Water Conserv. 73, 5678–5767.
- Nyawira L., 2018. Potato farmers abandon crop, despite county free seed program. <u>https://www.standardmedia.co.ke/business/article/2001300315/potato-farmers-abandon-</u> <u>crop-despite-county-free-seed-program</u>
- O'Neil Kathleen, A., 2013. Impact of rotation, cover crop and manure inputs on productivity, soil organic matter fractions and soil nitrogen in irrigated Michigan potato cropping systems. Dissertation. Michigan State University, Department of Crop and Sciences. pp 260.
- OECD, 2001. Environmental indicators for agriculture, methods and results. Paris, France: OECD publications service.
- Ong, C.K., Black, C.R., 1992. Complementarity in resource use in intercropping and Agroforestry systems. Paper presented at the 52nd International Easter School on "Resource capture by crops", School of Agriculture, Sutton, Bonington, England.
- Onwuka, B., Mang, M., 2018. Effects of soil temperature on some soil properties and plant growth. Adv Plants Agric Res 8(1), 34–37.
- Osterholz, W.R., 2016. Nitrogen pools and fluxes in diversified cropping systems. Graduate Theses and Dissertations. 15059. 15059. https://lib.dr.iastate.edu/etd/15059.
- Palm CA, Sanchez PA (1991) Nitrogen release from some tropical legumes as affected by lignin and polyphenol contents. Soil Biol Biochem. 23, 83–88.
- Parker, M.L., Low J.W., Andrade M., Schulte-Geldermann E., Andrade-Piedra J., 2019. Climate Change and Seed Systems of Roots, Tubers and Bananas: The Cases of Potato in Kenya

and Sweetpotato in Mozambique. In: Rosenstock T., Nowak A., Girvetz E. (eds) The Climate-Smart Agriculture Papers. Springer, Cham.

- Paul E.A., Clark, F.E., 1989. Soil microbiology and biochemistry. Academic Press, Orlando, FL.
- Pavlista, A.D., 2015. Scheduling reduced irrigation on 'Atlantic' potato for minimaleffect. Am J Potato Res. 92, 673–683.
- Peoples, M.B., Giller, K.E., Herridge, D.F., Vessey, J.K., 2002. Limitations to biological nitrogen fixation as a renewable source of nitrogen for agriculture. In: Finan T, O'Brian M, Layzell D, Vessey K, Newton W (Eds.), Nitrogen Fixation: Global.
- Petr, D., Tomášek, J., Kuchtová, P., Hamouz, K., Hajšlová, J., Schulzová, V., 2012. Effect of mulching materials on potato production in different soil-climatic conditions. Rom Agric Res. 29, 201–209.
- Pielke, R. A., 2001. Influence of the spatial distribution of vegetation and soils on the prediction of cumulus convective rainfall. Review Geophysics. 39, 151–177.
- Pikul, J.J.L., Osborne, S., Ellsbury, M., Riedell, W., 2007. Particulate organic matter and waterstable aggregation of soil under contrasting management. Soil Sci Soc Am J. 71,766–776.
- Plaza-Bonilla, D., Jean-Marie, N., Didier R., Justes, E., 2015. Cover crops mitigate nitrate leaching in cropping systems including grain legumes: Field evidence and model simulations. Agric Ecosyst Environ. 212, 1–12.
- Polgar, Z., Laszlo, G., Zsolt, V., Istvan, W., 2017. Manifestation of the abiotic stress tolerance of the new Hungarian potato varieties. In: Twentieth European Association for Potato Research (EAPR) Triennial Conference. Versailles, France, July 9–14, 2017.

- Poly, F., Ranjard, L., Nazaret, S., Gourbiere, F., Jocteur-Monrozier, L., 2001. Comparison of nifH gene pools in soils and soil microenvironments with contrasting properties. Appl Environ Microbiol. 67, 2255–2262.
- Prasad, R., Hochmuth, G., 2016. How to calculate a partial nitrogen mass budget for potato. Soil and water sciences department, UF/IFAS Extension. http://edis.ifas.ufl.edu.\
- Prasad, R., Hochmuth, G.J., Boote, K.J., 2015. Estimation of nitrogen pools in irrigated potato production on sandy soil using the model SUBSTOR. PLoS ONE 10(1), e0117891.
- Qiang, C., Xiapong, Z., Zewdi, T., Seppo, K., Dengyu, L., Kristina, L., 2003. Diversity of peanut (*Arachis hypogaea*) bradyrhizobia and their host plants. Plant and Soil. 255, 605–617.
- R Core Team, 2017. R: a Language and Environment for Statistical Computing. R Foundation for Statistical Computing, Vienna, Austria.
- Radeni, G., Caesar, K., 1986. Effects of soil temperature on the carbohydrate status in the potato plant. J Agron Crop Sci. 156, 217–24.
- Rakshit, R., Patra, A.K., Pal, D., Kumar, M., Singh, R., 2012. Effect of elevated CO<sub>2</sub> and temperature on nitrogen dynamics and microbial activity during wheat (Triticum aestivum L.) growth on a subtropical inceptisol in India. J Agron Crop Sci. 198, 452–465.
- Rana, S.S., Rana, M.C., 2011. Cropping System. Department of Agronomy, College of Agriculture, CSK Himachal Pradesh Krishi Vishvavidyalaya, Palampur, 80 pages.
- Rasmussen, P.E., Rohde, C.R., 1988. Stubble burning effects on winter wheat yield and nitrogen utilization under semiarid conditions. Agron J. 80, 940–942.
- Reddy K.N., Zablotowicz R.M., Locke M.A., Koger C.H., 2003. Cover crop, tillage, and herbicide effects on weeds, soil properties, microbial populations, and soybean yield. Weed Sci. 51 (6), 987–994.

- Ren, J., Zhang, L., Duan, Y., Zhang, J., Evers, J. B., Zhang, Y., van der Werf, W., 2018. Intercropping potato (*Solanum tuberosum* L.) with hairy vetch (Vicia villosa) increases water use efficiency in dry conditions. Field Crops Res. doi:10.1016/j.fcr.2018.12.002.
- Romero, A.P., Alarcón, A., Valbuena, R.I., Galeano, C.H., 2017. Physiological Assessment of Water Stress in Potato Using Spectral Information. Front Plant Sci. 8. 1608.
- Roper, M., Gupta, S.R., Daniel, V., Murphy, J., 2010. Tillage practices altered labile soil organic carbon and microbial function without affecting crop yields. Aust J Soil Res. 48, 274–285.
- Rosa, R.D., Paredes, P., Rodrigues, G.C., Alves, I., Fernando, R.M., Pereira, L.S., Allen, R.G.,
  2012. Implementing the dual crop coefficient approach in interactive software. 1.
  Background and computational strategy. Agric Water Manage. 103, 8–24.
- Ruijter, F.J., Huijsmans, J.F.M., Rutgers, B., 2010. Ammonia volatilization from crop residues and frozen green manure crops. Atmos Environ 44, 3362–3368.
- Rutherford, P.M., Juma, N.G., 1992. Influence of soil texture on protozoa induced mineralization of bacteria carbon and nitrogen. Can J Soil Sci. 72, 183–200.
- Rykaczewska, K., 2015. The effect of High temperature occurring in subsequent stages of plant development on potato yield and tuber physiological defects. Am J Potato Res. 92, 339–349.
- Rykaczewska, K., 2017. Impact of heat and drought stresses on size and qualityof the potato yield. Plant Soil Environ. 63, 40–46.
- Sakai, M., Toride, N., Šimůnek, J., 2009. Water and vapor movement with condensation and evaporation in a sandy column. Soil Sci Soc Am J. 73,707–717.
- Salisbury, F.B., Spomer, G.G., 1964. Leaf temperatures of alpine plants in the field. *Planta*. 60, 497–505.

- Sanchez, P.A., 2002. Ecology: soil fertility and hunger in Africa. Science, 295(5562), 2019–2020. https://doi.org/10.1126/ science.1065256.
- Sandana, P. 2016. Phosphorus uptake and utilization efficiency in response to potato genotype and phosphorus availability. Eur J Agron. 76, 95–106.
- Sándor, R., Fodor, N., 2012. Stimulation of soil temperature dynamics with models using different concepts. Scientific world J. 200, 12–20.
- Sanginga, N., Woomer, P.L., 2009. Integrated soil fertility management in Africa: principles, practices and development process, tropical soil biology and fertility institute of the international centre for tropical agriculture, Nairobi.
- Scanlon, B.R., Keese, K., Reedy, R.C., Šimůnek, J., Andraski, B.H., 2003. Variations in flow and transport in thick desert vadose zones in response to paleoclimacti c forcing (0–90 kyr):
   Field measurements, modeling, and uncertainties. Water Resour Res. 39(7), 1179.
- Scanlon, B.R., Levi, D.G., Reedy, R.C., Keese, K.E., Sully, M.J., 2005. Ecological controls on water-cycle response to climate variability in deserts. Proc Natl Acad Sci. 102, 6033–6038.
- Schimel, J.P., and Bennett, J., 2004. Nitrogen mineralization: Challenges of a changing paradigm. Ecology 85(3), 591–602.
- Schneider, K.D., Martens, J.R.T.F., Zvomuya, D.K., Reid, T.D., Fraser, D.H., Lynch, I.P.,
  O'Halloran, Wilson, H.F., 2019. Options for improved phosphorus cycling and use in agriculture at the field and regional scales. J Environ Qual. 48 (5),1247–64.
- Schnurer J., Rosswall T., 1982. Fluorescein diacetate hydrolysis as a measure of total microbial activity in soil and litter. Appl Environ Microbiol. 43, 1256–1261.
- Schulte-Geldermann, E., 2013. Tackling low potato yields in Eastern Africa: an overview of constraints and potential strategies. <u>https://cgspace.cgiar.org/handle/10568/57049.</u>

- Schulte-Geldermann, E., Gildemacher, P.R., Struik, P.C., 2012. Improving seed health and seed performance by positive selection in three Kenyan potato varieties. Am J Potato Res. 89, 429–437.
- Schwarz, P.A., Fahey, T.J., Dawson, T.E., 1997. Seasonal air and soil temperature effects on photosynthesis in red spruce (*Picea rubens*) saplings. Tree Physiol 17(3), 187–194.
- Shah, S.F.A., McKenzie, B.A., Gaunt, R.E., Marshall, J.W., Frampton, C.M., 2004. Effect of production environments on radiation interception and radiation use efficiency of potato (*Solanum tuberosum* L.) grown in Canterbury, New Zealand, New Zealand J Crop Hort Sci. 32 (1), 113–119.
- Shekofteh, H., Majid, A., Mohammad, A.H., Iversen, V., Nezamabadi-Pour, H., Fariborz, A., Farid, S., Shirani, H., 2013. Modeling of nitrate leaching from a potato field using HYDRUS-2D. Soil Sci Plant Anal 44, 1–15.
- Sicardi, M., García-Prechac, F., Frioni, L., 2004. Soil microbial indicators sensitive to land use conversion from pastures to commercial Eucalyptus grandis (Hillex Maiden) plantations in Uruguay. Appl Soil Ecol. 27, 125–133.
- Sinclair T.R., Muchow, 1999. Radiation use efficiency. Adv agron. 65, 215–265.
- Sinclair, T.R., 2000. Model analysis of plant traits leading to prolonged crop survival during severe drought. Field Crops Res. 68, 211–217.
- Singh, A.K., Bharati, R.C., Chadra, N., 2013. Response of intercrops and nutrient management on the performance of tobacco based intercropping system and assessment of system sustainability. Bangladesh J Bot 42(2), 343–348.
- Six, J., Conant, R.T., Paul, E.A., Paustian, K., 2002. Stabilization mechanisms of soil organic matter: Implications for C- saturation of soils. Plant Soil. 241,155–176.

- Six, J., Elliott, E.T., Paustian, K., 2000. Soil macroaggregate turnover and microaggregate formation: a mechanism for C sequestration under no-tillage agriculture. Soil Biol Biochem. 32, 2099–2103.
- Smith, J.L., Paul, E.A., 1990. The significance of soil microbial biomass estimations. In "Soil Biochemistry" (J. M. Bollag and G. Stotzky, Eds.), 6, 357–396. Marcel Dekker, New York, NY.
- Smolander, A., Kitunen, V., 2002. Soil microbial activities and characteristics of dissolved organic C and N in relation to tree species. Soil. Biol Biochem. 34, 651–660.
- Snoeck, D., Zapata, F., Domenach, A.M., 2000. Isotopic evidence of the transfer of nitrogen fixed by legumes to coffee trees. Biotechnol Agron Soc Environ. 95–100.
- Sohi, S.P., Mahieu, N., Arah, J.R.M., Powlson, D.S., Madari, B., Gaunt, J.L., 2001. A procedure for isolating soil organic matter fractions suitable for modeling. Soil Sci Soc Am J. 65, 1121–1128.
- Somasundaram, J.R.S., Chaudhary, D., Awanish K., Biswas, N.K., Sinha, M., Mohanty K.K., M., Hatia, P., Jhaa, M., Sankar, A.K., Patra, R., Dalal, S.K., Chaudha, I., 2018. Effect of contrasting tillage and cropping systems on soil aggregation, carbon pools and aggregateassociated carbon in rainfed Vertisols. Eur. J. Soil Sci. doi: 10.1111/ejss.12692.
- Sombroek, W.G., Braun, H.M.H., Van der Pouw, B.J.A., 1982. The exploratory soil map and agroclimate zone map of Kenya.
- Sood, M.C., Singh, N., 2003. Water management. In: Khurana, S.M.P., Minhas, J.S., Pandey, S.K. (Eds.), The Potato: Production and Utilization in Sub-tropics. Mehta Publishers, New Dehli, pp. 111–112.

- Steinweg, J.M., Dukes J., Wallenstein, M., 2012. Modeling the effects of temperature and moisture on soil enzyme activity: Linking laboratory assays to continuous field data. Soil Biol Biochem. 55, 85–92.
- Stevens, W., Hoeft, R., Mulvaney. R., 2005. Fate of nitrogen-15 in a long-term nitrogen rate study:II. Nitrogen uptake efficiency. Agron J. 97, 1046–1053.
- Stone, P.J., Wilson, D.R., Reid, J.B., Gillespie, R.N., 2001. Water deficit effects on sweet corn. I. Water use, radiation use efficiency, growth, and yield. Aust J Crop Past. Sci Agric Res. 52, 103–113.
- Struik, P.C., Geertsema J., Custers C.H.M.G., 1989. Effects of shoot, root and stolon temperature on the development of potato (*Solanum tuberosum* L.) tubers. Potato Res. 32, 151–158.
- Thornton, M.K., N.J. Malik, and R.B. Dwelle 1996. Relationship between leaf gas exchange characteristics and productivity of potato clones grown at different temperatures. American J Potato Res. 73, 63–77.
- Thorup-Kristensen, K., Kristian, Kirkegaard, J., 2016. Root system-based limits to agricultural productivity and efficiency: the farming systems context. Annal Bot. 1–20.
- Thorup-Kristensen, K., Magid, J., Jensen, L.S., 2003. Catch crops and green manuresas biological tools in nitrogen management in temperate zones. Adv Agron. 79, 227–302.
- Tian, G., Brussaard, L., Kang, B.T., 1993. Biological effects of plant residues with contrasting chemical compositions under humid tropical conditions-effects on soil fauna. Soil Eiol Eiochmi. 25, 731–737.
- Tian, G., Brussaard, L., Kang, B.T., Swift, M.J., 1997. Soil fauna-mediated decomposition of plant residues under constrained environmental and residue quality conditions. In: Driven

by nature: plant litter quality and decomposition (Cadisch G, Giller KE, eds.), pp. 125– 134. CAB International, Wallingford, UK.

- Tian, G., Kang, B.T., Brussard. L., 1992. Biological effects of plant residues with contrasting chemical compositions under humid tropical conditions-decomposition and nutrient release. Soil Biol Biochem. 24, 1051–1060.
- Tiessen, K.H.D., D.A., Lobb, G.R., Mehuys, H.W., Rees., 2007. Tillage translocation and tillage erosivity by planting, hilling and harvesting operations common to potato production in Atlantic Canada. Soil Till Res 97, 123–139.
- Tisdall, J.M., Oades, J.M., 1982. Organic matter and water-stable aggregates in soils. Europ J Soil Sci. 33, 141–163.
- Tosti, G., Benincasa, P., Farneselli, M., Pace, R., Tei, F., 2012. Green manuring effect of pure and mixed barley-hairy vetch winter cover crops on maize and processing tomato N nutrition. Europ J Agron. 43, 136–146.
- Trebejo, Midmore, 1990. Effect of water stress on potato growth, yield and water use in a hot and a cool tropical climate. J Agric Sci, Cam. 114, 321–334.
- Trinsoutrot, I., Recous, S., Bentz, B., Lineres, M., Cheneby, D., Nicolardotv, B., 2000. Biochemical quality of crop residues and carbon and nitrogen mineralization kinetics under nonlimiting nitrogen conditions. Soil Sci Soc Am J. 64, 918–926.
- Valladares, F., Pearcy, R.W., 2000. The role of crown architecture for light harvesting and carbon gain in extreme light environments assessed with a realistic 3-D Model. Anales del Jardín Botánico de Madrid. 58, 3–16.

- van Gestel, M., Merckx, R., Vlassak, K., 1993. Microbial biomass responses to soil drying and rewetting: The fate of fast- and slow-growing microorganisms in soils from different climates. Soil Biol Biochem. 25, 109–123.
- van Herwaarden, A.F., Farquhar, G.D., Angus, J.F., Richards, R.A., Howe, G.N., 1998. "Hayingoff", the negative grain yield response of dryland wheat to nitrogen fertiliser. I. Biomass, grain yield, and water use. Aust J Agric Res. 49, 1067–1081.
- Van Soest, P.J., Wine, R.H., 1968. Determination of Lignin and Cellulose in Acid-Detergent Fiber with Permanganate. Anal Chem. 51, 780–785.
- Vance, C.P., Uhde-Stone, C., Allan, D.L., 2003. Phosphorus acquisition and use: critical adaptations by plants for securing a nonrenewable resource. New Phytol. 157, 423– 447.
- Vance, E.D., Brookes, P.C., Jenkinson, D.S., 1987. An extraction method for measuring soil microbial biomass C. Soil Biol Biochem. 19, 703–707.
- Vandam, J., Kooman, P.L., Struik, P.C., 1996. Effects of temperature and photoperiod on early growth and final number of tubers in potato (*Solanum tuberosum* L.). Potato Res. 39, 51–62.
- Vargas-Gastelum, L., Romero-Olivares, A.L., Escalante, A.E., Rocha-Olivares, A., Brizuela C., Riquelme, M., 2015. Impact of seasonal changes on fungal diversity of a semi-arid ecosystem revealed by 454 pyrosequencing. FEMS Microb Ecol. 91:fiv044.
- Verhoef, H.A., 1995. Litter bag method. In: Alef K, Nannipieri P (eds) Methods in applied soil microbiology and biochemistry. Academic Press, New York, pp 485–487
- Vigil, M.F., Kissel, D.E., 1995. Rate of nitrogen mineralized from incorporated crop residues as influenced by temperature. Soil Sci Soc Am J. 59, 1636–1644.

- Wahid, A., Gelani, S., Ashraf, M., Foolad, M., 2007. Heat tolerance in plants: an overview. Environment Experiment Bot. 61(3), 199–223.
- Walvoord, M.A., D.A., Stonestrom, B.J., Andraski, Striegl, R.G., 2004. Constraining the inferred paleohydrologic evoluti on of a deep unsaturated zone in the Amargosa Desert. Vadose Zone J. 3, 502–512.
- Wang, B.L., Tang, X.Y., Cheng, L.Y., Zhang, A.Z., Zhang, W.H., Zhang, F.S., Liu, J.Q., Cao, Y., Allan, D.L., Vance, C.P., 2010. Nitric oxide is involved in phosphorus deficiencyinduced cluster-root development and citrate exudation in white lupin. New Phytol. 187, 1112–1123
- Wang, L., D'Odorico, P., Evans, J.P., Eldridge, D.J., McCabe, M.F., Caylor, K.K., King, E.G.,
  2012. Dryland ecohydrology and climate change: critical issues and technical advances.
  Hydrol Earth Syst Sci. 16, 2585–2603.
- Wang, Y.L., Almvik, M., Clarke, N., Eich-Greatorex, S., Ogaard, A.F., Krogstad, T., Lambers, H., Clarke, J.L., 2015. Contrasting responses of root morphology and root-exuded organic acids to low phosphorus availability in three important food crops with divergent root traits. Ann Bot. 7,1–11.
- West, T.O., Post, 2002. Soil organic carbon sequestration rates by tillage and crop rotation: A global data analysis. Soil Sci Soc Am J. 66, 1930–1946.
- Whitbread, A.M., Ayisi, K., Mabapa, P., Odhiambo, J.J.O., Maluleke, N., Pengelly, B.C., 2011. Evaluating *Lablab purpureus* L.: Sweet germplasm to identify short-season accessions suitable for crop and livestock farming systems in southern Africa Afr J Range Forage Sci. 28, 21–28.

- Whitehead, D C., Lockyer, D.R., Raistrick, N., 1988. The volatilization of ammonia from perennial ryegrass during decomposition, drying, and induced senescence. Ann Bot 61, 567–571.
- Wilkinson, S., Davies, J., 2002. ABA-based chemical signaling: the coordination of responses to stress in plants. Plant Cell Environ. 25, 195–210.
- Wyeh, R.D., Rains, D.W., 1978. Simultaneous measurements of nitrogen fixation estimated by acetylene-ethylene assay and nitrate absorption by soybean. Plant Physiol. 62, 442–448.
- Xiaolei, S., Zhifeng, W., 2002. The optimal leaf area index for cucumber photosynthesis and Production in Plastic Green House. ISHSActaHorticulturae, 633.XXVI International Horticultural Congress.
- Xiaolin, L., Su, Z., Liu, G., Zotarelli, L., Cui, Y., Snodgrass, C., 2016. Impact of soil moisture and temperature on potato production using seepage and center pivot. Agric Water Manage. 230–236.
- Xing, Z., Lien, C., Herb, W., Rees, F., Monteith, J., Lionel, S., 2011. A comparison of effects of one-pass and conventional potato hilling on water runoff and soil erosion under simulated rainfall. Can J Soil Sci. 91, 279–290.
- Yamanaka, T., Kaihotsu, I., Oyunbaatar, D., 2007 Summer time soil hydrological cycle and surface energy balance on the Mongolian steppe. J Arid Environ. 69, 65–79.
- Zahoor, A., Riaz, M., Ahmad, S., Ali, M., Khan, M.B., Javed, K., Anjum, M.A., Zia-Ul-Haq, M.A., Khan. 2010. Growth and radiation use efficiency of pearl millet as affected by hybrids, nitrogenous regimes and planting geometry under irrigated arid conditions. Pak J of Bot. 42, 3197–3207.

- Zemba, A.A., Wuyep, S.Z., Adebayo, A.A., Jahknwa, C.J., 2013. Growth and yield response of Irish potato (*Solanum tuberosum* L.) to climate in Jos-South, Plateau State, Nigeria. Global J Hum Soc Sci. 13(5), 13–18.
- Zeng, Q., Brown, P.H., 2000. Soil potassium mobility and uptake by corn under differential soil moisture regimes. Plant Soil, 221(2), 121–134.

## APPENDICES

Appendix 1: Soil water balance components measured under different cropping systems at the depth of 0–30 cm.

Agro-	- ·	∆SWC	Ι	R	RO	DP	Ер	ET	Σ
ecological zone	Cropping system	mm							
Upper	Pure potato	-18.0a	92.1	298.7	14.4e	32.5c	118.8c	218a	6.5a
midland	Pure lima bean	22.2c	92.1	298.7	6.7d	16.7b	40.6a	334c	21.7b
	Pure dolichos	25.4c	92.1	298.7	3.8bc	14.3b	41.5a	329bc	31.4c
	Potato + lima bean	-3.6b	92.1	298.7	2.4ab	15.6b	52.8b	301b	17.8b
	Potato + dolichos	-2.3b	92.1	298.7	5.6cd	10.7a	55.4b	306b	16.4b
Lower highland	Pure potato	-3.4a	71.9	411.1	55.5d	66.4e	76.7c	274a	9.0a
	Pure lima bean	28.8c	71.9	411.1	15.5b	22.3b	10.4a	427c	32.6b
	Pure dolichos	29.2c	71.9	411.1	9.8a	12.3a	16.9b	432c	37.2c
	Potato + lima bean	6.2b	71.9	411.1	23.3c	48.9d	19.5b	362b	31.5b
	Potato + dolichos	6.5b	71.9	411.1	17.7b	33.7c	17.3b	379b	37.8c
Upper highland	Pure potato	5.6а	14.6	469.7	60.2c	94.7d	23.7c	282a	14.7a
	Pure lima bean	14.3c	14.6	469.7	44.7b	46.3a	7.7a	345bc	36.3c
	Pure dolichos	9.8b	14.6	469.7	71.1d	84.4c	24.6	277a	22.4b
	Potato + lima bean	10.9b	14.6	469.7	38.6a	40.9a	9.8ab	347c	40.3c
	Potato + dolichos	9.5b	14.6	469.7	68.1d	64.4b	11.8b	316b	18.9b

Soil water storage ( $\Delta$ SWC), irrigation (I), rainfall (R), surface runoff (RO), deep percolation (DP), soil evaporation (Ep), evapotranspiration (ET) and total water balance ( $\Sigma$ ). Values followed by different letters within a column indicate significant differences at Tukey's p≤0.05. Values are 4 replicates expressed as averages over the four seasons.



Appendix 2: Ratio of actual evapotranspiration (ETa) to potential evapotranspiration (ETp) under different treatments in the three agro-ecological zones. Values are 4 replicates expressed as averages.



ELight fraction organic matter-nitrogen (LFOM-N)

Appendix 3: Distributions of SOM fractions within aggregate classes (µm) under different cropping systems. Bars with different letters for a given aggregate-size differ significantly between treatments (Tukey's  $p \le 0.05$ ).



Appendix 4: Carbon to nitrogen ratio of microbial biomass and total soil organic matter as influenced by soil depth (cm) and cropping systems. Bars with different letters within a soil depth denote significant differences between treatments ( $p \le 0.05$ ). Error bars present standard error of means. MBC and MBN=microbial biomass carbon and nitrogen respectively, TOC=total organic carbon; TN=total nitrogen.



Appendix 5: Microbial carbon biomass (MBC) in relation to lignin content under different cropping systems.

		2017 LR	2017 SR	2018 LR	2018 SR			
	Cropping system	Leached N (kg N ha <sup>-1</sup> )						
Upper	Pure potato	9.5±0.18d <sup>C</sup>	4.8±1.01d <sup>A</sup>	7.9±0.21c <sup>B</sup>	4.2±0.45c <sup>A</sup>			
midland	Pure lima bean	$3.4 \pm 0.49 c^{C}$	$1.8 \pm 0.61 c^{A}$	$2.4\pm0.32b^{B}$	$1.7 \pm 0.09 b^{A}$			
	Pure dolichos	$2.3 \pm 0.36 b^{B}$	$1.1 \pm 0.09 b^{A}$	$2.1\pm0.31b^{B}$	$1.9{\pm}0.05b^{A}$			
	Potato + lima bean	1.6±0.59a <sup>B</sup>	$1.2 \pm 0.21 b^{A}$	1.3±0.28a <sup>A</sup>	1.1±0.50a <sup>A</sup>			
	Potato + dolichos	1.3±0.17a <sup>B</sup>	$0.8 \pm 0.04 a^{A}$	1.4±0.33a <sup>B</sup>	0.9±0.09a <sup>A</sup>			
Lower	Pure potato	$37.0 \pm 2.95 c^{C}$	$25.5 \pm 2.15c^{A}$	$30.7\pm3.01d^B$	22.1±2.01c <sup>A</sup>			
highland	Pure lima bean	$10.1 \pm 0.39 b^{B}$	$4.1 \pm 1.04b^{A}$	$7.1\pm0.54c^{B}$	$4.9 \pm 1.54b^A$			
	Pure dolichos	$9.5 \pm 1.12b^{C}$	3.3±1.01ab <sup>A</sup>	$5.2 \pm 1.34 b^{B}$	2.9±0.36a <sup>A</sup>			
	Potato + lima bean	$5.8\pm0.35a^{B}$	3.1±0.56a <sup>A</sup>	$5.6 \pm 0.20 b^{A}$	$3.2 \pm 0.96 a^{B}$			
	Potato + dolichos	$4.7 \pm 0.12 a^{B}$	$3.4\pm0.28ab^A$	$4.3\pm0.56a^{B}$	3.0±0.21a <sup>A</sup>			
Upper highland	Pure potato	$46.6 \pm 3.42 d^{C}$	$34.4 \pm 3.62 d^{AB}$	$38.9{\pm}3.98c^{\rm B}$	$27.8 \pm 2.89 c^{A}$			
	Pure lima bean	14.0±2.05a <sup>B</sup>	9.1±2.14a <sup>A</sup>	10.8±2.23a <sup>A</sup>	9.7±0.43a <sup>A</sup>			
	Pure dolichos	$30.8 \pm 1.24 c^{C}$	25.3±1.12c <sup>B</sup>	$20.3 \pm 2.01 b^A$	$20.8 \pm 1.43 b^{A}$			
	Potato + lima bean	12.8±0.37a <sup>B</sup>	10.1±1.09a <sup>A</sup>	10.9±0.09a <sup>A</sup>	13.1±1.11a <sup>B</sup>			
	Potato + dolichos	$24.0{\pm}2.13b^{\text{B}}$	$19.9 \pm 0.51 b^{A}$	$19.0{\pm}2.01b^{A}$	$20.7{\pm}0.89b^{\rm A}$			
Analyses of variance								
CS	Seaso n	AEZ	CS* season	CS* AEZ	AEZ*season			
**	**	**	**	ns	ns			

Appendix 6: Seasonal variations in nitrogen leached under different cropping systems.

\*Significant at the 0.05 probability level, \*\*Significant at the 0.01 probability level, ns, not significant at the 0.05 probability level, SR, short rains season, LR, long rains season, CS, cropping system. Lower and uppercase letters indicate comparisons for means ( $\pm$  standard error) between the cropping systems and seasons, respectively at  $p \le 0.05$  by Tukey's HSD test.

		2017 LR	2017 SR	2018 LR	2018 SR
Agro-ecological zone	Cropping system	BNF (kg N ha <sup>-1</sup> )			
Upper	Pure potato	-	-	-	-
midland	Pure lima bean	3.4a <sup>B</sup>	2.3a <sup>A</sup>	3.0a <sup>B</sup>	2.0a <sup>A</sup>
	Pure dolichos	10.4b <sup>B</sup>	18.8b <sup>C</sup>	4.9a <sup>A</sup>	15.2c <sup>C</sup>
	Potato + lima bean	9.8b <sup>B</sup>	3.7a <sup>A</sup>	8.8b <sup>B</sup>	4.9b <sup>A</sup>
	Potato + dolichos	18.9c <sup>A</sup>	25.5b <sup>B</sup>	18.2c <sup>A</sup>	25.0d <sup>B</sup>
Lower	Pure potato	-	-	-	-
highland	Pure lima bean	5.2a <sup>B</sup>	3.6a <sup>A</sup>	5.0a <sup>B</sup>	3.1a <sup>A</sup>
	Pure dolichos	12.8c <sup>A</sup>	17.8c <sup>B</sup>	11.5c <sup>A</sup>	17.3c <sup>B</sup>
	Potato + lima bean	7.9b <sup>A</sup>	9.7b <sup>B</sup>	8.0b <sup>A</sup>	9.9b <sup>B</sup>
	Potato + dolichos	22.2d <sup>A</sup>	26.6d <sup>B</sup>	21.6d <sup>A</sup>	22.4d <sup>A</sup>
Upper	Pure potato	-	-	-	-
highland	Pure lima bean	5.3c <sup>A</sup>	3.3c <sup>B</sup>	5.7c <sup>A</sup>	6.4c <sup>B</sup>
	Pure dolichos	1.4a <sup>A</sup>	1.5a <sup>B</sup>	1.5a <sup>A</sup>	2.9a <sup>B</sup>
	Potato + lima bean	8.4d <sup>B</sup>	5.5d <sup>A</sup>	$8.0d^{B}$	9.8d <sup>C</sup>
	Potato + dolichos	3.2b <sup>A</sup>	2.8b <sup>A</sup>	3.1b <sup>A</sup>	4.1b <sup>B</sup>

Appendix 7: Seasonal nitrogen fixation (BNF) by legume intercropping under different cropping systems.

SR, short rains season, LR, long rains season. Lower and uppercase letters indicate comparisons for means between the cropping systems and seasons, respectively at  $p \le 0.05$  by Tukey's HSD test. Values are 4 replicates expressed as averages over the four seasons.

		2017 LR	2017 SR	2018 LR	2018 SR	
Agro-ecological zone	Cropping system	N uptake (kg N ha <sup>-1</sup> )				
Upper	Pure potato	90.8b <sup>B</sup>	70.8b <sup>A</sup>	91.2b <sup>B</sup>	64.2b <sup>A</sup>	
midland	Pure lima bean	49.9a <sup>C</sup>	30.8a <sup>A</sup>	36.6a <sup>B</sup>	27.1a <sup>A</sup>	
	Pure dolichos	55.7a <sup>C</sup>	40.8a <sup>B</sup>	34.4a <sup>A</sup>	32.5a <sup>A</sup>	
	Potato + lima bean	103.7c <sup>B</sup>	82.0c <sup>A</sup>	113.7c <sup>B</sup>	70.1b <sup>A</sup>	
	Potato + dolichos	131.6d <sup>C</sup>	114.8d <sup>B</sup>	120.1c <sup>B</sup>	101.2c <sup>A</sup>	
Lower	Pure potato	123.6c <sup>B</sup>	98.1b <sup>A</sup>	103.8c <sup>A</sup>	99.8c <sup>A</sup>	
highland	Pure lima bean	49.2a <sup>C</sup>	36.5a <sup>B</sup>	39.9a <sup>B</sup>	28.7a <sup>A</sup>	
	Pure dolichos	60.8b <sup>B</sup>	48.7a <sup>A</sup>	63.3b <sup>B</sup>	60.9b <sup>B</sup>	
	Potato + lima bean	117.9c <sup>C</sup>	92.9bc <sup>A</sup>	123.3d <sup>BC</sup>	128.6d <sup>B</sup>	
	Potato + dolichos	158.8d <sup>C</sup>	120.5d <sup>A</sup>	138.7e <sup>B</sup>	125.3d <sup>A</sup>	
Upper	Pure potato	128.2d <sup>C</sup>	119.7d <sup>B</sup>	118.1d <sup>B</sup>	101.5c <sup>A</sup>	
highland	Pure lima bean	49.7b <sup>A</sup>	48.1b <sup>A</sup>	45.5b <sup>A</sup>	40.5a <sup>A</sup>	
	Pure dolichos	18.1a <sup>A</sup>	27.8a <sup>B</sup>	20.2a <sup>A</sup>	32.9a <sup>B</sup>	
	Potato + lima bean	139.1e <sup>C</sup>	134.8e <sup>C</sup>	132.8e <sup>B</sup>	129.8d <sup>AB</sup>	
	Potato + dolichos	72.6c <sup>CD</sup>	69.2 <sup>BC</sup>	65.7c <sup>AB</sup>	76.3b <sup>D</sup>	

Appendix 8: Seasonal nitrogen uptake under different cropping systems.

SR, short rains season, LR, long rains season. Lower and uppercase letters indicate comparisons for means between the cropping systems and seasons, respectively at  $p \le 0.05$  by Tukey's HSD test. Values are 4 replicates expressed as averages over the four seasons.

-		2017 LR	2017 SR	2018 LR	2018 SR
Agro-ecological zone	Cropping system	N balance (kg N ha <sup>-1</sup> )			
Upper	Pure potato	-19.1a <sup>AB</sup>	-17.8a <sup>C</sup>	-14.9a <sup>D</sup>	-18.1a <sup>BC</sup>
midland	Pure lima bean	15.1d <sup>B</sup>	8.7d <sup>A</sup>	14.6d <sup>B</sup>	6.1d <sup>A</sup>
	Pure dolichos	27.5e <sup>C</sup>	16.7e <sup>A</sup>	20.8e <sup>B</sup>	13.1e <sup>A</sup>
	Potato + lima bean	3.0b <sup>B</sup>	2.1b <sup>A</sup>	3.1b <sup>B</sup>	$2.0b^{A}$
	Potato + dolichos	6.3c <sup>C</sup>	5.2c <sup>B</sup>	5.9c <sup>B</sup>	4.1c <sup>A</sup>
Lower	Pure potato	-16.3a <sup>A</sup>	-11.5a <sup>D</sup>	-12.9a <sup>CD</sup>	-13.2a <sup>BC</sup>
highland	Pure lima bean	24.4d <sup>C</sup>	19.6d <sup>B</sup>	21.7c <sup>B</sup>	14.2d <sup>A</sup>
	Pure dolichos	32.9e <sup>B</sup>	25.8e <sup>A</sup>	30.4d <sup>B</sup>	22.3e <sup>A</sup>
	Potato + lima bean	7.5b <sup>B</sup>	4.9b <sup>A</sup>	8.8b <sup>B</sup>	3.7b <sup>A</sup>
	Potato + dolichos	10.2c <sup>B</sup>	7.8c <sup>A</sup>	9.9b <sup>B</sup>	6.6c <sup>A</sup>
Upper	Pure potato	-8.0a <sup>B</sup>	-9.7a <sup>AB</sup>	-11.0a <sup>A</sup>	-10.7a <sup>A</sup>
highland	Pure lima bean	26.6e <sup>B</sup>	21.3d <sup>A</sup>	20.4d <sup>A</sup>	18.8d <sup>A</sup>
	Pure dolichos	2.5b <sup>AB</sup>	3.8b <sup>C</sup>	2.4b <sup>A</sup>	2.9b <sup>B</sup>
	Potato + lima bean	9.1d <sup>C</sup>	6.9c <sup>A</sup>	7.3c <sup>B</sup>	6.3c <sup>A</sup>
	Potato + dolichos	3.5c <sup>B</sup>	3.7b <sup>B</sup>	2.9b <sup>A</sup>	6.0c <sup>C</sup>

Appendix 9: Seasonal soil nitrogen balance under different cropping system	ns.
--	-----

SR, short rains season, LR, long rains season. Lower and uppercase letters indicate comparisons for means between the cropping systems and seasons, respectively at  $p \le 0.05$  by Tukey's HSD test. Values are 4 replicates expressed as averages over the four seasons.



Appendix 10: Schematic presentation of soil organic matter fractionation. Procedures are outlined in section 5.2.7.



Appendix 11: Schematic presentation of determination of soil microbial respiration. Procedures are outlined in section 5.2.9.



Appendix 12: Schematic presentation of fluorescein diacetate hydrolysis. Procedures are outlined in section 5.2.10.1.



(a)

(b)

Appendix 13: Runoff plots at 2 weeks after planting in the lower highland (a) and upper highland (b) agro-ecological zones.